TOWN OF BARNSTABLE

DEPARTMENT OF PUBLIC WORKS

FY25 Bismore Park Marina Bulkhead Rehabilitation



CONSULTING ENGINEER:	Foth Infrastructure & Environment, LLC Jeremy Packard, P.E.
DATE ISSUED:	January 29, 2025 12:00pm noon
BIDS DUE:	February 19, 2025 12:00pm noon
Virtual Bid Opening Via Zoom:	https://townofbarnstable-us.zoom.us/j/83287425104
Non-Mandatory Pre-Bid Meeting Date:	February 10, 2025 12:00pm noon
Pre-Bid Meeting Location:	Department of Public Works 382 Falmouth Road
Last Day To Submit Questions:	Hyannis, MA 02601 February 12, 2025 12:00pm noon
-	-

DIRECT BIDS TO: The Town's Bonfire Online Procurement Portal: https://townofbarnstable.bonfirehub.com

THIS PROJECT IS AVAILABLE ONLINE ONLY AND HARD COPY BIDS WILL NOT BE ACCEPTED. IT IS THE BIDDERS SOLE RESPONSIBILITY TO FAMILIARIZE THEMSELVES WITH THE BONFIRE ONLINE BID SUBMISSION PLATFORM AND BID SUBMISSION PROCESS AND REQUIREMENTS.

Contact:

Danielle Lamminen, Purchasing Agent, MCPPO ProcurementMailbox@town.barnstable.ma.us

All potential bidders are required to register on the Town of Barnstable Bonfire Online Procurement Portal: https://towonfbarnstable.bonfirehub.com. This document and any addenda thereto are issued electronically only. It is the responsibility of every bidder who receives this bid document and all associated documents to check the Town of Barnstable Bonfire Bid Submission portal above for any addenda or modification to this solicitation, if they intend to respond. The Town of Barnstable accepts no liability to provide accommodation to bidders who submit a response based upon an out of date solicitation document or documents obtained from a source other than the Town. Bidders may not alter (manually or electronically) the bid language or any bid documents. Unauthorized modifications to the body of the bid, specifications, terms, or conditions, or which change the intent of this bid are prohibited and may disqualify a response.

SECTION 1 – INVITATION FOR BID					
SECTION 2 – INSTRUCTIONS TO BIDDERS					
Subsection 1.	SECURING DOCUMENTS	8			
Subsection 2.	BID FORMS	8			
Subsection 3.	BID SECURITY	8			
Subsection 4.	DEFINITIONS	9			
Subsection 5.	BIDDER'S REPRESENTATION	9			
Subsection 6.	EXAMINATION OF BIDDING DOCUMENTS	10			
Subsection 7.	ADDENDA	10			
Subsection 8.	REJECTION OF BIDS	11			
Subsection 9.	QUALIFICATIONS OF BIDDER	11			
Subsection 10.	LABOR AND MATERIALS PAYMENT & PERFORMANCE BONDS	11			
Subsection 11.	"OR EQUAL" CLAUSE:	12			
Subsection 12.	SUBSTITUTIONS	12			
Subsection 13.	WORK TIME LIMITS	12			
Subsection 14.	TAX EXEMPTION	13			
Subsection 15.	ACCEPTANCE OF BIDS	13			
Subsection 16.	TIME FOR EXECUTING CONTRACT AND PROVIDED CONTRACT BONDS.	13			
Subsection 17.	PAYMENT OF EMPLOYEES	13			
Subsection 18.	WITHDRAWAL OF BIDS	14			
Subsection 19. PRICE ADJUSTMENT CLAUSES (Fuel, Asphalt, Portland Cement, Steel as defined) 14					
SECTION 3 – GENERAL CONDITIONS					
ARTICLE 1.	GENERAL PROVISIONS	15			
ARTICLE 2.	THE OWNER	17			
ARTICLE 3.	THE CONTRACTOR	18			
ARTICLE 4.	THE SUBCONTRACTORS	29			
ARTICLE 5.	WAGES AND EMPLOYMENT PRACTICES	31			
ARTICLE 6.	EQUAL EMPLOYMENT OPPORTUNITY & AFFIRMATIVE ACTION	35			
ARTICLE 7.	CONTRACTOR'S ACCOUNTING METHODS AND REQUIREMENTS	36			
ARTICLE 8.	THE CONSULTING ENGINEER	39			
ARTICLE 9.	CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS	41			
ARTICLE 10.	CHANGES IN THE WORK	43			
ARTICLE 11.	TIME	46			
ARTICLE 12.	PAYMENTS AND COMPLETION	48			
ARTICLE 13.	PROTECTION OF PERSONS AND PROPERTY	57			
ARTICLE 14.	INSURANCE AND BONDS	60			
ARTICLE 15.	UNCOVERING AND CORRECTION OF WORK	65			
ARTICLE 16.	TESTS, INSPECTIONS AND APPROVALS	66			

ARTICLE 17.	TERMINATION OR SUSPENSION OF THE CONTRACT	67		
ARTICLE 18.	CLAIMS AND DISPUTES	69		
ARTICLE 19.	MISCELLANEOUS PROVISIONS	70		
ARTICLE 20.	TRAFFIC CONTROL AND BARRICADES	71		
ARTICLE 21.	PARTIAL AWARD	71		
ARTICLE 22.	PRE-BID CONFERENCE	71		
ARTICLE 23.	ROAD OPENING/TRENCH PERMIT	71		
ARTICLE 24.	NOTIFICATION OF UTILITIES	72		
SECTION 4 – SF	PECIAL CONDITIONS	74		
SECTION 5 – BI	D SUBMITTAL REQUIREMENTS	74		
REQUIRED BID	SUBMITTAL FORM – FORM FOR GENERAL BID	75		
REQUIRED BID	SUBMITTAL FORM- SCHEDULE OF PRICES SUBMITTAL FORM – COMBINED CERTIFICATIONS	_77		
	SUBMITTAL FORM – CONTRACTOR REFERENCES			
	SUBMITTAL FORM – LIST OF SUB-CONTRACORS			
	SUBMITTAL FORM – LIST OF ALL EQUIPMENT TO BE USED			
	SECTION 6 – MINORITY & WOMEN OWNED BUSINESS PARTICIPATION REQUIREMENTS 8			
	REVAILING WAGE RATES			
	VNER-CONTRACTOR CONTRACT FORM			
	BOR AND MATERIALS PAYMENT BOND			
	PERFORMANCE BOND			
	CCEPTANCE OF BID NOTICE			
	PPLICATION & CERTIFICATION FOR PAYMENT			
	AX EXEMPTION NUMBER			
	OWN OF BARNSTABLE CHANGE ORDER			
SECTION 15 – N	IOTICE TO PROCEED	97		
APPENDIX A T	own of Barnstable Fuel Adjustment Clauses Barnstable Rev 8/2018	98		
	- Speciation Provisions, Monthly Price Adjustment for Hot Mix Asphalt (HMA)			
	 Speciation Provisions, Monthly Price Adjustment for Diesel Fuel and Gasolin 			
	- Speciation Provisions, Monthly Price Adjustment for Diesel Fuel and Gasolin			
	- Speciation Provisions, Price Adjustment for Structural Steel and Reinforcing			
DOCUMENT 00814	- Speciation Provisions, Price Adjustment for Portland Cement Concrete Mixes	104		
	upplemental General & Special Conditions Separate pdf			
SECTION 17 - T	echnical Specifications - Separate PDF	106		

Project Estimate:	\$ <u>800,000.00</u>

Prevailing Wages Apply dated: 1/23/2025

All bid documents are posted online on the Town of Barnstable Bonfire Procurement Portal. It is the bidders' responsibility to ensure that you have downloaded all posted documents for this bid. All Addendums are also issued in the Bonfire system.

FY25 Bismore Park Marina Bulkhead Rehabilitation

Please be sure to register as a user on the Town's Bonfire Procurement Portal: https://townofbarnstable.bonfirehub.com and login to view project obtain documents, submit bids and receive updates.

SECTION 1 – INVITATION FOR BID

TOWN OF BARNSTABLE INVITATION FOR BID

The Procurement Office on behalf of the Department of Public Works is requesting bids for the following:

FY25 BISMORE PARK MARINA BULKHEAD REHABILITATION

Bid specifications and Bid documents are available online only on the Town's Bonfire Procurement Portal: https://townofbarnstable.bonfirehub.com where they are publicly available as of 1/29/2025 at 12:00pm noon. Questions are due by Noon on 2/12/2025 in the Vendor Discussion section of the online Procurement Portal.

A non-mandatory, pre-bid meeting will take place at the DPW Conference room on 2/10/2025 at 12:00pm noon at the Department of Public Works, 382 Falmouth Road, Hyannis, MA 02601.

BIDS are due on or before 12:00 noon on 2/19/2025. ONLY BIDS SUBMITTED THROUGH THE TOWN'S BONFIRE ONLINE PROCUREMENT PORTAL WILL BE ACCEPTED. It is the Bidders' sole responsibility to familiarize themselves with the Bonfire Online Procurement Portal requirements and submission process. **Virtual Bid Opening Via ZOOM:** <u>https://townofbarnstable-us.zoom.us/j/83287425104</u>

This project involves rehabilitating the bulkhead at Bismore Park Marina, 180 Ocean Street in Hyannis. Work to be performed under this contract shall include, but is not limited to furnishing all labor, materials, and equipment necessary to perform the scope of work outlined below:

- Mobilization and Demobilization
- Bulkhead Section Loss Inspection (UT Readings)
- Bulkhead Plating
- Epoxy Repairs
- Splash Zone Epoxy Repairs
- Cathodic Protection
- Utility Installation and Pedestrian Walkway Restoration; and
- All other associated and incidental work stipulated in the construction plans and specifications.

A general description of the work to be performed under this Contract shall include, but will not be limited to, the construction operations described within the contract documents.

To receive consideration, Bids shall be submitted no later than the above date and time for the opening.

The successful bidder will be required to furnish a Labor and Materials Payment Bond and a Performance Bond each in the amount of one hundred percent (100%) of the contract amount.

To receive consideration, Bids shall be submitted on the appropriate forms no later than the above date and time schedule for the opening. Bids must be accompanied by a bid security in the amount of <u>five percent (5%) of the Bid Price</u> in the form of a bid bond or certified, treasurer's or cashier's check issued by a responsible bank or trust company. If submitting cashier's checks they must be received in the Procurement Office Lockbox before the bid close date. If, upon acceptance of the bid, a Bidder fails to enter into a Contract with the Town of Barnstable, the bid security shall be forfeited to and become the property of the Town.

LIQUIDATED DAMAGES: Contractor and Owner recognize that time is of the essence and that Owner will suffer financial loss if the Work is not completed within the times specified in the Contract, plus any extensions thereof allowed in accordance the General Conditions and this contract. The parties also recognize the delays, expense, and difficulties involved in proving in a legal or arbitration proceeding the actual loss suffered by Owner if the Work is not completed on time. Accordingly, instead of requiring any such proof, Owner and Contractor agree that as liquidated damages for delay (but not as a penalty), Contractor shall pay Owner **\$2,000.00** for each day that expires after the time specified in the Contract for Substantial Completion until the Work is substantially complete unless otherwise extended per an agreed and approved change order for work that was not included in the original bid. After Substantial Completion, if Contractor shall neglect, refuse, or fail to complete the remaining Work within the Contract Time or any proper extension thereof granted by Owner, Contractor shall pay Owner **\$2,000.00** for each day that expires after the time specified in the Contract for completion, if Contractor shall neglect, refuse, or fail to complete the remaining Work within the Contract Time or any proper extension thereof granted by Owner, Contractor shall pay Owner **\$2,000.00** for each day that expires after the time specified in the Contract for completion and readiness for final payment until the Work is completed and ready for final payment.

Full compliance with Federal, State and Municipal Wage Laws is required of all work done for the Town of Barnstable. Minimum Wage Rates as determined by the Commissioner of Department of Labor and Industries under the provision of the Massachusetts General Laws, Chapter 149, Section 26 to 27D, as amended, apply to this project. It is the responsibility of the contractor, before bid opening, to request if necessary, any additional information on Minimum Wage Rates for those trades people who may be employed for the proposed work under this contract. Prevailing Wage Rates dated are applicable to this project.

All bidders shall be required to provide Certification of Occupational Safety and Health Administration (OSHA) Training in accordance with Massachusetts General Law 30: Section 39S, as amended by Chapter 306 of the Acts of 2004, effective 7/1/06.

Bidders are not to include in their Bid sales and compensating use taxes on materials and supplies purchased for this project. All materials used are tax exempt.

Minority Business Enterprise (MBE) and Women's Business Enterprise (WBE) policies of the Town of Barnstable are applicable. The "Fair Share" construction <u>requirement</u> for this project is a minimum of **10.4%** combined MBE/WBE participation by state certified MBEs and WBEs. Within five days after the Bid Opening, the Bidder shall submit a "Schedule for Participation by Minority Business Enterprises" with accompanying Letters of Intent by each minority subcontractor proposed to be used by the Bidder. See Town Policy and Section 6 of this document for details of required submittals.

Contractors shall be required to comply with all applicable Massachusetts General Laws, Chapter 30 S.39M, and other applicable Massachusetts General Laws.

A weekly certified payroll submittal shall be required of the successful bidder in accordance with MGL C149, S27B. No payments will be made by the Town until all payroll information necessary for the Town to determine compliance with prevailing wage law requirements for the time period of the payment request have been submitted.

The Contractor shall not discriminate with regard to the personnel employed on this project on the basis of race, color, creed, national origin, gender, sexual preference, handicap or age.

The Town of Barnstable reserves the right to reject any or all Bids or waive any formalities that appear to be in the best interest of the Town. A Bid which includes, for any item, a bid that is abnormally low or high may be rejected as unbalanced. The right is also reserved to accept any Bid deemed to be best for the Town of Barnstable. In any event, bids to be deemed acceptable shall comply in each and every way with all applicable Massachusetts General Laws.

All inquiries with respect to this Invitation to Bid and the Contract Documents, including the Plans and Specifications, must be in writing through the Bonfire online procurement portal by the time and date listed above.

Please contact Bonfire at <u>Support@goBonfire.com</u> for technical questions related to your submission or visit Bonfire's help forum at https://bonfirehub.zendesk.com/hc.

END OF SECTION

SECTION 2 – INSTRUCTIONS TO BIDDERS

INSTRUCTIONS TO BIDDERS

Subsection 1.SECURING DOCUMENTS

- A. The Invitation for Bid, Instructions to Bidders, General Conditions, Special Conditions, Bid Drawings and Specifications, all Addenda issued prior to the execution of the Owner-Contractor Contract, Performance and Labor and Materials Payment Bonds, all amendments, Change Orders and written interpretations of the Contract Documents issued by the Town, Labor Rates, completed Bid and supporting forms signed and submitted by the Contractor, Applications and Certification for Payment, Owner-Contractor Contract and all other documents in these Project Specifications and Drawings referenced in the Contract compose the Contract Documents.
- B. All questions regarding this invitation to bid should be made in writing and submitted through the Town's Bonfire online procurement portal in the Vendor Discussion section in the FY25 Bismore Park Marina Bulkhead Rehabilitation project file. All bid documents are available on the Town of Barnstable's Bonfire procurement portal at: https://townofbarnstable.bonfirehub.com.

Subsection 2.BID FORMS

- A. All bid responses will be received on the Town's Bonfire online procurement portal: <u>https://townofbarnstable.bonfirehub.com</u> prior to the date and time specified in this Invitation to Bid. All required documents must be submitted in the format as specified, all fields must be completed as specified and the documents must be submitted the correct project folder in order for the bid response to be complete. Hard copy bid responses will not be accepted.
- B. All bids must be submitted per the above instructions. It is the responsibility of the bidder to ensure that bids are delivered to the specified location prior to the time and date designated.
- C. The Town may consider informal any bid not prepared and submitted in accordance with the provisions hereof and may waive any informalities or reject any and all bids, except as limited under the General Laws, Chapters 30 and 149, applicable sections, as amended to date.
- D. Any bid received after the time and date designated will not be considered. A bid will not be considered delivered unless the bid has been received online by the Town's Bonfire Online Procurement Portal (in the correct project) prior to the date and time specified in this document. The official date and time shall be the date and time showing online at https://townofbarnstable.bonfirehub.com.
- E. All bidders must note the maximum upload file size 1000MB. Do not embed any documents within uploaded files, as they will not be accessible or evaluated.
- F. Bid Table response templates can be obtained at <u>https://townofbarnstable.bonfirehub.com</u>. NOTE THAT BID TABLES MAY TAKE A SIGNIFICANT AMOUNT OF TIME TO PREPARE.
- G. Bids must upload submissions at https://townofbarnstable.bonfirehub.com. Bidder's submissions must be uploaded, submitted and finalized prior to the Closing Time and Due date posted in the portal. The Town strongly recommends that bidders give themselves sufficient time and at least one (1) day before the Due Date to begin the uploading process and to finalize its submission

- H. Bidders will receive an email confirmation receipt from the Bonfire portal with a unique confirmation number once its submission is finalized.
- I. Minimum system requirements: Internet Explorer 11, Microsoft Edge, Google Chrome, or Mozilla Firefox. JavaScript must be enabled, Browser cookies must be enabled.

Subsection 3.BID SECURITY

- A. Bid Security in the amount of FIVE PERCENT (5%) of the bid dollars (this includes all alternates, if any included in this bid) shall accompany each bid submittal. At the option of the Bidder, the security may be a bid bond issued by a surety authorized to do business in the Commonwealth, certified check, or treasurer's or cashier's check issued by a responsible bank or trust company, payable to the Town of Barnstable. A copy of the BID Security needs to be uploaded to the procurement portal in the correct project file. If submitting a check the original check needs to be received in the Procurement Office Lockbox, 367 Main Street, Hyannis, MA 02601 prior to bid closing. Personal or business checks will not be accepted.
- B. The bid security shall secure the execution of the Contract.
- C. Should any bidder to whom an award is made fail to enter into a Contract therefore within ten (10) days, Saturdays, Sundays, and legal holidays excluded, after notice of award has been mailed to him or fail within such time to furnish Performance and Payment Bonds as required, the amount so received from such bidder through their bond, certified check, treasurer's or cashier's check as bid deposit shall become the property of the Town of Barnstable, as liquidated damages; provided that the amount of the bid deposit which becomes the property of the Town of Barnstable shall not, in any event, exceed the difference between their bid price and the bid price of the next lowest responsible and eligible bidder; and that provided further that in the case of death, disability, bona fide clerical or mechanical error of a substantial nature, or other unforeseen circumstances affecting the bidder, their deposit shall be returned to them.
- D. Bid deposits of the three lowest responsible and eligible bidders will be held by the Awarding Authority during the time stipulated for the execution of the contracts and the submission of the performance bonds, and may be disposed of in such a manner as will accomplish the purpose for which they are submitted. After expiration of such period, bid guarantees not disposed, or the amounts thereof, will be returned within five (5) days, Saturdays, Sundays and legal holidays excluded.

Subsection 4.DEFINITIONS

- A. All definitions set forth in the General Conditions are applicable to all bidding documents, which include the Advertisement, Instructions to Bidders, Addenda issued prior to receipt of general bids.
- B. Addenda are written or graphic instruments issued prior to the execution of the contract which modify or interpret the bidding documents, including drawings and specifications, by additions, deletions, clarifications or corrections. Addenda will become part of the Contract Documents upon execution of the Contract.

Subsection 5.BIDDER'S REPRESENTATION

- A. Each bidder, in submitting their Bid, represents that they have read and understand the bidding documents, reports, test results, drawings, or other such documents provided by the Town pursuant to this bid.
- B. Each bidder represents that they have visited the site, familiarized themselves with the local conditions under which the work is to be performed, compared the site with the drawings and specifications, satisfied themselves of the conditions of delivery, handling

and storage of materials, and all other matters that may be incidental to the work that may affect: 1) the cost, progress or performance of the work; 2) the means, methods, techniques, sequences and procedures of construction; and 3) the bidder's safety precautions and programs, before submitting their Bid.

- C. Each bidder agrees at the time of submitting its bid that 1) the bidding documents are generally sufficient to indicate and convey understanding of all terms and conditions for the performance of the work; and 2) no further examinations, investigations, explorations, tests, studies or data are necessary for the determination of its bid for performance of the work at the price(s) bid and within the times required, and in accordance with the other terms and conditions of the bidding documents.
- D. Each bidder is familiar with all federal, state and local laws and regulations that may affect cost, progress or performance of the work.
- E. Each bidder represents that their bid is based upon the materials and equipment described in the bidding documents, including any addenda issued thereto.
- F. The submission of a bid will constitute an incontrovertible representation by the bidder that: 1) the bidder has complied with every requirement of this Section; 2) without exception, the bid submitted is premised upon performing and furnishing the work required by the bidding documents and applying any specific means, methods, techniques, sequences and procedures of construction that may be shown or indicated or expressly required by the bidding documents; 3) the bidder has given the Town written notice of all conflicts, errors, ambiguities and discrepancies that the bidder has discovered in the bidder; and 4) the bidding documents are generally sufficient to indicate and convey understanding of all terms and conditions for performing and furnishing the work. No allowance will subsequently be made to the successful bidder by reason of any error or omission on his part, due to his neglect in complying with the requirements of this article, except with respect to conflicts with the General Laws.

Subsection 6.EXAMINATION OF BIDDING DOCUMENTS

Each bidder shall examine the bidding documents carefully and, not later than date stated in IFB shall make a written request through the Vendor Discussion section in the project file on the Town's Bonfire Procurement Portal for interpretation or correction of any ambiguity, inconsistency or error therein which they may discover. Any interpretation or correction will be issued as an addendum and posted in the procurement portal. Only interpretations or correction by addendum shall be binding. No bidder shall rely upon any interpretation or correction given by any other method.

Subsection 7.ADDENDA

A. Prior to the receipt of the bids, addenda will be available for inspection on the Town's procurement portal.

B. Addenda issued during the time of bidding shall be listed on Bid forms in the space provided. Failure of a bidder to receive any addendum shall not release the bidder from any obligations under their bid, provided said addendum was sent by e-mail. Bidders shall check the Town's Bonfire Online Procurement Portal at

https://townofbarnstable.bonfirehub.com to verify number of addendums prior to submitting bid.

Subsection 8.REJECTION OF BIDS

The bidder acknowledges the right of the Town of Barnstable to reject any or all bids and to waive any informality or irregularity in any bid received. In addition, the bidder recognizes the right of the Town of Barnstable to reject a bid if the bidder fails to furnish any required bid security, or fails to submit the data required by the bidding documents, or if the bid is in any way incomplete or irregular.

Subsection 9.QUALIFICATIONS OF BIDDER

A. Any bidder, if requested, shall submit a financial statement, experience records, and an equipment schedule, on forms to be provided by the Town of Barnstable. Financial statements shall reflect true financial conditions of bidder within three months prior to date of bid opening and shall be validated by a Certified Public Accountant.

B. A bidder, in order to be eligible for the contract, must be able to show their financial ability to carry on the work until the project is complete and accepted by the Town of Barnstable.

- C. Experience references provided must be relevant in size and scope to the work being bid on and current. Town reserves the right to request additional financial or reference information.
- D. In evaluating the qualifications of bidders, the Town will consider past performance with the Town and outside references. Negative or poor references or poor performance on past work/projects for the Town shall constitute a reason to consider the bid non-responsible.

Subsection 10. LABOR AND MATERIALS PAYMENT & PERFORMANCE BONDS

A. Within ten (10) days after the date of Notice of Award of Contract, Saturdays, Sundays and legal holidays excluded, the bidder to whom the award is made shall furnish a performance bond and labor and materials bond, each equal to the full amount of the contract price, including accepted alternates (if applicable) to guarantee the faithful performance of all terms, covenants and conditions of the same. The bonds are to be issued by an acceptable bonding company qualified to do business under the laws of the Commonwealth of Massachusetts and satisfactory to the Town of Barnstable. Bonds must be in the form specified by the Town of Barnstable.

B. The performance bond shall guarantee the satisfactory completion of the project and that the contractor will make good any faults or defects in their work which may develop during the period of said guarantee as a result of improper or defective workmanship, material or apparatus. The full performance bond shall remain in effect until final payment is received by the Contractor or until any warranty period under the Contract expires, whichever is later.

The payment bond shall guarantee that the contractor shall pay in full all persons, firms or corporations who furnish labor or material or both labor and materials for, or on account of the work included herein. Payment bonds will be in effect until such time as the contractor furnishes proof that payment in full has been made for all materials used on the contract work. The bonds shall be paid for by the contractor. The Town of Barnstable shall have the right to demand proof that parties signing the bonds are duly authorized to do so.

C. Every such bond shall have a power of attorney attached thereto, authorizing the Town of Barnstable to enter judgment thereon in any court in the United States of America or elsewhere against the obligors therein named for the amount therein named and shall be conditioned for the honest and faithful compliance with all provisions of the bidder or bidders.

D. Separate Performance Bond and Labor and Materials Payment Bond forms shall be provided with Notice of Acceptance.

Subsection 11. "OR EQUAL" CLAUSE:

(Statutory reference: M.G.L. Ch.30, §39M(b)) Where products, materials or equipment are prescribed by manufacturer name, trade name, or catalog reference, the word "or approved equal" shall be understood to follow. An item shall be considered equal to the item so named or described if, in the opinion of the Consulting Engineer:

- a. it is at least equal in quality, durability, appearance, strength and design;
- b. it performs at least equally the function imposed by the general design for the public work being contracted for or the material being purchased; and
- c. it conforms substantially, even with deviations, to the detailed requirements for the items as indicated by the specifications.

Any structural or mechanical changes made necessary to accommodate substituted Equipment under this paragraph shall be at the expense of the Contractor or Subcontractor responsible for the work item. See other paragraphs of the General Conditions for any procedures that may be used in determining compliance with the standards of this paragraph.

Subsection 12. SUBSTITUTIONS

- A. The Bid shall be based on using the materials or products as specified and provided. Where several materials are specified by name for one use, any of those so specified may be supplied.
- B. Whenever the specified products or class of materials is specified exclusively by trade name, by manufacturer's name or by catalog reference, only such items shall be used, unless the Town's written approval for substitution is secured in accordance with the Conditions of the Contract.

Subsection 13. WORK TIME LIMITS

A. Contractor shall furnish a proposed construction schedule, in writing, with their bid. A final construction schedule shall be submitted at the time of delivery of the properly executed contract, bonds and certificates of insurance to the Awarding Authority, allowing for completion of the contract work prior to the date specified below and appearing on the Contract Form. Thereupon the Town of Barnstable will review the completed documents and proposed schedule, ask for revisions or corrections, or issue a "<u>NOTICE TO PROCEED</u>" indicating its Contract with final contract terms. The construction schedule shall be added as an exhibit to the contract and will contain construction start dates and substantial completion dates measured in days following the Notice to Commence. Failure to meet both the commencement date and/or the substantial completion date, and any other dates set forth in the

construction schedule, unless extended by the town in writing prior to the indicated date's, shall be deemed a default of a material condition of the contract.

B. The completion date for all of the work to be performed under this contract is 4/30/2025. Construction may begin following the receipt of notice to proceed and approval of all required Contractor submittals.

Subsection 14. TAX EXEMPTION

The project is exempt from payment of Massachusetts sales tax to the extent permitted by MGL Ch. 64H, Subsection 6F. **Exemption Certificate E-046-001-079** shall be used in lieu thereof and will be provided in the signed contract documents.

Subsection 15. ACCEPTANCE OF BIDS

Within thirty (30) days after the opening of the Bids the Town of Barnstable will act upon them. The acceptance of a Bid will be a Notice of Acceptance in writing signed by a duly authorized representative of the Town of Barnstable and accompanied by Performance and Labor and Materials Payment Bond forms. No other act of the Town of Barnstable shall constitute the acceptance of a Bid. The acceptance of the Bid shall bind the successful bidder to the contract. The rights and obligations provided for in the contract shall become effective and binding upon the parties only upon its formal execution.

Subsection 16. TIME FOR EXECUTING CONTRACT AND PROVIDED CONTRACT BONDS

Any contractor whose Bid shall be accepted will be required to execute the contract and furnish contract bonds within ten (10) days, Saturdays, Sundays and legal holidays excluded after the notice that the contract has been awarded to them.

Subsection 17. PAYMENT OF EMPLOYEES

- A. For work done in the Town of Barnstable, the payment for employees of the Contractor and any or all Sub-contractors and suppliers shall comply with the wage scale current at the commencement of construction, as published by the Department of Labor and Industries, under provisions of the Massachusetts General Laws. The Contractor and each of his Sub-contractors and suppliers shall pay each of their employees engaged in work on the project under the contract in full, less deductions made mandatory by law, and not less often than once a week. All forms required by local authorities, the Commonwealth of Massachusetts, and the United States Government, shall be properly submitted. No payments will be made on any application for payment until all required payroll and Affirmative Action/Equal Opportunity information for the period covered by the application has been submitted to the Town.
- B. A copy of applicable wage rate schedules is attached and forms part of the contract documents. For MULTI-YEAR projects, Awarding Authorities must request and provide to contractor, an Annual Update to this Prevailing Wage Schedule each year for the duration of the project, no later than two weeks before the anniversary date of the execution of the general contract. Annual updates are not required for projects that last LESS THAN ONE YEAR. Updated rates do not impact the contract value as the contractor is to take into account contract duration in pricing the initial bid.

Subsection 18. WITHDRAWAL OF BIDS

- A. At any time prior to the scheduled closing time for receipt of Bids, any bidder may withdraw his Bid, either personally or by email or written request. If withdrawal is made personally, proper receipt shall be given therefore.
- B. After the scheduled time for receipt of Bids and before award of contract, no bidder will be permitted to withdraw his Bid unless said award is delayed for a period exceeding thirty (30) days. Negligence on the part of the bidder in preparing his bid confers no rights for the withdrawal of the Bid after it has been opened.

Subsection 19. PRICE ADJUSTMENT CLAUSES (Fuel, Asphalt, Portland Cement, Steel as defined)

A. In accordance with Massachusetts General Laws, Chapter 30, Section 38A, Price adjustment clauses are applicable to this project for Diesel Fuel and Gasoline, Hot Mix Asphalt, Portland Cement Concrete Mixes, and Structural Steel and Reinforcing Steel and are included per Appendix A.

END OF SECTION

SECTION 3 – GENERAL CONDITIONS

ARTICLE 1. GENERAL PROVISIONS

1.1 CONTRACT DOCUMENTS:

The following Documents form the Contract and what is required by any one shall be as binding as if required by all:

- a. Invitation for Bid
- b. Instructions to Bidders
- c. General Conditions
- d. Special Conditions
- e. Bid Drawings and Technical Specifications
- f. All Addenda issued prior to the execution of this Contract
- g. Performance and Labor and Materials Payment Bonds
- h. All amendments, Change Orders, and written interpretations of the
- i. Contract Documents issued by the Town
- j. Labor Rates
- k. Completed Bid and supporting forms signed and submitted by
- I. Contractor
- m. Owner-Contractor Contract
- n. All other documents in these Project Specifications and Drawings referenced in the Bid compose the Contract Documents.

The intention of the Contract Documents is to include all labor, materials, equipment and other items necessary for the proper execution and completion of the Work and the terms and conditions of payment therefore, and also to include all Work which may be reasonably inferable from the Contract Documents as being necessary to produce the intended results. The Contract Documents are intended to be complementary; however, any inconsistency, ambiguity or conflict among the Contract Documents shall be resolved in the following order of precedence (with (i) having the highest priority) and in the manner most favorable to the Town: (i) Contract Modifications; (ii) Owner-Contractor Contract; (iii) General Conditions of the Contract; (iv) Drawings and Specifications.

1.1.1 One copy of the Contract Documents shall be signed by the Town of Barnstable and the Contractor. By executing the Contract, the Contractor represents that he has visited the site and familiarized himself with the local conditions under which the Work is to be performed.

1.1.2 The Contract Document represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations or agreements, either written or oral. The Contract may be amended or modified only by a Change Order. The Contract Documents shall not be construed to create a contractual relationship of any kind between 1) the Contractor and the Consulting Engineer or the Consulting Engineer's consultants, 2) the Owner and a Subcontractor or a Subsubcontractor, 3) the Owner and the Consulting Engineer or the Consultants, or 4) any persons or entities other than the Owner and the Contractor. The Consulting Engineer shall, however, be entitled to performance and enforcement of obligations under the Contract intended to facilitate performance of the Consulting Engineer's duties.

FY25 Bismore Park Marina Bulkhead Rehabilitation

1.2 WORK:

The term "Work" as used in the Contract Documents includes all labor necessary to produce the construction required by the Contract Documents, and all permits, materials and equipment incorporated or to be incorporated in such construction.

1.2.1 Organization of the Specifications into divisions, sections or articles of Work, and arrangement of Drawings, shall not control the Contractor in dividing the Work among Subcontractors.

1.3 PROJECT:

The Project is the total construction of which the Work performed under the Contract Documents may be the whole or a part and which may include construction by the Owner and by separate contractors.

1.4 DRAWINGS & SPECIFICATIONS:

The Drawings are the graphic and pictorial portions of the Contract Documents showing the design, location and dimensions of the Work, generally including plans, elevations, sections, details, schedules and diagrams. The Specifications are that portion of the Contract Documents consisting of the written requirements for materials, equipment, systems, standards and workmanship for the Work, and performance of related services. Each of these are produced for the purpose of the Contractor providing a formal bid for consideration by the Owner.

1.5 INSTRUMENTS OF SERVICE:

Instruments of Service are representations, in any medium of expression now known or later developed, of the tangible and intangible creative work performed by the Consulting Engineer and the Consulting Engineer's consultants under their respective professional services Contracts. Instruments of Service may include, without limitation, studies, surveys, models, sketches, drawings, specifications, and other similar materials.

1.6 INTERPRETATION:

In the interest of brevity, the Contract Documents frequently omit modifying words such as "all" and "any" and articles such as "the" and "an", but the fact that a modifier or an article is absent from one statement and appears in another is not intended to affect the interpretation of either statement.

1.7 USE OF DRAWINGS, SPECIFICATIONS AND OTHER INSTRUMENTS OF SERVICE:

The Contractor, Subcontractors, Sub-subcontractors and material or equipment suppliers are authorized to use and reproduce the drawings, specifications and other Instruments of Service provided to them solely and exclusively for execution of the Work. All copies made under this authorization shall bear the copyright notice, if any, shown on said documents. FY25 Bismore Park Marina Bulkhead Rehabilitation Page **16** of **106**

The Contractor, Subcontractors, Sub-subcontractors, and material or equipment suppliers may not use said documents on other projects or for additions to this Project outside the scope of the Work without the specific written consent of the Owner, Consulting Engineer and the Consulting Engineer's consultants.

ARTICLE 2. THE OWNER

2.1 GENERAL:

The terms "Owner" or "awarding authority" is the Town of Barnstable and is referred to throughout the Contract Documents as if singular in number. The term "Owner" means the owner's authorized representative, the Owner's Project Manager and/or Town Engineer or any other employee or agent of the Owner. With the exception of Article 8 or any other provision herein which grants the Consulting Engineer certain decision-making responsibilities, the Owner's Project Manager shall exercise authority, responsibility or otherwise bind the Owner with respect to all matters requiring the Owner's approval in any portion of the Work.

2.2 OWNER'S PROJECT REPRESENTATIVE:

The Town of Barnstable's Representative for this project and Owner's Project Manager will be: Paul Graves, P.E. for the Town of Barnstable. Once the project contract is signed, all project questions, shop drawings, samples and requirements for approvals shall be directed to:

Town of Barnstable Attn: Paul Graves, P.E. Department of Public Works 382 Falmouth Road Hyannis, MA 02601 Phone: ((774) 487-0641 (mobile/direct) (508) 790-6400 (main office)

2.2.1 The Owner's Project Manager shall represent the Owner throughout this Project. The Owner's Project Manager shall be responsible for all services associated with the management of the project. The Owner's Project Manager shall not have any responsibility for the design or the construction of the project. Such responsibilities shall remain with the Consulting Engineer and Contractor.

2.3 OWNER'S RIGHT TO ACCESS WORK SITE:

The Owner shall at all times have access to the Work wherever it is in preparation and progress. The Owner will make periodic visits to the site to become thoroughly familiar with the progress and quality of the Work in accordance with the Contract Documents. On the basis of on-site observations, the Owner's Project Representative will endeavor to guard against defects and deficiencies in the Work of the Contractor. The Owner will not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. The Owner will not be responsible for construction means, methods, techniques, sequences or procedures, or for safety precautions and programs in connection with the Work, and the Owner will not be responsible for the Contractor's failure to carry out the Work in accordance with Contract Documents.

2.4 INFORMATION AND SERVICES REQUIRED OF THE OWNER:

The Owner shall furnish information or services required of the Owner by the Contract Documents with reasonable promptness.

2.4.1 As deemed necessary by the Owner, the Owner shall furnish surveys describing physical characteristics and utility locations for the site of the Project.

2.4.2 The Owner will be, in the first instance, the interpreter of the requirements of the Contract Documents.

2.4.3 AutoCad files will be released to the Contractor for use for construction and in the preparation of record drawings of the Project to be provided upon completion of the Project to the Consulting Engineer for review and submission to the Owner.

2.4.4 The Owner shall receive from the Contractor periodic Applications for Payment for work performed by the Contractor. Based on observations, site visits and other such information, the Owner shall determine amounts owed to the Contractor and will issue Certificates for Payment in accordance with Article 12 herein.

2.5 OWNER'S RIGHT TO STOP THE WORK:

If the Contractor fails to correct Work that is not in accordance with the requirements of the Contract Documents as required by Article 15.2 in its entirety, or repeatedly fails to carry out Work in accordance with the Contract Documents, the Owner may issue a written order to the Contractor to stop the Work, or any portion thereof, until the cause for such order has been eliminated; however, the right of the Owner to stop the Work shall not give rise to a duty on the part of the Owner to exercise this right for the benefit of the Contractor or any other person or entity.

2.6 OWNER'S RIGHT TO CARRY OUT THE WORK:

If the Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents and fails within a ten (10) day period after receipt of written notice from the Owner to commence and continue correction of such default or neglect with diligence and promptness, the Owner may, without prejudice to other remedies the Owner may have, correct such deficiencies. In such case, an appropriate Change Order shall be issued, deducting from payments, then or thereafter due to the Contractor, the reasonable cost of correcting such deficiencies, including Owner's expenses and compensation for the Consulting Engineer's additional services made necessary by such default, neglect or failure. Such action by the Owner and amounts charged to the Contractor are both subject to prior approval of the Consulting Engineer. If payments then or thereafter due the Contractor are not sufficient to cover such amounts, the Contractor shall pay the difference to the Owner.

ARTICLE 3. THE CONTRACTOR

3.1 GENERAL:

The Contractor is the person or entity identified as such in the Contract and is referred to throughout the Contract Documents as if singular in number. The Contractor shall be lawfully licensed, if required in the Commonwealth of Massachusetts where the Project is located. The Contractor shall designate in writing a representative who shall have the express authority to bind the Contractor with respect to all matters under this Contract. The term "Contractor" means the Contractor or the Contractor's authorized representative.

3.1.1 The Contractor shall perform the Work in accordance with the Contract Documents.

3.1.2 The Contractor shall not be relieved of obligations to perform the Work in accordance with the Contract Documents either by activities or duties of the Consulting Engineer in the Consulting Engineer's administration of the Contract, or by tests, inspections or approvals required or performed by persons or entities other than the Contractor.

3.2 REVIEW OF CONTRACT DOCUMENTS AND FIELD CONDITIONS BY CONTRACTOR:

Execution of the Contract by the Contractor is a representation that the Contractor has visited the site, become generally familiar with local conditions under which the Work is to be performed and correlated personal observations with requirements of the Contract Documents.

3.2.1 Because the Contract Documents are complementary, the Contractor shall, before starting each portion of the Work, carefully study and compare the various Contract Documents relative to that portion of the Work, as well as the information furnished by the Owner pursuant to Article 2.4, shall take field measurements of any existing conditions related to that portion of the Work, and shall observe any conditions at the site affecting it. These obligations are for the purpose of facilitating coordination and construction by the Contractor and are not for the purpose of discovering errors, omissions, or inconsistencies in the Contract Documents; however, the Contractor shall promptly report to the Consulting Engineer any errors, inconsistencies or omissions discovered by or made known to the Contractor as a Request for Information in such form as the Consulting Engineer may require. It is recognized that the Contractor's review is made in the Contractor's capacity as a contractor and not as a licensed design professional, unless otherwise specifically provided in the Contract Documents.

3.2.2 The Contractor is not required to ascertain that the Contract Documents are in accordance with applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, but the Contractor shall promptly report to the Consulting Engineer any nonconformity discovered by or made known to the Contractor as a Request for Information in such form as the Consulting Engineer may require.

3.2.3 If the Contractor believes that additional cost or time is involved because of clarifications or instructions that the Consulting Engineer issues in response to the Contractor's notices or Requests for Information pursuant to Article 3.2.1 or 3.2.2, the Contractor shall make Claims as provided in Article 18. If the Contractor fails to perform the obligations of Articles 3.2.1 or 3.2.2, the Contractor shall pay such costs and damages to the Owner as would have been avoided if the Contractor had performed such obligations. If the Contractor performs those obligations, the Contractor shall not be liable to the Owner or Consulting Engineer for damages resulting from errors, inconsistencies or omissions in the Contract Documents, for differences between field

measurements or conditions and the Contract Documents, or for nonconformities of the Contract Documents to applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities, unless the Contractor recognized such error, inconsistency, omission or difference, and knowingly failed to report it to the Consulting Engineer.

3.2.4 The Contractor shall reimburse the Owner for costs incurred by the Consulting Engineer for design and construction administration services which are caused by the Contractor's inefficient or otherwise faulty administration or execution of its Work. These may include, but are not limited to the cost of the Consulting Engineer or Owner's Project Manager to perform:

- Repeated review of the Contractor's submittals and resubmittals substantially out of sequence from the submittal schedule provided by the Contractor and agreed to by the Consulting Engineer;
- b. An extensive number of responses to the Contractor's Requests for Information where such information is available to the Contractor from a careful study and comparison of the Contract Documents, field conditions, other Owner-provided information, Contractor-prepared coordination drawings, or prior Project correspondence or documentation;
- c. An extensive number of Change Orders and Change Directives requiring evaluation of Bids and the preparation or revision of Instruments of Service and not otherwise caused by the design defects of the Consulting Engineer;
- d. Consultation regarding replacement of Work resulting from fire or other cause during construction;
- e. Evaluation of an extensive number of claims not otherwise caused by design defects;
- f. Evaluation of substitutions proposed by the Contractor and making subsequent revisions to Instruments of Service resulting therefrom;
- g. Preparation of design and documentation for alternate bid or Bid requests proposed by the Contractor; or
- h. Contract administration services provided 45 days or more after Substantial Completion.

3.2.5 The Contractor shall conduct a preconstruction inspection of the worksite and notify the Owner in writing of any existing damage to the property or any unsafe conditions at the site prior to commencing the Work.

3.3 AUDIO VIDEO RECORDING SITE CONDITIONS:

Except as stipulated otherwise in the supplemental conditions, the Contractor shall submit a quality audio-video recording documenting Pre-Construction field conditions for the entire project and adjacent areas within 50 feet of the limits of work. The Pre-Construction video shall be submitted to the Owner and Consulting Engineer as one or more MP4 files viewable on Windows Media Player. The video(s) must be accepted by the Owner prior to commencing any Work or using any Contractor laydown areas.

3.4 SUPERVISION AND CONSTRUCTION PROCEDURES:

The Contractor shall supervise and direct the Work, using the Contractor's best skill and attention. The Contractor shall be solely responsible for, and have control over, construction means, methods, techniques, sequences and procedures and for coordinating all portions of the Work under the Contract, unless the Contract Documents give other specific instructions concerning these matters. If the Contract Documents give specific instructions concerning construction means, methods, techniques, sequences or procedures, the Contractor shall evaluate the jobsite safety thereof and, except as stated below, shall be fully and solely responsible for the jobsite safety of such means, methods, techniques, sequences or procedures. If the Contractor determines that such means, methods, techniques, sequences or procedures may not be safe, the Contractor shall give timely written notice to the Owner and Consulting Engineer and shall not proceed with that portion of the Work without further written instructions from the Consulting Engineer. If the Contractor is then instructed to proceed with the required means, methods, techniques, sequences or procedures without acceptance of reasonable changes proposed by the Contractor, the Owner shall be solely responsible for any loss or damage arising solely from those Owner-required means, methods, techniques, sequences or procedures.

3.4.1 The Contractor shall be responsible to the Owner for acts and omissions of the Contractor's employees, Subcontractors and their agents and employees, and other persons or entities performing portions of the Work for, or on behalf of the Contractor or any of its Subcontractors.

3.4.1.1 The Contractor shall, upon written request of the Owner, remove and replace workers whom the Owner deems to be disorderly, careless, incompetent, are illegal aliens, or to be employed in violation of the terms of the Contract Documents, at no increase in the Contract Sum or the Contract Time.

3.4.2 The Contractor shall be responsible for inspection of portions of Work already performed to determine that such portions are in proper condition to receive subsequent Work.

3.5 LABOR AND MATERIALS:

Unless otherwise provided for in the Contract Documents, the Contractor shall provide and pay for labor, materials, equipment, tolls, construction equipment and machinery, water, heat, utilities, transportation, and other facilities and services necessary for proper execution and completion of the Work, whether temporary or permanent and whether or not incorporated or to be incorporated in the Work.

3.5.1 Except in the case of minor changes in the Work authorized by the Consulting Engineer in accordance with Articles 3.16.7 or 10.4, the Contractor may make substitutions only with the consent of the Owner, after evaluation by the Consulting Engineer and in accordance with a Change Order.

3.5.2 The Contractor shall enforce strict discipline and good order among the Contractor's employees and other persons carrying out the Work. The Contractor shall not permit employment of unfit persons or persons not properly skilled in tasks assigned to them.

3.6 WARRANTY:

The Contractor warrants to the Owner that materials and equipment furnished under the Contract will be of good quality and new unless the Contract Documents require or permit otherwise. The Contractor further warrants that the Work will conform to the requirements of the Contract Documents and will be free from defects, except for those inherent in the quality of the Work the Contract Documents require or permit. Work, materials, or equipment not conforming to these requirements may be considered defective. The Contractor's warranty excludes remedy for damage or defect caused by abuse, alterations to the Work not executed by the Contractor, improper or insufficient maintenance, improper operation, or normal wear and tear and normal usage. If required by the Owner, the Contractor shall furnish satisfactory evidence as to the kind and quality of materials and equipment.

3.6.1 All warranties and guarantees regarding the workmanship by the General Contractor, Subcontractors and Sub-Subcontractors shall commence on the date of Substantial Completion. Warranties and guarantees shall extend for a minimum of one (1) year or longer if required by a particular specification. Notwithstanding the foregoing, any special warranties, as defined by the Consulting Engineer, required by the Contract Documents or manufacturer's standard warranties, extending longer than a year, shall remain in effect for the full warranty period.

3.7 TAXES:

The Contractor shall not pay, and the Owner shall not reimburse or pay the Contractor for any sales taxes for building supplies or materials for which exemption is provided by law. The Owner's Tax Exemption number to be used by the Contractor in this regard will be provided by the Owner to the Contractor.

3.8 PERMITS AND FEES:

Unless otherwise provided in the Contract Documents, the Contractor shall secure and pay for any applicable permits, fees, licenses, and inspections by government agencies necessary for proper execution and completion of the Work that are customarily secured after execution of the Contract and legally required at the time bids are received or negotiations concluded. The Town of Barnstable permitting authorities will not waive permit fees for the project.

3.9 NOTICES:

The Contractor shall comply with and give notices required by the Contract Documents, applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities applicable to performance of the Work.

3.9.1 Notice shall be deemed to have been duly served if delivered in person to the individual, to a member of the firm or entity, or to an officer of the corporation for which it was intended; or if delivered at, or sent by registered or certified mail or by courier service providing proof of delivery to the last business address known to the party giving notice.

3.10 COMPLIANCE WITH LAWS:

If the Contractor performs Work contrary to the Contract Documents, applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, the Contractor shall be responsible for such Work and shall bear the costs attributable to correction of said work, along with any other damages incurred by the Owner.

3.11 DEVIATIONS:

(Statutory reference: M.G.L. Ch.30, §39I): The Contractor shall perform all the work required by this contract in conformity with the plans and specifications contained herein. No willful and substantial deviation from said plans and specifications shall be made unless authorized in writing by the Owner or by the Consulting Engineer in charge of the work who is duly authorized by the Owner to approve such deviations. In order to avoid delays in the prosecution of the work required by such contract, such deviation from the plans or specifications may be authorized by a written order of the Owner or such Consulting Engineer so authorized to approve such deviation. Within thirty (30) days thereafter, such written order shall be confirmed by a certificate of the Owner stating:

- a. If such deviation involves any substitution or elimination of materials, fixtures, or equipment, the reasons why such materials, fixtures, or equipment were included in the first instance and the reasons for substitution or elimination, and, if the deviation is of any other nature, the reasons for such deviation, giving justification therefore;
- b. The specified deviation does not materially injure the project as a whole;
- c. Either the work substituted for the work specified is of the same cost and quality, or that an equitable adjustment has been agreed upon between the Owner and the Contractor and the amount in dollars of said adjustment; and
- d. The deviation is in the best interest of the Owner; such certificate shall be signed under the penalties of perjury and shall be a permanent part of the file record of the work contracted for.

3.12 DIFFERING SITE CONDITIONS

(Statutory reference: M.G.L. Ch.30, §39N): If during the progress of the Work, the Contractor or the Owner discovers that the actual subsurface or latent physical conditions encountered at the Work site differ substantially or materially from those shown on the plans or indicated in the Contract Documents, either the Contractor or the Owner may request an equitable adjustment in the contract price of the contract applying to Work affected by the differing site conditions. A request for such an adjustment shall be in writing and shall be delivered by the party making such claim to the other party within the time period provided and pursuant to Article 18.

3.12.1 (Statutory reference: M.G.L. Ch.30, §39N) Upon receipt of such a claim from a Contractor, or upon its own initiative, the Owner shall make an investigation of such physical conditions, and, if they differ substantially or materially from those shown on the plans or indicated in the Contract Documents or from those ordinarily encountered and generally recognized as inherent in work of the character provided for in the plans and Contract Documents and are of such a nature as to cause an increase or decrease in the cost of performance of the Work or a change in the construction methods required for the performance of the Work which results in an increase or decrease in the cost of the Work, the Owner shall make an equitable adjustment in the contract price and the contract shall be modified in writing accordingly.

3.13 ALLOWANCES/UNIT PRICES:

- a. **Allowances**: If applicable and requested in the bid documents, the Contractor shall include in the Contract Sum all allowances stated in the Contract Documents. Items covered by allowances shall be supplied for such amounts and by such persons or entities as the Owner may direct, but the Contractor shall not be required to employ persons or entities to whom the Contractor has reasonable objection.
- b. **Unit Prices**: If applicable. Unit pricing is only applicable if unit pricing requested in the bid documents, the Contractor shall include in the bid submittal package unit pricing on provided form in accordance with provided standards as designed or with reference as defined by MassDOT, Federal Highway, FAA, etc. for example as applicable. Quantities shall be controlled by change order.
 - i. Bids will be compared on the estimate of quantities of work to be done, as shown on the bid.
 - ii. The contractor expressly agrees that these quantities are being set forth for the comparison of bids only and that the actual amount of work may not correspond therewith. The Town expressly reserves the right to adjust said quantities in accordance with actual conditions as found to exist during the course of work. The Contractor further agrees that any increase or decrease in the quantity for any item shall not be regarded as cause for an increase in the contract unit prices, or in the time allowed for completion of the work except as provided in the contract.

3.13.1 Unless otherwise provided in the Contract Documents:

- a. Allowances shall cover the cost to the Contractor of materials and equipment delivered at the site, less any applicable trade discounts;
- b. Contractor's costs for unloading and handling at the site, labor, installation costs, overhead, profit and other expenses contemplated for stated allowance amounts shall be included in the Contract Sum but not in the allowances; and
- c. Whenever costs are more than or less than the allowances, the Contract Sum shall be adjusted accordingly by a Change Order. The amount of the Change Order shall reflect
 - i. the difference between actual costs and the allowances under Article 3.12.1 (a), and
 - ii changes in the Contractor's costs under Article 3.12.1 (b).

3.13.2 Materials and equipment under an allowance shall be selected by the Owner with reasonable promptness.

3.14 CONTRACTOR'S AUTHORIZED SITE REPRESENTATIVE:

The Contractor shall employ a competent project manager/superintendent and necessary assistants who shall be in attendance at the Project site during performance of the Work. The Contractor's project manager/superintendent shall represent the Contractor, and communications:

- a. Given by the Owner to the Contractor's project manager/superintendent shall be as binding as if given to the Contractor;
- b. Given by the Contractor's project manager/superintendent to the Owner shall be as binding as if given by the Contractor; and

c. Given by the Contractor's project manager/superintendent to any Subcontractor, Sub-subcontractor or any other party performing work or providing supplies on behalf the Contractor shall be a binding as if given by the Contractor.

3.14.1 The Contractor, as soon as practicable after award of the Contract, shall furnish in writing to the Owner through the Consulting Engineer the name and qualifications of a proposed Contractor's project manager/superintendent. The Consulting Engineer may reply within fourteen (14) days to the Contractor in writing, stating:

- a. Whether the Owner has reasonable objection to the proposed Contractor's project manager/superintendent, or
- b. That the Owner requires additional time to review. Failure of the Consulting Engineer to reply within the 14 day period shall constitute notice of no reasonable objection, unless the Owner has deemed it necessary for additional time to review.

3.14.2 The Contractor shall not employ a proposed Contractor's project manager/superintendent to whom the Owner has made reasonable and timely objection. Additionally, the Contractor shall not change the Contractor's project manager/superintendent without the Owner's consent, which shall not unreasonably be withheld or delayed. Any Contractor's project manager/superintendent or assistant(s) replaced by the Contractor for whatever reason shall not cause any increase or change in the Contract Sum or Contract Time.

3.14.3 The Contractor's representative, as well as representatives of any Subcontractor, or material or supply contractor shall attend any meetings as requested by the Owner.

3.15 CONTRACTOR'S CONSTRUCTION SCHEDULES:

The Contractor, promptly after being awarded the Contract, shall prepare and submit for the Owner's and Consulting Engineer's approval a Contractor's Construction Schedule for the Work. The schedule shall not exceed time limits currently under the Contract Documents, shall be revised at appropriate intervals as required by the conditions of the Work and Project, shall be related to the entire Project to the extent required by the Contract Documents, and shall provide for expeditious and practicable execution of the Work.

3.15.1 The Contractor shall prepare a Submittal Schedule, promptly after being awarded the Contract and thereafter as necessary to maintain a current Submittal Schedule, and shall timely submit revised schedules for the Consulting Engineer's and Owner's approval. The Consulting Engineer's and Owner's Project Manager's approval shall not unreasonably be delayed or withheld. The Submittal Schedule shall:

- a. Be coordinated with the Contractor's Construction Schedule, and
- b. Allow the Consulting Engineer reasonable time to review submittals. If the Contractor fails to submit a Submittal Schedule, the Contractor shall not be entitled to any increase in Contract Sum or extension of Contract Time based on the time required for review of submittals.

3.15.2 The Contractor shall perform the Work in general accordance with the most recent schedules submitted to the Owner and Consulting Engineer.

3.16 DOCUMENTS AND SAMPLES AT THE SITE:

The Contractor shall maintain at the site for the Owner one (1) copy of the Drawings, Specifications, Addenda, Change Orders and other Modifications, in good order, appropriately organized and sequenced, and marked currently to indicate field changes and selections made during construction, and one (1) copy of approved Shop Drawings, Product Data, Samples and similar required submittals. These shall be available to the Consulting Engineer and shall be delivered to the Owner upon completion of the Work as a record of the Work as constructed.

3.17 SHOP DRAWINGS, PRODUCT DATA AND SAMPLES:

Shop Drawings are drawings, diagrams, schedules and other data specially prepared for the Work by the Contractor or a Subcontractor, Sub-subcontractor, manufacturer, supplier or distributor to illustrate some portion of the Work.

3.17.1 Product Data are illustrations, standard schedules, performance charts, brochures, diagrams and other information furnished by the Contractor to illustrate materials or equipment for some portion of the Work.

3.17.2 Samples are physical examples that illustrate materials, equipment or workmanship and establish standards by which the Work will be judged.

3.17.3 Shop Drawings, Product Data, Samples and similar submittals are not Contract Documents. Their purpose is to demonstrate the way by which the Contractor proposes to conform to the information given and the design concept expressed in the Contract Documents for those portions of the Work for which the Contract Documents require submittals. Review by the Consulting Engineer is subject to the limitations of Article 8.3.3. Informational submittals upon which the Consulting Engineer is not expected to take responsive action may be so identified in the Contract Documents. Submittals that are not required by the Contract Documents may be returned by the Consulting Engineer without action.

3.17.4 The Contractor shall review for compliance with the Contract Documents, approve and submit to the Consulting Engineer Shop Drawings, Product Data, Samples and similar submittals required by the Contract Documents in accordance with the Submittal Schedule approved by the Consulting Engineer or, in the absence of an approved Submittal Schedule, with reasonable promptness and in such sequence as to cause no delay in the Work or in the activities of the Owner or separate contractors.

3.17.5 By submitting Shop Drawings, Product Data, Samples and similar submittals, the Contractor represents to the Owner and Consulting Engineer that the Contractor has 1) reviewed and approved them, 2) determined and verified materials, field measurements and field construction criteria related thereto, or will do so, and 3) checked and coordinated the information contained within such submittals with the requirements of the Work and of the Contract Documents.

3.17.6 The Contractor shall perform no portion of the Work for which the Contract Documents require submittal and review of Shop Drawings, Product Data,

Samples or similar submittals until the respective submittal has been approved by the Consulting Engineer as authorized by the Owner's Project Manager.

3.17.7 The Work shall be in accordance with approved submittals except that the Contractor shall not be relieved of responsibility for deviations from requirements of the Contract Documents by the Consulting Engineer's approval of Shop Drawings, Product Data, Samples or similar submittals unless the Contractor has specifically informed the Consulting Engineer in writing of such deviation at the time of submittal and 1) the Consulting Engineer has given written approval as authorized by the Owner's Project Manager to the specific deviation as a minor change in the Work, or 2) a Change Order has been issued authorizing the deviation. The Contractor shall not be relieved of responsibility for errors or omissions in Shop Drawings, Product Data, Samples or similar submittals by the Consulting Engineer's approval thereof.

3.17.8 The Contractor shall direct specific attention, in writing or on resubmitted Shop Drawings, Product Data, Samples or similar submittals, to revisions other than those requested by the Consulting Engineer on previous submittals. In the absence of such written notice, the Consulting Engineer's approval of a resubmission shall not apply to such revisions.

3.17.9 The Contractor shall not be required to provide professional services that constitute the practice of architecture or Consulting Engineering unless such services are specifically required by the Contract Documents for a portion of the Work unless the Contractor needs to provide such services in order to carry out the Contractor's responsibilities for construction means, methods, techniques, sequences and procedures. The Contractor shall not be required to provide professional services in violation of applicable law. If professional design services or certifications by a design professional related to systems, materials or equipment are specifically required of the Contractor by the Contract Documents, the Owner and the Consulting Engineer will specify all performance and design criteria that such services must satisfy. The Contractor shall cause such services or certifications to be provided by properly licensed design professional whose signature and seal shall appear on all drawings, calculations, specifications, certifications, Shop Drawings and other submittals prepared by such professional. Shop Drawings and other submittals related to the Work designed or certified by such professional, if prepared by others, shall bear such professional's written approval when submitted to the Consulting Engineer. The Owner and the Consulting Engineer shall be entitled to rely upon the adequacy, accuracy and completeness of services, certifications and approvals performed or provided by such design professionals, provided the Owner and Consulting Engineer have specified to the Contractor all performance and design criteria that such services must satisfy. Pursuant to this paragraph, the Consulting Engineer will review, approve or take other appropriate action on submittals only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Contractor shall not be responsible for the adequacy of the performance and design criteria specified in the Contract Documents.

3.18 USE OF THE SITE:

The Contractor shall confine operations at the site to areas permitted by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities and the Contract Documents and shall not reasonably encumber the site with materials or equipment.

3.19 CUTTING AND PATCHING:

The Contractor shall be responsible for cutting, fitting or patching required to complete the Work or to make its parts fit together properly. All areas requiring cutting, fitting, and patching shall be restored to the condition existing prior to the cutting, fitting and patching, unless otherwise required by the Contract Documents.

3.19.1 The Contractor shall not damage or endanger a portion of the Work or fully or partially completed construction of the Owner or separate contractors by cutting, patching or otherwise altering such construction, or by excavation. The Contractor shall not cut or otherwise alter such construction by the Owner or a separate contractor except with written consent of the Owner and of such separate contractor; such consent shall not be unreasonably withheld. The Contractor shall not unreasonably withhold from the Owner or a separate contractor the Contractor's consent to cutting or otherwise altering the Work.

3.20 CLEANING UP:

The Contractor shall regularly keep the premises and surrounding area free from accumulation of waste materials or rubbish caused by operations under the Contract. At completion of the Work, the Contractor shall remove waste materials, rubbish, the Contractor's tools, construction equipment, machinery and surplus materials from and about the Project Site. Prior to removal of surplus material, the Contractor shall notify the Owner of surplus material inventory and the Owner reserves the right to retain surplus materials if deemed by the owner to be in the owner's best interests.

3.20.1 If the Contractor fails to clean up as provided in the Contract Documents, the Owner may do so and the Owner shall be entitled to reimbursement from the Contractor.

3.20.2 At the end of each work day, the Contractor will be responsible to secure the job site being worked on in a manner satisfactory to the Owner's Project Manager.

3.21 INDEMNIFICATION:

To the fullest extent permitted by law, the Contractor shall indemnify, defend, and save harmless the Town and all of the Town officers, agents and employees from and against all suits and claims of liability of every name and nature, including attorney's fees and costs of defending any action or claim, for or on account of any claim, loss, liability or injuries to persons or damage to property of the Town or any person, firm, corporation or association arising out of or resulting from any act, omission, or negligence of the Contractor, subcontractors and their agents or employees in the performance of the work covered by this Contract and/or their failure to comply with terms and conditions of this Contract, regardless of whether said claim is caused in part by the Town or any third party. The foregoing provisions shall not be deemed to be released, waived or modified in any respect by reason of any surety or insurance provided by the Contractor under this Contract with the Town. The provisions of this paragraph shall survive the termination or expiration of the Contract.

3.21.1 In claims against any person or entity under Article 3.20 by an employee of the Contractor, a Subcontractor, anyone directly or indirectly employed by them or anyone for whose acts they may be liable, the indemnification obligation under Article 3.20 shall not be limited by a limitation on amount or type of damages, compensation

or benefits payable by or for the Contractor or a Subcontractor under workers' compensation acts, disability benefit acts or other employee benefit acts.

3.22 ROYALTIES, PATENTS AND COPYRIGHTS:

The Contractor shall pay all royalties and license fees. The Contractor shall defend suits or claims for infringement of copyrights and patent rights and shall hold the Owner harmless from loss on account thereof, but shall not be responsible for such defense or loss when a particular design, process or product of a particular manufacturer is required by the Contract Documents, or where the copyright violation is contained in Drawings, Specifications or other documents prepared by the Owner or Consulting Engineer.

3.23 AS-BUILT DRAWINGS:

Except as stipulated otherwise in the supplemental conditions, upon completion of the project and prior to final payment, the Contractor shall provide the Owner an as-built survey prepared in conformance with 250 CMR, Sections 6.01 and 6.02, from update Nov. 22, 2013 by a 3rd Party Professional Licensed Surveyor (P.L.S). The completed survey shall be presented on a black line print(s) and stamped and signed by a Professional Surveyor registered in the Commonwealth of Massachusetts. A digital copy of the survey in AutoCad shall accompany the print(s). The as-built shall include all changes made to the work during construction and shall use the NAVD 88 and NAD 83, unless otherwise agreed by the Owner in writing. The survey shall include elevations at centerline, face of curb, top of curb, back of sidewalk and any off-site areas of work. An appropriate number of benchmarks shall be set in the job vicinity. Manholes shall be shown with rim elevations, inverts, pipe types and sizes and all pertinent information to determine flow rates. Water lines shall be shown with the size and type of pipe, services and valves well as depth beneath finished grades. All utilities shall be shown on the as-built. All pole relocations, light poles, etc. shall be depicted. Pavement striping, color and character. If building construction was included, an accurate location with dimensions to the boundary line(s) shall be shown. Floor elevations of each floor as well as building height, outside lighting and all parking and striping. Accurately record the details of all completed works.

3.24 SURVEY MONUMENTS, BOUNDS, AND MARKERS:

The Contractor shall protect survey monuments during the course of construction and shall replace any property corners, roadway bounds or geodetic control markers destroyed during construction.

ARTICLE 4. THE SUBCONTRACTORS

4.1 GENERAL:

"Subcontractor" as used in this Article for contracts awarded:

- a. As provided in Massachusetts General Laws, Chapter 30, Section 39M, shall mean a person approved by the Owner in writing as a person performing labor or both performing labor and furnishing materials pursuant to a contract with the Contractor, and
- b. Shall also mean a person contracting with the Contractor to supply materials used or employed in a public works project for a price in excess of five thousand dollars.

4.1.1 A Sub-subcontractor is a person or entity who has a direct or indirect contract with a Subcontractor to perform a portion of the Work. The term "Sub-subcontractor" is referred to throughout the Contract Documents as if singular in number.

4.2 AWARD OF SUBCONTRACTS AND OTHER CONTRACTS FOR PORTIONS OF THE WORK:

Unless otherwise stated in the Contract Documents or the bidding requirements, the Contractor, as soon as practicable after the award of the Contract, shall furnishing in writing to the Owner through the Consulting Engineer the names of persons or entities (including those who are to furnish materials or equipment fabricated to a special design) proposed for the Work. The Consulting Engineer may reply within seven (7) days to the Contractor in writing stating:

- a. Whether the Owner has reasonable objection to any such proposed person or entity, or
- b. That the Consulting Engineer requires additional time for review. Unless the Consulting Engineer fails to request additional time for review, failure of the Owner or Consulting Engineer to reply within seven (7) days shall constitute notice of no reasonable objection.

4.2.1 The Contractor shall not contract with a proposed person or entity to whom the Owner or Consulting Engineer has made reasonable and timely objection.

4.2.2 If the Owner or Consulting Engineer has reasonable objection to a person or entity proposed by the Contractor, the Contractor shall propose another to whom the Owner or Consulting Engineer has no reasonable objection.

4.2.3 The Contractor shall not substitute a Subcontractor to perform any portion of the Work without the approval of the Owner, pursuant to Article 4.2.

4.3 CONTRACTUAL RELATIONS:

By appropriate Contract pursuant to Massachusetts General Laws, Chapters 30 and 149, the Contractor shall require each Subcontractor, to the extent of the Work to be performed by the Subcontractor, to be bound to the Contractor by terms of the Contract Documents, and to assume toward the Contractor all the obligations and responsibilities, including the responsibility for safety of the Subcontractor's Work, which the Contractor, by these Contract Documents, assumes toward the Owner and Consulting Engineer. Each subcontract Contract shall preserve and protect the rights of the Owner and Consulting Engineer under the Contract Documents with respect to the Work to be performed by the Subcontractor so that subcontracting thereof will not prejudice such rights, and shall allow to the Subcontractor, unless specifically provided otherwise in the subcontract Contract, the benefit of all rights, remedies and redress against the Contractor that the Contractor, by the Contract Documents, has against the Owner or Consulting Engineer. Where appropriate, the Contractor shall require each Subcontractor to enter into similar Contracts with Sub-subcontractors. The Contractor shall make available to each proposed Subcontractor, prior to the execution of the subcontract Contract, copies of the Contract Documents to which the Subcontractor will be bound, and upon written request of the Subcontractor, identify to the Subcontractor terms and conditions of the proposed subcontract Contract that may be at variance with

the Contract Documents. Subcontractors will similarly make copies of applicable portions of such documents available to their respective Sub-subcontractors.

4.4 CONTINGENT ASSIGNMENT OF SUBCONTRACTS:

Each subcontract Contract for a portion of the Work is assigned by the Contractor to the Owner, provided that:

- Assignment is effective only after termination of the Contract between the Contractor and the Owner for cause pursuant to Article 17, and only for those subcontract Contracts that the Owner accepts by notifying the Subcontractor and Contractor in writing; and
- b. Assignment is subject to the prior rights of the Surety, if any, obligated under bond relating to the Contract.

4.4.1 When the Owner accepts the assignment of a subcontract Contract, the Owner assumes the Contractor's rights and obligations under the subcontract.

4.4.2 Upon such assignment, if the Work has been suspended for more than thirty (30) days, the Subcontractor's compensation may be equitably adjusted for increases in cost resulting from the suspension.

4.4.3 Upon such assignment to the Owner under Article 4.4, the Owner may further assign the subcontract to a successor contractor or other entity. If the Owner assigns the subcontract to a successor contractor or other entity, the Owner shall nevertheless remain legally responsible for all of the successor contractor's obligations under the subcontract as it had with the original contractor.

ARTICLE 5. WAGES AND EMPLOYMENT PRACTICES

5.1 PREFERENCE TO VETERANS AND CITIZENS, AND RATES OF PAY

(Statutory reference: M.G.L. Ch.149, §26): In the employment of mechanics and apprentices, teamsters, chauffeurs and laborers, preference shall first be given to citizens of the Commonwealth who have been residents of the Commonwealth for at least six (6) months at the commencement of their employment, who are veterans as defined in Massachusetts General Laws, Chapter 4, Section 7, Clause 43, and who are qualified to perform the work to which the employment relates; and secondly, to citizens of the Commonwealth generally who have been residents of the Commonwealth for at least six months at the commencement of their employment relates; of the Commonwealth for at least six months at the commencement of their employment, and if they cannot be obtained in sufficient numbers, then to citizens of the United States, and every contract for such work shall contain a provision to this effect. Each county, town or district in the construction of public works, or persons contracting or subcontracting for such works, shall give preference to veterans and citizens who are residents of such county, town or district.

5.1.1 The rate per hour of the wages paid to said mechanics and apprentices, teamsters, chauffeurs and laborers in the construction of public works shall not be less than the rate or rates of wages to be determined by the Commissioner of Labor and Industries as hereinafter provided; provided, that the wages paid to laborers employed on said works shall not be less than those paid to laborers in the municipal service of the town or towns where said works are being constructed; provided, further, that where the same public work is to be constructed in two or more towns, the wages paid FY25 Bismore Park Marina Bulkhead Rehabilitation Page **31** of **106**

to laborers shall not be less than those paid to laborers in the municipal service of the town paying the highest rate; provided further, that if, in any of the towns where the works are to be constructed, a wage rate or wage rates have been established in certain trades and occupations by collective agreements or understandings between organized labor and employers, the rate or rates to be paid on said works shall not be less than the rates so established; provided, further, that in towns where no such rate or rates have been so established, the wages paid to mechanics, teamsters, chauffeurs and laborers on public works, shall not be less than the wages paid to the employees in the same trades and occupations by private employers engaged in the construction industry. This Article shall also apply to regular employees of the Commonwealth or of a county, town or district, when such employees are employed in the construction, addition to or alteration of public buildings for which special appropriations of more than one thousand dollars are provided. Payments by employers to health and welfare plans, pension plans and supplementary unemployment benefit plans under collective bargaining agreement or understandings between organized labor and employers shall be included for the purpose of establishing minimum wage rates as herein provided.

5.2 LIST OF JOBS, CLASSIFICATIONS, DETERMINATION OF RATES OF PAY AND SCHEDULES

(Statutory reference: M.G.L. Ch. 149, §27): The Commissioner of Labor and Industries shall prepare, for the use of such public officials or public bodies whose duty it shall be to cause public works to be constructed, a list of the several jobs usually performed on various types of public works upon which mechanics and apprentices, teamsters, chauffeurs and laborers are employed. The Commissioner shall classify said jobs, and he may revise such classifications from time to time, as he may deem advisable. Prior to awarding a contract for the construction of public works, said public official or public body shall submit to the Commissioner a list of the jobs upon which mechanics and apprentices, teamsters, chauffeurs and laborers are to be employed, and shall request the Commissioner to determine the rate of wages to be paid on each job. The Commissioner, subject to the provisions of Article 5.1.1, shall proceed forthwith to determine the same, and shall furnish said official or public body with a schedule of such rate or wages as soon as said determination shall have been made. In advertising or calling for bids for said works, the awarding official or public body shall incorporate said schedule in the advertisement or call for bids by an appropriate reference thereto, and shall furnish a copy of said schedule without cost, to any person requesting the same, said schedule shall be made a part of the contract for said works and shall continue to the minimum rate or rates of wages for said employees during the life of the contract. Any person engaged in the construction of said works shall cause a legible copy of said schedule to be kept posted in a conspicuous place at the site of said works during the life of the contract. The aforesaid rates of wages in the schedule of wage rates shall include payments by employers to health and welfare plans, pension plans and supplementary unemployment benefit plans as provided in the previous section, and such payments shall be considered as payments to persons under this section performing work as herein provided. Any employer engaged in the construction of such works that does not make payments to a health and welfare plan, a pension plan and a supplementary unemployment benefit plan, where such payments are included in said rates of wages, shall pay the amount of said payments directly to each employee engaged in said construction. Note: The Owner does not guarantee the accuracy of any schedule of any schedule of wage rates furnished to the Contractor hereunder, and the Contractor shall be responsible for ascertaining the prevailing wages in the area where the work will be performed.

5.3 MAINTENANCE OF EMPLOYMENT RECORDS AND STATEMENT OF COMPLIANCE BY CONTRACTOR & SUBCONTRACTOR

(Statutory reference: M.G.L. Ch.149, §27B):

Every Contractor, Subcontractor or public body engaged in said public works to which Article 5.2 applies shall keep a true and accurate record of all mechanics and apprentices, teamsters, chauffeurs and laborers employed thereon, showing the name, address and occupational classification of each such employee, and shall furnish to the Commissioner of Labor and Industries, upon his request, a copy of said record, signed by the employer or his authorized agent under the penalties of perjury, such records shall be open to inspection by any authorized representative of the Department of Labor and Industries at any reasonable time, and as often as may be necessary.

5.3.1(Statutory reference: M.G.L. Ch.149, §27B) Each such Contractor, Subcontractor or public body shall preserve its payroll records for a period of three (3) years from the date of completion of the contract.

5.3.2(Statutory reference: M.G.L. Ch.149, §27B) Each such Contractor, Subcontractor or public body shall furnish to the Commissioner of labor and Industries within fifteen (15) days after completion of its portion of the work a statement, executed by the Contractor, Subcontractor, or public body who supervises the payment of wages, in the following form:

STATEMENT OF COMPLIANCE

Ι,

(Name and Title of signatory party)

do hereby state:

That I pay or supervise the payment of the persons employed by

(Contractor, subcontractor or public body)

On the

(Name of Project)

and that all mechanics and apprentices, teamsters, chauffeurs and laborers employed on said project have been paid in accordance with wages determined under the provisions of Massachusetts General Laws, Chapter 149, Section 27.

Signature:

1. Title:

The above mentioned copies of payroll records and statements of compliance shall be available for inspection by any interested party filing a written request to the Contractor for such inspections.

5.4 WAGES PAID TO OPERATORS OF TRUCKS AND OTHER EQUIPMENT

(Statutory reference: M.G.L. Ch.149, §27F): Prescribed rates of wages, as determined by the Commissioner of Labor and Industries, shall be paid to the operators of all trucks, vehicles or equipment employed on the Project. Said rates of wages shall be requested of said Commissioner by the Owner and shall be furnished by the Commissioner in a schedule containing the classification of jobs, and the rate of wage to be paid for each job. Said rates of wages shall include payments to health and welfare plans, or, if no such plan is in effect between employers and employees, the amount of such payments shall be paid directly to said operators.

5.5 RESERVE POLICE OFFICERS

(Statutory reference: M.G.L. Ch.149, §34B): The Contractor shall pay to any reserve police officer employed by him in any city or town the prevailing rate of wages paid to regular police officers in such city or town. As identified in Article 12.3.3, the Contractor shall timely submit for payment to the Owner costs incurred in the employment of police officers necessary for the control and safety of the Project Work. Within seven (7) days of receipt of payment by the Owner, the Contractor shall make full payment to every police officer whose work was included for payment in the Contractor's original request for payment.

5.6 EIGHT HOUR DAY AND WORK WEEK

Statutory reference: M.G.L. Ch.149, §§30, 34 and 34A): No laborer, worker, mechanic, foreman or inspector working within the Commonwealth in the employ of the Contractor, Subcontractor or other person doing or contracting to do the whole or part of the work contemplated by the contract, shall be required or permitted to work more than eight (8) hours in any one (1) day or more than forty-eight (48) hours in any one (1) week, or more than six (6) days in any one (1) week, except in cases of extraordinary emergency.

5.7 LODGING

(Statutory reference: M.G.L. Ch.149, §25): Every employee in public work shall lodge, board and trade where and with whom he elects; and no person or his agents or employees under contract with the commonwealth, a county, city or town, or with a department, board, commission or officer acting therefore, for the doing of public work shall directly or indirectly require, as a condition of employment therein, that the employee shall lodge, board or trade at a particular place or with a particular person.

5.8 ACCESS TO CONTRACTOR'S RECORDS

(Executive Order No. 195): The Governor or his designee, the secretary of administration and finance, and the state auditor or his designee shall have the right at reasonable times and upon reasonable notice to examine the books, records and other compilations of data of the Contractor which pertain to the performance and requirements of this contract.

ARTICLE 6. EQUAL EMPLOYMENT OPPORTUNITY & AFFIRMATIVE ACTION

(Statutory reference: M.G.L. Ch.151B; Executive Orders No. 74, No. 116, and No. 143)

6.1 GENERAL:

Article 6 in its entirety applies to all contractors employing six or more persons and to all state or state-assisted contracts for public buildings and public works or for goods and services exceeding the dollar amount set forth in Executive Order No. 116, as amended. The provisions of this Article 6 are intended to comply with the Commonwealth's Supplemental Equal Employment Opportunity Anti-Discrimination and Affirmative Action Program, referred to in Executive Order No. 116 and administered by the Massachusetts Commission Against Discrimination. If no specific percentage has been inserted in Article 6 herein, the applicable minimum percentage provided for in such Supplemental Program shall be deemed to have been so inserted.

6.2 DEFINITIONS:

For purposes of this contract, "minority" refers to Asian-Americans, Blacks, Spanish Surnamed Americans, North American Indians, and Cape Verdeans. "Commission" refers to the Massachusetts Commission Against Discrimination.

6.3 CONTRACTOR REQUIREMENTS AND OBLIGATIONS:

In connection with the performance of work under this contract, the Contractor shall not discriminate against any employee or applicant for employment because of race, color, religious creed, national origin, age or sex. The aforesaid provision shall include, but not be limited to the following: employment upgrading, demotion, or transfer; recruitment advertising; recruitment layoff; termination; rates of pay or other forms of compensation; conditions or privileges of employment; and selection for apprenticeship. The Contractor shall post hereafter in conspicuous places, available for employees and applicants for employment, notices to be provided by the Commission setting forth the provisions of the Fair Employment Practices law of the Commonwealth.

6.3.1 In connection with the performance of work under this contract, the Contractor shall undertake in good faith affirmative action measures designed to eliminate any discriminatory barriers in the terms and conditions of employment on the grounds of race, color, religious creed, national origin, age or sex, and to FY25 Bismore Park Marina Bulkhead Rehabilitation Page **35** of **106**

eliminate or remedy any effects of such discrimination in the past. Such affirmative action shall entail positive and aggressive measures to ensure equal opportunity in the areas of hiring, upgrading, demotion or transfer, recruitment, layoff or termination, rate of compensation, and in-service or apprenticeship training programs. This affirmative action shall include all action required to guarantee equal employment opportunity for all persons, regardless of race, color, religious creed, national origin, age or sex.

6.4 COMPLIANCE WITH EXECUTIVE ORDERS AND LAWS:

The Contractor shall comply with the provisions of Executive Order No. 74, as amended by Executive Order No. 116, dated May 1, 1975, and of M.G.L. Ch.151B, both of which are herein incorporated by reference and made a part of this contract.

6.5 NON-DISCRIMINATION:

The Contractor, in the performance of all work after award, and prior to completion of the contract work, will not discriminate on grounds of race, color, religious creed, national origin, age, sexual orientation or sex in employment practices, in the selection or retention of Subcontractors, or in the procurement of materials and rentals of equipment.

6.6 SOLICITATIONS FOR SUB-CONTRACTS, AND PROCUREMENT OF MATERIALS AND EQUIPMENT:

In all solicitations either by competitive bidding or negotiation made by the Contractor either for work to be performed under a subcontract or for the procurement of materials or equipment, each potential Subcontractor or supplier shall be notified in writing by the Contractor of the Contractor's obligations under this contract relative to non-discrimination and affirmative action.

6.7 EQUAL EMPLOYMENT OPPORTUNITY FOR THE HANDICAPPED:

The Contractor shall comply with the provisions of Executive Order No. 143, entitled "Equal Employment Opportunity for the Handicapped," which is herein incorporated by reference and made a part of this contract. In connection with the performance or work under this contract, the Contractor, Subcontractors and suppliers of goods and services shall not discriminate against the handicapped.

6.8 SUSPENSION OF PAYMENTS:

If the Owner determines after investigation that the Contractor or any Subcontractor is not in compliance with the terms of any provision of Article 6, it may suspend any payment or portion thereof due under the contract until the Contractor demonstrates compliance with the terms of Article 6.

6.8.1 This temporary suspension of payments by the awarding authority is separate from the sanctions set forth in Article 17 herein.

ARTICLE 7. CONTRACTOR'S ACCOUNTING METHODS AND REQUIREMENTS

(Entire provisions of Article 7 are Statutory reference: M.G.L. Ch. 30, §39R):

FY25 Bismore Park Marina Bulkhead Rehabilitation

7.1 GENERAL:

This Article applies to "Contracts" and "Contractors" as defined in Articles 7.1.1 and 7.1.2 below. The words defined herein shall have the meaning stated below whenever they appear in this Article.

7.1.1 "Contractor" means any person, corporation, partnership, joint venture, sole proprietorship, or other entity awarded a contract pursuant to Massachusetts General Laws, Chapter 7C, Sections 44 to 57, inclusive, and any contract awarded or executed pursuant to Chapter 25A, Section 11C, Chapter 30, Section 39M, or Chapter 149, Sections 44A to 44H, inclusive, which is for an amount or estimated amount greater than one hundred thousand dollars (\$100,000.00).

7.1.2 "Contract" means any contract awarded or executed pursuant to Massachusetts General Laws, Chapter 7C, Sections 44 to 57, inclusive, and any contract awarded or executed pursuant to section eleven C of chapter twenty-five A, and any contract awarded or executed pursuant to Chapter 25A, Section 11C, Chapter 30, Section 39M, or Chapter 149, Sections 44A to 44H, inclusive, which is for an amount estimated amount greater than one hundred thousand dollars (\$100,000.00).

7.1.3 "Records" means books of original entry, accounts, checks, bank statements and all other banking documents, correspondence, memoranda, invoices, computer printouts, tapes, discs, papers and other documents or transcribed information of any type, whether expressed in ordinary or machine language.

7.1.4 "Independent Certified Public Accountant" means a person duly registered in good standing and entitled to practice as a certified public accountant under the laws of the place of his residence or principal office and who is in fact independent. In determining whether an accountant is independent with respect to a particular person, appropriate consideration should be given to all relationships between the accountant and that person or any affiliate thereof. Determination of an accountant's independence shall not be confined to the relationships existing in connection with the filing of reports with the awarding authority.

7.1.5 "Audit", when used in regard to financial statements, means an examination of records by an independent certified public accountant in accordance with generally accepted accounting principles and auditing standards for the purpose of expressing a <u>CERTIFIED</u> opinion thereon, or, in the alternative, a qualified opinion or a declination to express an opinion for stated reasons.

7.1.6 "Accountant's Report", when used in regard to financial statements, means a document in which an independent certified public accountant indicates the scope of the audit which he has made and sets forth his opinion regarding the financial statements taken as a whole with a listing of noted exceptions and qualifications, or an assertion to the effect that an overall opinion cannot be expressed. When an overall opinion cannot be expressed. When an overall opinion cannot be expressed the reason therefore shall be stated. An accountant's report shall include as a part thereof a signed statement by the responsible corporate officer attesting that management has fully disclosed all material facts to the independent certified public accountant, and that the audited financial statement is a true and complete statement of the financial condition of the contractor.

7.1.7 "Management," when used herein, means the chief executive officers, partners, principals or other person or persons primarily responsible for the financial and operational policies and practices of the Contractor.

FY25 Bismore Park Marina Bulkhead Rehabilitation

7.1.8 Accounting terms, unless otherwise defined herein, shall have a meaning in accordance with generally accepted accounting principles and auditing standards.

7.1.9 Article 7.1.3 notwithstanding, every agreement or contract awarded or executed pursuant to Massachusetts General Laws, Chapter 7C, Sections 44 to 57, inclusive, or Chapter 25A, Section 11C, and pursuant to Chapter 30, Section 39M or to Chapter 149, Sections 44A through 44H, inclusive, shall provide that:

- a. The Contractor shall make, and keep for at least six (6) years after final payment, books, records, and accounts which in reasonable detail accurately and fairly reflect the transactions and dispositions of the Contractor;
- b. Until the expiration of six (6) years after final payment, the Office of Inspector General, the office of Attorney General, and the Commissioner of Capital Asset Management and Maintenance shall have the right to examine any books, documents, papers or records of the contractor or of his Subcontractors that directly pertain to, and involve transactions relating to, the Contractor or his Subcontractors;
- c. If the agreement is a contract as defined herein, the Contractor shall describe any change in the method of maintaining records or recording transactions which materially affect any statements filed with the awarding authority, including in his description the date of the change and reasons therefore, and shall accompany said description with a letter from the Contractor's independent certified public accountant approving or otherwise commenting on the changes;
- d. If the agreement is a contract as defined herein, the Contractor has filed a statement of management on internal accounting controls as set forth in Section 7.1.10 prior to the execution of the contract, and
- e. If the agreement is a contract as defined herein, the Contractor has filed prior to the execution of the contracts and will continue to file annually, an audited financial statement for the most recent completed fiscal year as set forth in paragraph (d) below.

7.1.10 Every contractor awarded a contract shall file with the awarding authority a statement of management as to whether the system of internal accounting controls of the contractor and its subsidiaries reasonably assures that:

- a. Transactions are executed in accordance with management's general and specific authorization;
- b. Transactions are recorded as necessary:

i. To permit preparation of financial statements in conformity with generally accepted accounting principles, and

ii. To maintain accountability for assets;

- c. Access to assets is permitted only in accordance with management's general or specific authorization; and
- d. The recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action was taken with respect to any difference.

7.1.11 Every Contractor awarded a contract shall also file with the awarding authority a statement prepared and signed by an independent certified public accountant,

stating that he has examined the statement of management on internal accounting controls, and expressing an opinion as to:

- a. Whether the representations of management in response to this paragraph and paragraph (b) above are consistent with the result of management's evaluation of the system of internal accounting controls; and
- b. Whether such representations of management are, in addition, reasonable with respect to transactions and assets in amounts which would be material when measured in relation to the applicant's financial statements.

7.1.12 Records and statements required to be made, kept or filed under the provisions of this section shall not be public records as defined in Massachusetts General Laws, Chapter 4, Section 7, and shall not be open to public inspection; provided, however, that such records and statements shall be made available pursuant to the provisions of Clause (2) of Paragraph (b).

ARTICLE 8. THE CONSULTING ENGINEER

8.1 GENERAL:

The Owner has retained a Consulting Engineer lawfully licensed to perform and provide Consulting Engineering (civil, electrical, mechanical) services in the State of Massachusetts. That person or entity is identified as the Consulting Engineer in the Contract and is referred to throughout the Contract Documents as if singular in number.

8.1.1 Duties, responsibilities and limitations of authority of the Consulting Engineer as set forth in the Contract Documents shall not be restricted, modified or extended without written consent of the Owner.

8.1.2 If the employment of the Consulting Engineer is terminated, the Owner shall solely and promptly employ a successor Consulting Engineer, whose status shall be that of the Consulting Engineer.

8.2 ADMINISTRATION OF THE CONTRACT:

The Consulting Engineer will provide administration of the Contract as described in the Contract Documents and will be an Owner's representative during construction until the date the Consulting Engineer, with the Owner's consent, issues the final Certificate for Payment. The Consulting Engineer will have authority to act on behalf of the Owner only to the extent provided for in the Contract Documents.

8.2.1 The Consulting Engineer will visit the site at intervals appropriate to the stage of construction, to become generally familiar with the progress and quality of the portion of the Work completed, and to determine in general if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. However, the Consulting Engineer will not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. The Consulting Engineer will not have control over, charge of, or be responsible for the construction means, methods, techniques, sequences or procedures, or for the safety precautions and programs in connection with the Work, since these are solely the Contractor's rights and responsibilities under the Contract Documents, except as provided in Article 3.3.

8.2.2 On the basis of the site visits, communication with the Contractor and Subcontractors, and communication with the Owner's Project Manager, the Consulting Engineer will keep the Owner fully informed about the progress and quality of the portion of the Work completed, and report to the Owner:

- a. Known deviations from the Contract Documents and from the most recent construction schedule submitted by the Contractor, and
- b. Defects and deficiencies observed in the Work.

8.3 COMMUNICATIONS AND OTHER ACTIONS FACILITATING CONTRACT ADMINISTRATION:

Except as otherwise provided in the Contract Documents, or when direct communications have been specially authorized, the Owner and Contractor shall endeavor to communicate with each other through the Consulting Engineer about matters arising out of or relating to the Contract. Communications by and with the Consulting Engineer's consultants shall be through the Consulting Engineer. Communications by and with Subcontractors and material and equipment suppliers shall be through the Contractor. Communications by and with separate contractors shall be through the Owner.

8.3.1 Based on the Consulting Engineer's and Owner's Project Manager's evaluations of the Contractor's Applications for Payment, the Consulting Engineer will review and certify the amounts due the Contractor and will issue Certificates for Payment in such amounts for the Owner's approval.

8.3.2 The Consulting Engineer as authorized by the Owner's Project Manager has authority to reject Work that does not conform to the Contract Documents. Whenever the Consulting Engineer considers it necessary or advisable, the Consulting Engineer will have authority to require inspection or testing of the Work in accordance with Article 16, whether or not such Work is fabricated, installed or completed. However, neither this authority of the Consulting Engineer nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Consulting Engineer to the Contractor, Subcontractors, material and equipment suppliers, their agents or employees, or other persons or entities performing portions of the Work.

8.3.3 The Consulting Engineer, pursuant to Article 3.16, will review and approve, or take other appropriate action upon the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Consulting Engineer's action will be taken in accordance with the submittal schedule approved by the Consulting Engineer or, in the absence of an approved Submittal Schedule, with reasonable promptness while allowing sufficient time in the Consulting Engineer's professional judgment to permit adequate review. Review of such submittals is not conducted for the purpose of determining the accuracy and completeness of other details such as dimensions and quantities, or for substantiating instructions for installation or performance of equipment or systems, all of which remain the responsibility of the Contractor as required by the Contract Documents. The Consulting Engineer's review of the Contractor's submittals shall not relieve the Contractor of the obligations under Articles 3.3, 3.5 and 3.16. The Consulting Engineer's review shall not constitute approval of safety precautions or, unless otherwise specifically stated by the Consulting Engineer, or any construction means, methods, techniques, sequences or procedures. The Consulting Engineer's

approval of a specific item shall not indicate approval of an assembly of which the item is a component.

8.3.4 The Consulting Engineer, in consultation with the Owner's Project Manager, will prepare Change Orders and may authorize minor changes in the Work as provided in Article 10.4. The Consulting Engineer will investigate and make determinations and recommendations regarding concealed and unknown conditions as provided in Article 3.11.

8.3.5 The Consulting Engineer and the Owner's Project Manager will conduct inspections to:

- a. Determine the date or dates of Substantial Completion and the date of final completion;
- b. Issue Certificates of Substantial Completion pursuant to Article 12.7;
- c. Receive and forward to the Owner, for the Owner's review and records, written warranties and related documents required by the Contract Documents and assembled by the Contractor pursuant to Article 12.9; and
- d. Issue a final Certificate for Payment pursuant to Article 12.9.

8.3.6 The Consulting Engineer will interpret and decide matters concerning performance pursuant to, and requirements of, the Contract Documents on a verbal or written request of the Owner or by a written request of the Contractor. The Consulting Engineer's response to such requests will be made in writing within any time limits agreed upon or otherwise with reasonable promptness.

8.3.7 Interpretations and decisions of the Consulting Engineer will be consistent with the intent of, and reasonably inferable from the Contract Documents, and will be in writing or in the form of drawings.

8.3.8 The Consulting Engineer will review and respond to Requests for Information about the Contract Documents. The Consulting Engineer's response to such requests will be made in writing with the time limit as defined in Article 8.4 or otherwise with reasonable promptness. If appropriate, the Consulting Engineer will prepare and issue supplemental Drawings and Specifications in response to the Requests for Information.

8.4 TIMELY DECISIONS BY OWNER:

(Statutory reference: M.G.L. C.30, §39P): In every case in which the Contract Documents requires the Owner, any official, or its Consulting Engineer to make a decision in interpretation of the specifications, approval of equipment, material or any other approval, or progress of the work, the decision shall be made promptly and, in any event, no later than thirty days after the written submission for decision. But if such decision requires extended investigation and study, the Owner, the official, or Consulting Engineer shall, within thirty (30) days after the receipt of the submission, give the party making the submission written notice of the reasons why the decision cannot be made within the thirty (30) day period and the date by which the decision will be made.

ARTICLE 9. CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS

9.1 OWNER'S RIGHT TO PERFORM CONSTRUCTION AND TO AWARD SEPARATE CONTRACTS:

FY25 Bismore Park Marina Bulkhead Rehabilitation

The Owner reserves the right to perform construction or operations related to the Project with the Owner's own forces, and to award separate contracts in connection with other portions of the Project or other construction or operations on the Work site under conditions of the Contract identical or substantially similar to these, including those portions related to insurance and waiver of subrogation.

9.1.1 When separate contracts are awarded for different portions of the Project or other construction or operations on the Work site, the term "Contractor" in the Contract Documents in each case shall mean the Contractor who executes each separate Owner-Contractor Contract.

9.1.2 The Owner shall provide for coordination of the activities of the Owner's own forces and of each separate contractor with the Work of the Contractor, who shall cooperate with them. The Contractor shall participate with other separate contractors and the Owner in reviewing their Construction Schedules. The Contractor shall make any revisions to the Construction Schedule deemed necessary after a joint review and mutual agreement. The Construction Schedule shall then constitute the schedule to be used by the Contractor, separate contractors, the Owner and the Consulting Engineer until subsequently revised.

9.2 MUTUAL RESPONSIBILITY:

The Contractor shall afford the Owner and separate contractors' reasonable opportunity for introduction and storage of their materials and equipment and performance of their activities, and shall connect and coordinate the Contractor's construction and operations with theirs as required by the Contract Documents.

9.2.1 If part of the Contractor's Work depends for proper execution or results upon construction or operations by the Owner or a separate contractor, the Contractor shall, prior to proceeding with that portion of the Work, promptly report to the Consulting Engineer apparent discrepancies or defects in such other construction that would render it unsuitable for such proper execution and results. Failure of the Contractor so to report shall constitute an acknowledgment that the Owner's or separate contractors completed or partially completed construction is fit and proper to receive the Contractor's Work except as to defects not then reasonably discoverable.

9.2.2 The Contractor shall reimburse the Owner for costs the Owner incurs that are payable to a separate contractor because of the Contractor's delays, improperly timed activities or defective construction. The Owner shall be responsible to the Contractor for costs the Contractor incurs because of a separate contractor's delays, improperly timed activities, damage to the Work or defective construction.

9.2.3 The Contractor shall promptly remedy damage the Contractor wrongfully causes to completed or partially completed construction or to property of the Owner or separate contractors as provided by Article 13.2.4.

9.2.4 The Owner and each separate contractor shall have the same responsibilities for cutting and patching as described in Article 3.18.

9.3 OWNER'S RIGHT TO CLEAN UP:

If a dispute arises among the Contractor, separate contractors and the Owner as to the responsibility under their respective contracts for maintaining the premises and surrounding

area free from waste materials and rubbish, the Owner may clean up and the Consulting Engineer shall allocate the cost among those responsible parties.

ARTICLE 10. CHANGES IN THE WORK

10.1 GENERAL:

Changes in the Work may be accomplished after execution of the Contract, and without invalidating the Contract, by Change Order, Construction Change Directive or order for a minor change in the Work, subject to the limitations stated throughout the entirety of this Article.

10.1.1 A Change Order shall be based upon agreement among the Owner, Contractor and Consulting Engineer. A Construction Change Directive requires agreement between the Owner and Consulting Engineer, but may or may not be agreed to by the Contractor.

10.1.2 Changes in the Work shall be performed under applicable provisions of the Contract Documents, and the Contractor shall proceed promptly, unless otherwise provided in the Change Order.

10.1.3 No Change Order or Construction Change Directive affecting either Contract Sum or Contract Time shall be granted because of seasonal or abnormal variations in temperature, humidity, or precipitation, which conditions shall be wholly at the risk of the Contractor.

10.2 CHANGE ORDERS:

A Change Order is a written instrument prepared by the Consulting Engineer and signed by the Owner, Contractor and Consulting Engineer stating their agreement upon all of the following:

- a. The change of Work;
- b. The amount of the adjustment, if any, in the Contract Sum; and
- c. The amount of the adjustment, if any, in the Contract Time.

10.2.1 Upon the request of the Owner or the Consulting Engineer, the Contractor shall, without cost to the Owner, submit to the Consulting Engineer in such form as the Consulting Engineer may require, a written Bid for a Change Order in the Work. The Bid shall include the quantity and the cost of each item of material, the number of hours of work and the hourly rate for each class of labor, as well as the description and amounts of all other costs sought by the Contractor to perform the proposed Change. The Contractor shall also furnish to the Consulting Engineer, bona fide Bids from Subcontractors or suppliers for all labor, materials or equipment to be included in such work. The Bid shall be furnished promptly so as not to delay the Work and shall include an estimate of any additional time required to finish the Work.

10.2.2 The Contractor shall also have the authority to initiate Change Order Bids for consideration by the Owner and Consulting Engineer. The submittal of such Bids shall be in accordance with the requirements and format as identified in this Article.

10.2.3 Change Order Bids shall be complete and definitive and the amount of the adjustment in the Contract Sum and the Contract Time, if any, shall be stated in the

Bid for all Work affected by the proposed Change. Once a Change Order is executed, the Contractor shall be required to perform all of the Work required therein (including incidental work and changes to related Work which may be required to complete the Change Order) in accordance with the Contract Documents for the amounts stated in the Change Order.

10.2.4 In the cost or credit to the Owner resulting from a Change in the Work, the value of such cost or credit shall be determined as follows, absent the applicability of a unit price for such item(s) set forth in the Contract:

- a. Costs of labor, including social security, old age and unemployment insurance, fringe benefits and workers' compensation insurance;
- b. Cost of materials, supplies and equipment, including cost of transportation, whether incorporated or consumed;
- c. Rental costs of machinery and equipment, exclusive of hand tools, whether rented from the Contractor or others used directly for the work;
- d. Costs of premiums for all bonds and insurance, and permit fees related to the Work that may not otherwise exempted by state or federal law; and
- e. Fifteen percent (15%) of Subparagraphs A,B,C of this section, for overhead, superintendence and profit, however, if the work to be performed results in a credit to the Owner, an equal percentage for overhead and profit will also be credited.
- f. On work to be performed by a sub-contractor, the Contractor's allowance is to be five percent (5%), applied to a total cost of Sub-Contractor's Work, including his allowance.
- g. On any change involving the Contractor, Subcontractor or any Subsubcontractor, their total cost or credit shall be combined before application of the percentage allowed for the Contractor's overhead and profit.
- h. On work to be performed by a Sub-contractor, The subcontractor's allowance is to be fifteen percent (15%) for his overhead and profit.
- i. On work to be performed by a Sub-subcontractor, the subcontractor's allowance is to be five percent (5%) for his overhead and profit, and the sub-subcontractor's allowance (the one that actually performs the work) is to be fifteen percent (15%) for his overhead and profit, however, that for any such subcontracted work the maximum total fee to be paid by the Owner shall be no greater than 27 percent of the costs incurred by the Subcontractor that actually performs the work.

Additional costs of supervision and field office personnel directly attributable to the change, only if the "change" is approved with an extension of the substantial completion date.

The Owner reserves the right to review all T&M (time and material) and change order labor burden breakdowns for all contractors and subcontractors prior to final acceptance of the cost change.

10.2.5 The amount of credit to be allowed by the Contractor to the Owner for a deletion or change that result in a net decrease in the Contract Sum shall be actual net cost as confirmed by the Consulting Engineer. When both additions and credits covering related Work or substitutions are involved in a change, the allowance for overhead and profit shall be figured on the basis of net increase, if any, with respect to that change.

10.3 CONSTRUCTION CHANGE DIRECTIVES:

A Construction Change Directive is a written order prepared by the Consulting Engineer and signed by the Owner and Consulting Engineer, directing a change in the Work prior to agreement on adjustment, if any, in the Contract Sum or Contract Time, or both. The Owner may issue a Construction Change Directive, without invalidating the Contract, order changes in the Work within the general scope of the Contract consisting of additions, deletions or other revisions, with Contract Sum and Contract Time adjustments made accordingly if necessary.

10.3.1 A Construction Change Directive shall be used in the absence of total agreement on the terms of a Change Order.

10.3.2 If the Construction Change Directive provides for an adjustment to the Contract Sum, the adjustment shall be based on one of the following:

- a. Mutual acceptance of a lump sum properly itemized and supported by sufficient substantiating data to permit evaluation;
- b. Unit prices stated in Contract Documents or subsequently agreed upon;
- c. Cost to be determined in a manner agreed upon by the parties and a mutually acceptable fixed percentage fee; or
- d. As provided for in Article 10.3.6.

10.3.3 If unit prices are stated in the Contract Documents or subsequently agreed upon, and if quantities originally contemplated are materially changed in a proposed Change Order or Construction Change Directive so that application of such unit prices to quantities of Work proposed will cause substantial inequity to the Owner or Contractor, the applicable unit prices shall be equitably adjusted.

10.3.4 Upon receipt of a Construction Change Directive, the Contractor shall promptly proceed with the change in the Work involved and advise the Consulting Engineer of the Contractor's Contract or disagreement with the method, if any, provided in the Construction Change Directive for determining the proposed adjustment in the Contract Sum or Contract Time.

10.3.5 A Construction Change Directive signed by the Contractor indicates the Contractor's agreement therewith, including adjustment in Contract Sum and Contract Time or the method for determining them. Such agreement shall be recorded and formalized as a Change Order based on the final negotiated cost of the change in work and shall be duly signed by the Town Manager and thereby made part of the contract.

10.3.6 If the Contractor does not respond promptly or disagrees with the method for adjustment in the Contract Sum, the Consulting Engineer as authorized by the Owner's Project Manager shall determine the method and the adjustment on the basis of reasonable expenditures and savings of those performing the Work attributable to the change, including, in case of an increase in the Contract Sum, an amount for overhead and profit as set forth in the Contract, or if no such amount is set forth in the Contract, a reasonable amount. In such case, the Contractor shall keep and present, in such form as the Consulting Engineer may prescribe, an itemized accounting together with appropriate supporting data. Unless otherwise provided in the Contract Documents, costs for the purposes of this Article shall be limited to the following:

- a. Costs of labor, including social security, old age and unemployment insurance, fringe benefits and workers' compensation insurance;
- b. Cost of materials, supplies and equipment, including cost of transportation, whether incorporated or consumed;
- c. Rental costs of machinery and equipment, exclusive of hand tools, whether rented from the Contractor or others;
- d. Costs of premiums for all bonds and insurance, permit fees, and taxes related to the Work that may not otherwise exempted by state or federal law; and
- e. Additional costs of supervision and field office personnel directly attributable to the change.
- f. The appropriated level of funding for this project.

10.3.7 The amount of credit to be allowed by the Contractor to the Owner for a deletion or change that results in a net decrease in the Contract Sum shall be actual net cost as confirmed by the Consulting Engineer. When both additions and credits covering related Work or substitutions are involved in a change, the allowance for overhead and profit shall be figured on the basis of net increase, if any, with respect to that change.

10.3.8 Pending final determination of the total cost of a Construction Change Directive to the Owner, the Contractor may request payment for Work completed under the Construction Change Directive in Applications for Payment. The Consulting Engineer will make an interim determination for purposes of monthly Certifications for Payment for those costs and certify for payment the amount that the Consulting Engineer determines, in the Consulting Engineer's professional judgment, to be reasonably justified, subject to the Owner's final approval.

10.4 MINOR CHANGES IN THE WORK:

The Consulting Engineer has authority to order minor changes in the Work not involving adjustment in the Contract Sum or Contract Time, and not inconsistent with the intent of the Contract Documents. Such changes will be effected by written order by the Consulting Engineer and shall be binding on the Owner and Contractor.

ARTICLE 11. TIME

11.1 **DEFINITIONS**:

Unless otherwise provided, Contract Time is the period of time, including authorized adjustments, allotted in the Contract Documents for Substantial Completion of the Work.

11.1.1 The date of commencement of the Work is the date established in the Contract.

11.1.2 The date of Substantial Completion is the date certified by the Consulting Engineer in accordance with Article 12.7.

11.1.3 The term "day" as used in the Contract Documents shall mean calendar day unless otherwise specifically defined.

11.2 **PROGRESS AND COMPLETION:**

Time limits stated in the Contract Documents are of the essence of the Contract. By executing the Contract, the Contractor confirms that the Contract Time is a reasonable period for performing the Work.

11.2.1 The Contractor shall not knowingly, except by agreement or instruction of the Owner in writing, prematurely commence operations on the site or elsewhere prior to the effective date of insurance required by Article 14 to be furnished by the Contractor and Owner. The date of commencement of the Work shall not be changed by the effective date of such insurance.

11.2.2 The Contractor shall proceed expeditiously with adequate forces and shall achieve Substantial Completion within the Contract Time.

11.2.3 If the Consulting Engineer and/or the Owner's Project Manager determine that the amount of Work properly completed is less than ninety percent (90%) of the Work required to be performed pursuant to the Construction Schedule, or that there have been delays to critical paths and that in the Owner's sole discretion, there is reasonable concern that the Project will not be Substantially Complete by the date described in the Contract, the Owner may, in addition to any other remedy it may have, direct the Contractor to take some or all of the following actions at no additional cost:

- a. Increase the number or workers in such quantities and trades as the Consulting Engineer recommends;
- b. Increase the number of working hours per shift, shift per day, working days per week, amount of construction equipment, or any combination of the foregoing in accordance with the Consulting Engineer's and Owner's Project Manager's recommendation;
- c. Re-Schedule activities at the Consulting Engineer's and Owner's Project Manager's discretion.

11.2.4 Nothing contained herein shall limit the Owner's rights to withhold or recover damages for delays caused by the Contractor, or any other remedy to which the Owner is entitled, pursuant to the Contract Documents or by law.

11.3 DELAYS AND EXTENSIONS OF TIME:

(Statutory reference: M.G.L. Ch.30, §39O): Except as otherwise provided by law and by this Paragraph, the Contractor shall not be entitled to damages on account of any hindrances or delays, avoidable or unavoidable; but if such delay is caused by awarding authority, the Contractor may be entitled to an extension of time only in which to complete the work. Said decision is to be determined by the Consulting Engineer in conjunction with the Owner.

11.3.1 The awarding authority may order the Contractor in writing to suspend, delay, or interrupt all or any part of the work for such period of time as it may determine to be appropriate for the convenience of the awarding authority; provided, however, that if there is a suspension, delay or interruption for fifteen (15) days or more due to a failure of the awarding authority to act within the time specified in this contract, the awarding authority shall make an adjustment in the contract price for any increase in the cost of performance of this contract but shall not include any profit to the Contractor on such increase; and provided further, that the awarding authority shall not make any adjustment in the contract price under this provision for any suspension, delay, interruption or failure to act to the extent that such is due to any cause for which this contract provides for an equitable adjustment of the contract price under any other contract provisions.

11.3.2 The Contractor must submit the amount of a claim under this Article to the awarding authority in writing as soon as practicable after the end of the suspension, delay, interruption or failure to act and, in any event, not later than the date of final payment under this contract, and except for costs due to a suspension order, the awarding authority shall not approve any costs in the claim incurred more than twenty days (20) before the Contractor notified the awarding authority in writing of the act or failure to act involved in the claim.

11.3.3 In the event a suspension, delay, interruption or failure to act by the awarding authority increases the cost of performance to any Subcontractor, that Subcontractor shall have the same rights against the Contractor for payment for an increase in the cost of his performance as this Article gives the Contractor against the Owner, but nothing in this Article shall in any way change, modify or alter any other rights which the Contractor or the Subcontractor may have against each other.

11.3.4 Claims arising under this Article shall be made in accordance with applicable provisions of Article 18.

ARTICLE 12. PAYMENTS AND COMPLETION

12.1 CONTRACT SUM:

The Contract Sum is stated in the Contract, and including authorized adjustments, is the total amount payable by the Owner to the Contractor for performance of the Work under the Contract Documents.

12.2 SCHEDULE OF VALUES:

Where the Contract is based on a stipulated sum, the Contractor shall submit to the Consulting Engineer, before the first Application for Payment, a Schedule of Values allocating the entire Contract Sum to the various portions of the Work and prepared in such form and supported by such data to substantiate its accuracy as the Consulting Engineer may require. This schedule must be approved by the Consulting Engineer and Owner, and shall be used as a basis for reviewing the Contractor's Applications for Payment.

12.3 APPLICATIONS FOR PAYMENT

(Statutory reference: M.G.L. Ch.30, §39K): Within fifteen (15) days after receipt from the Contractor, at the place designated by the awarding authority if such a place is so designated, of an Applications for Payment requesting payment of the amount due for the preceding month, the Owner will make a periodic payment to the Contractor for the Work performed during the preceding month and for the materials not incorporated in the Work but delivered and suitably stored at the site (or at some location agreed upon in writing) to which the Contractor has title or to which a Subcontractor has title and has authorized the Contractor to transfer title to the Owner, upon certification by the Contractor that he is the lawful owner and that the materials are free from all encumbrances, but less:

- a. A retention based on its estimate of the fair value of its claims against the Contractor;
- b. A retention for direct payments to Subcontractors based on demands for same in accordance with the provisions of Massachusetts General Laws, Chapter 30, Section 39F, and

c. A retention not exceeding five percent (5%) of the approved amount of the periodic payment.

12.3.1 At least ten (10) days before the date established for each progress payment, the Contractor shall submit to the Consulting Engineer an itemized Application for Payment prepared in accordance with the Schedule of Values, for completed portions of the Work. Such application shall be signed by the appropriate Contractor representative (shall be notarized, if required), and supported by such data substantiating the Contractor's right to payment as the Owner or Consulting Engineer may require, such as copies of requisitions from Subcontractors and material suppliers, and shall reflect retainage if provided for in the Contract Documents.

12.3.2 (Statutory reference: M.G.L. Ch.30, §39K) All Applications for Payment shall be submitted to the Owner, or to its designee as set forth in writing to the Contractor, and the date of receipt by the Owner or its designee shall be marked on the estimate. All Applications for Payment shall contain a separate item listing the amount paid to each Subcontractor and Sub-subcontractor as of the date the periodic estimate is filed. The person making payment for the Owner shall add the daily interest provided for herein to each payment for each day beyond the due date based on the date of receipt marked on the estimate.

12.3.3 Submitted with every Application for Payment shall be the following minimum information relative to the time period reflected in the Application and the corresponding progress of the Work:

- Certified payroll records in accordance with Massachusetts General Laws, Chapter 149, Section 27B, and shall include hours worked by employees of the Contractor, Subcontractors and Sub-subcontractors;
- b. Copies of invoices of any supplies or materials used, or services performed in the completion of Work, or stored materials (see Article 12.3.7), and in particular, invoices for police officers for work performed on police details within the last thirty (30) days; and
- c. Any other forms or information that may be required by the Owner.

12.3.4 As provided in Article 10.3.8, such applications may include requests for payment on account of changes in the Work that have been properly authorized by Change Orders or Construction Change Directives.

12.3.5 Five percent (5%) of all payments due the Contractor for work done and materials furnished, or stored as applicable as herein defined, will be withheld until final completion of the work, as authorized by Massachusetts General Laws, Chapter 30, Section 39G.

12.3.6 Applications for Payment shall not include requests for payment for portions of the Work for which the Contractor does not intend to pay a Subcontractor or material supplier, unless such Work has been performed by others whom the Contractor intends to pay.

12.3.7 In no case shall store materials or equipment, whether stored at the site or at some other Massachusetts location (no other states allowed for storage), be considered for payment unless, at the sole discretion of the Owner, the materials or equipment are ready for and actually scheduled for prompt use. Written notice required for Payment of stored material must be made thirty (30) days in advance of the due

date for the Application for Payment. Payment for materials or equipment stored on site shall require submission and approval of:

- a. A valid invoice including the unit quantity, description of the material or equipment, and cost;
- b. Bill of Sale naming the Owner a purchaser;
- c. Certified statement identifying the exact location of materials or equipment, that the materials or equipment are properly stored and protected, and that it will not be diverted for use and installation on a different project;
- d. Such off-site material or equipment is to be clearly identified and set apart from other material or equipment;
- e. Photographs of said stored material or equipment; and
- f. An All Risk Insurance Certificate for the full invoiced value of the items, with the Owner as a Certificate Holder, Insured Party and Payee in the case of loss, with no deductible attached. Additionally, there shall be a minimum thirty (30) day notice of cancellation to the Certificate Holder.

12.3.8 Should the Owner need to visit the location of the stored material or equipment to further authenticate or verify the acquisition and storage of said material or equipment, the Contractor shall pay all costs associated for the Owner's Project Manager, Consulting Engineer or other representative of the Owner to visit the storage location.

12.3.9 The Contractor warrants that title to all Work covered by an Application for Payment will pass to the Owner no later than the time of payment. The Contractor further warrants that upon submittal of an Application for Payment, all Work for which Certificates for Payment have been previously issued and payments received from the Owner shall, be free and clear of liens, claims, security interests or encumbrances in favor of the Contractor, Subcontractors, material suppliers, or other persons or entities making a claim by reason of having provided labor, materials and equipment relating to the Work.

12.4 CERTIFICATES FOR PAYMENT:

(Statutory reference: M.G.L. Ch.30, §39K): The Owner may make changes in any Application for Payment submitted by the Contractor and the payment due on said Application shall be computed in accordance with the changes so made, but such changes or any requirement for a corrected Application for Payment shall not affect the due date for the periodic payment or the date for the commencement of interest charges on the amount of the periodic payment computed in accordance with the changes made, as provided herein; provided, that the Owner may, within seven (7) days after receipt, return to the Contractor for correction, any Application for Payment which is not in the required form or which contains computations not arithmetically correct and, in that event, the date of receipt of such periodic estimate shall be the date of receipt of the corrected periodic estimate in proper form and with arithmetically correct computations. The date of receipt of a periodic estimate received on a Saturday shall be the first working day thereafter.

12.4.1 The issuance of a Certificate for Payment will constitute a representation by the Consulting Engineer to the Owner, based on the Consulting Engineer's knowledge, information and belief, the Work has progressed to the point indicated and that the quality of the Work is in accordance with the Contract Documents. The foregoing representations are subject to an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, to results of subsequent tests and FY25 Bismore Park Marina Bulkhead Rehabilitation Page **50** of **106** inspections, to correction of minor deviations from the Contract Documents prior to completion and to specific qualifications expressed by the Consulting Engineer. The issuance of a Certificate for Payment will further constitute a representation that the Contractor is entitled to payment in the amount certified.

12.5 DECISIONS TO WITHHOLD CERTIFICATION:

The Consulting Engineer may withhold a Certificate for Payment in whole or in part to the extent reasonably necessary to protect the Owner, if in the Consulting Engineer's opinion the representations to the Owner required by Article 12 cannot be made or complete submittal of information required by Article 12 is not submitted. If the Consulting Engineer is unable to certify payment in the amount of the Application, the Consulting Engineer will notify the Contractor and Owner. If the Contractor and Consulting Engineer cannot agree on a revised amount, the Consulting Engineer will promptly issue a Certificate for Payment for the amount for which the Consulting Engineer is able to make such representations to the Owner. The Consulting Engineer may also withhold a Certificate for Payment previously issued to such extent as may be necessary in the Consulting Engineer's opinion to protect the Owner from loss for which the Contractor is responsible, including loss resulting from acts and omissions described in Article 3.3.1 because of:

- a. Defective Work not remedied;
- b. Third party claims filed or reasonable evidence indicating probable filing of such claims, unless security acceptable to the Owner is provided by the Contractor;
- c. Failure of the Contractor to make payments properly to Subcontractors or for labor, materials or equipment;
- a. Reasonable evidence that the Work cannot be completed for the unpaid balance of the Contract Sum;
- b. Damage to the Owner or a separate contractor;
- c. Reasonable evidence that the Work will not be completed within the Contract Time, and that the unpaid balance would not be adequate to cover actual or liquidated damages for the anticipated delay;
- d. Repeated failure to carry out the Work in accordance with the Contract Documents;
- e. Failure of the Contractors and Subcontractors to comply with mandatory requirements for maintaining Record Drawings. or
- f. Costs incurred by the Owner as described in Article 13.2.4.

12.5.1 When the above reasons for withholding certification are removed or addressed by the Contractor to the satisfaction of the Consulting Engineer, certification will be made for amounts previously withheld.

12.6 PROGRESS PAYMENTS:

After the Consulting Engineer has issued a Certificate for Payment, the Owner shall make payment in the manner and within the time provided in the Contract Documents, and shall so notify the Consulting Engineer.

12.6.1 The Contractor shall pay each Subcontractor and material and equipment suppliers no later than seven (7) days after receipt of payment from the Owner the amount to which the Subcontractor is entitled, reflecting percentages actually retained from payments to the Contractor on account of the Subcontractor's portion of the Work. The Contractor shall, by appropriate agreement with each Subcontractor,

require each Subcontractor to make payments to Sub-subcontractors in an exact manner.

12.6.2 The Consulting Engineer will, on request, furnish to a Subcontractor, if practicable, information regarding percentages of completion or amounts applied for by the Contractor and action taken thereon by the Consulting Engineer and Owner on account of portions of the Work done by each Subcontractor. The Owner may charge a reasonable cost based on the Consulting Engineer's time and expense to prepare the information.

12.6.3 The Owner has the right to request written evidence from the Contractor that the Contractor has properly paid Subcontractors and material and equipment suppliers amounts paid by the Owner to the Contractor for subcontracted Work. If the Contractor fails to furnish such evidence within seven (7) days, the Owner shall have the right to contact Subcontractors to ascertain whether they have been properly paid. Neither the Owner nor Consulting Engineer shall have an obligation to pay or to see the payment of money to a Subcontractor, except as may otherwise be required by law.

12.6.4 Contractor payments to material and equipment suppliers shall be treated in a manner similar to that as required in Articles 12.6.2 through 12.6.4.

12.6.5 A Certificate for Payment, a Progress Payment, or partial or entire use or occupancy of the Project by the Owner shall not constitute acceptance of the Work not in accordance with the Contract Documents.

12.7 SUBSTANTIAL COMPLETION:

Substantial Completion is the stage in the progress of the Work when the Work or designated portion thereof is sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work for its intended use.

12.7.1 (Statutory reference: M.G.L. Ch.30, §39K) Notwithstanding any provisions of this section, at any time after the value of the work remaining to be done is, in the estimation of the Owner, less than 1 percent of the adjusted contract price, or the Owner has determined that the Contractor has substantially completed the Work, the Owner may send to the Contractor by certified mail, return receipt requested, a complete and final list of all incomplete and unsatisfactory work items, including, for each item on the list, a good faith estimate of the fair and reasonable cost of completing such item. The Contractor shall then complete all such work items within 30 days of receipt of such list or before the contract completion date, whichever is later. If the Contractor fails to complete all incomplete and unsatisfactory work items within 45 days after receipt of such items furnished by the Owner or before the contract completion date, whichever is later, subsequent to an additional 14 days' written notice to the Contractor by certified mail, return receipt requested, the Owner may terminate the contract and complete the incomplete and unsatisfactory work items and charge the cost of same to the Contractor and such termination shall be without prejudice to any other rights or remedies the Owner may have under the Contract Documents.

12.7.2 When the Work is substantially complete, the Consulting Engineer will prepare a Certificate of Substantial Completion that shall establish the date of Substantial Completion, shall establish responsibilities of the Owner and Contractor for security,

maintenance, heat, utilities, damage to the Work and insurance, and shall fix the time within which the Contractor shall finish all items on the list accompanying the Certificate. Warranties required by the Contract Documents shall commence on the date of Substantial Completion of the Work, or designated portion thereof unless otherwise provided for in the Certificate of Substantial Completion.

12.7.3 The Certificate of Substantial Completion shall be submitted to the Owner and Contractor for their written acceptance of responsibilities assigned to them in such Certificate. The Owner shall make payment of retainage applying to such Work or designated portion thereof. Such payment shall be adjusted for Work that is incomplete or not in accordance with the requirements of the Contract Documents.

12.8 RESERVED – NOT USED

12.9 FINAL COMPLETION AND FINAL PAYMENT

(Statutory reference: M.G.L. Ch.30, §39K):

After the receipt of a request for final payment from the Contractor, and within sixty-five (65) days after (a) the Contractor fully completes the Work or substantially completes the Work so that the value of the Work remaining to be done is, in the estimate of the awarding authority, less than one percent (1%) of the original contract price, or (b) the Contractor substantially completes the Work, the awarding authority shall pay the Contractor the entire balance due on the contract less (1) a retention based on its estimate of the fair value of its claims against the Contractor and of the cost of completing the incomplete and unsatisfactory items of Work and less (2) a retention for direct payments to Subcontractors based on demands for same in accordance with the provisions of Massachusetts General Laws, Chapter 30, Section 39F, or based on the record of payments by the Contractor to the Subcontractors under this Contract if such record of payment indicates that the Contractor has not paid Subcontractors as provided in said Section 39F. If the awarding authority fails to make payment as herein provided, there shall be added to each such payment daily interest at the rate of three (3) percentage points above the rediscount rate than charged by the Federal Reserve Bank of Boston commencing on the first day after said payment is due and continuing until the payment is delivered or mailed to the contractor; provided, that no interest shall be due, in any event, on the amount due on a periodic estimate for final payment until fifteen (15) days after receipt of such a periodic estimate from the Contractor, at the place designated by the awarding authority if such a place is so designated. The Contractor agrees to pay to each Subcontractor a portion of any such interest paid in accordance with the amount due each Subcontractor.

12.9.1 Neither final payment nor any remaining retained percentage shall become due until the Contractor submits to the Consulting Engineer:

- a. An affidavit that payrolls, bills for materials and equipment, and other indebtedness connected with the Work for which the Owner or the Owner's property might be responsible or encumbered (less amounts withheld by Owner) have been paid or otherwise satisfied, and all certified payrolls have been submitted to the Owner;
- b. A certificate evidencing that insurance required by the Contract Documents to remain in force after final payment is currently in effect and will not be canceled or allowed to expire until at least thirty (30) days prior written notice has been given to the Owner;

- c. A written statement that the Contractor knows of no substantial reason that the insurance will not be renewable to cover the period required by the Contract Documents;
- d. If required by the Owner, other data establishing payment or satisfaction of obligations, such as receipts, releases and waivers of liens, claims, security interests or encumbrances arising out of the Contract, to the extent and in such form as may be designated by the Owner;
- e. All warranties;
- f. As-built drawings;
- g. Completed Owner training sessions for safety and operations; and
- h. All operating manuals.

12.9.2 If a Subcontractor refuses to furnish a release or waiver required by the Owner, the Contractor may furnish a bond satisfactory to the Owner to indemnify the Owner against such lien in full amount thereof. If such lien remains unsatisfied after payments are made, the Contractor shall refund to the Owner all money that the Owner may be compelled to pay in discharging such lien, including all costs and reasonable attorneys' fees.

12.9.3 If, after Substantial Completion of the Work, final completion thereof is materially delayed through no fault of the Contractor or by issuance of Change Orders affecting final completion, and the Consulting Engineer so confirms, the Owner shall, upon application by the Contractor and certification by the Consulting Engineer, and without terminating the Contract, make payment of the balance due for that portion of the Work fully completed and accepted. Such payment shall be made under the terms and conditions governing final payment, except that it shall not constitute a waiver of claims.

12.9.4 The making of final payment shall constitute a waiver of claims by the Owner except those arising from:

- a. Liens, claims, security interests or encumbrances arising out of the Contract and remain unsettled;
- b. Failure of the Work to comply with the requirements of the Contract Documents; or
- c. Terms of special warranties required by the Contract Documents.

12.9.5 The acceptance by the Contractor of the final payment, including the retainage of five percent (5%), shall operate as a release to the Town of all claims and all liabilities to the Contractor for all work done or materials furnished in connection with the Contract, not including replacements of any defective equipment or material or defects in work that arise over the one year maintenance period. Final payment shall be as provided in Massachusetts General Laws, Chapter 30, Section 39G.

12.9.6 Notwithstanding any article within Article 12.10, the payment to the Contractor does not, however, release them or their sureties from any obligation under this Contract.

12.10 DIRECT PAYMENT

(Entire provisions of Article 12.10 are Statutory reference: M.G.L. Ch.30, §39F): Forthwith after the Contractor receives payment on account of an Application for Payment, the

Contractor shall pay to each Subcontractor the amount paid for the labor performed and the materials furnished by that Subcontractor, less any amount specified in any court proceedings barring such payment, and also less any amount claimed due from the Subcontractor by the Contractor.

12.10.1 Not later than sixty-five (65) days after each Subcontractor substantially completes his work in accordance with the Contract Documents, the entire balance due under the subcontract less amounts retained by the Owner as the estimated cost of completing the incomplete and unsatisfactory items of work, shall be due the Subcontractor; and the Owner shall pay that amount to the Contractor. The Contractor shall forthwith pay to the Subcontractor the full amount received from the Owner less any amount specified in any court proceedings barring such payment and also less any amount claimed due from the Subcontractor by the Contractor.

12.10.2 Each payment made by the Owner to the Contractor pursuant to Article 12.10 for the labor performed and the materials furnished by a Subcontractor shall be made to the Contractor for the account of that Subcontractor; and the Owner shall take reasonable steps to compel the Contractor to make each such payment to each such Subcontractor. If the Owner has received a demand for direct payment from a Subcontractor for any amount which has already been included in a payment to the Contractor or which is to be included in a payment to the Contractor for payment to the Subcontractor as provided in this Article, the Owner shall act upon the demand as provided herein.

12.10.3 If, within seventy (70) days after the Subcontractor has substantially completed the subcontract work, the Subcontractor has not received from the Contractor the balance due under the subcontract including any amount due for extra labor and materials furnished to the Contractor, less any amount retained by the Owner as the estimated cost of completing the incomplete and unsatisfactory items of work, the Subcontractor may demand direct payment of that balance from the Owner. The demand shall be by a sworn statement delivered to or sent by certified mail to the Owner, and a copy shall be delivered to or sent by certified mail to the Contractor at the same time. The demand shall contain a detailed breakdown of the balance due under the subcontract and also a statement of the status of completion of the subcontract work. Any demand made after substantial completion of the subcontract work shall be valid even if delivered or mailed prior to the seventieth day after the Subcontractor has substantially completed the subcontract work. Within ten (10) days after the Subcontractor has delivered or so mailed the demand to the Owner and delivered or so mailed a copy to the Contractor, the Contractor may reply to the demand. The reply shall be by a sworn statement delivered to or sent by certified mail to the Owner and a copy shall be delivered to or sent by certified mail to the Subcontractor at the same time. The reply shall contain a detailed breakdown of the balance due under the subcontract including any amount due for extra labor and materials furnished to the Contractor and of the amount due for each claim made by the Contractor against the Subcontractor.

12.10.4 Within fifteen (15) days after receipt of the demand by the Owner, but in no event prior to the seventieth day after substantial completion of the subcontract work, the Owner shall make direct payment to the Subcontractor of the balance due under the subcontract including any amount due for extra labor and materials furnished to the Contractor, less any amount:

 Retained by the Owner as the estimated cost of completing the incomplete or unsatisfactory items of work;

- b. Specified in any court proceedings barring such payment, or
- c. Disputed by the Contractor in the sworn reply; provided, that the awarding authority shall not deduct from a direct payment any amount as provided herein if the reply is not sworn to, or for which the sworn reply does not contain the detailed breakdown required by Article 12.10.3. The Owner shall make further direct payments to the Subcontractor forthwith after the removal of the basis for deductions from direct payments made as provided by this Article.

12.10.5 The Owner shall forthwith deposit the amount deducted from a direct payment as provided in Article 12.10.4 in an interest-bearing joint account in the names of the Contractor and the Subcontractor in a bank in Massachusetts selected by the Owner or agreed upon by the Contractor and the Subcontractor and shall notify the Contractor and the Subcontractor of the date of the deposit and the bank receiving the deposit. The bank shall pay the amount in the account, including accrued interest, as provided in a Contract between the Contractor and the Subcontractor or as determined by decree of a court of competent jurisdiction.

12.10.6 All direct payments and all deductions from demands for direct payments deposited in an interest-bearing account or accounts in a bank pursuant to Article 12.10.5 shall be made out of amounts payable to the Contractor at the time of receipt of a demand for direct payment from a Subcontractor and out of amounts which later become payable to the Contractor and in the order of receipt of such demands from Subcontractors. All direct payments shall discharge the obligation of the Owner to the Contractor to the extent of such payment.

12.10.7 The Owner shall deduct from payments to a Contractor amounts which, together with the deposits in interest-bearing accounts pursuant Article 12.10.5, are sufficient to satisfy all unpaid balances of demands for direct payment received from Subcontractors. All such amounts shall be earmarked for such direct payments, and the Subcontractors shall have a right in such deductions prior to any claims against such amounts by creditors of the Contractor.

12.10.8 If the Subcontractor does not receive payment as provided in Article 12.10, or if the Contractor does not submit an Application for Payment for the value of the labor or materials performed or furnished by the Subcontractor and the Subcontractor does not receive payment for same when due less the deductions provided for in Article 12.10, the Subcontractor may demand direct payment by following the procedure in Article 12.10.4, and the Contractor may file a sworn reply as provided in that same Article. A demand made after the first day of the month following that for which the Subcontractor seeks payment shall be valid even if delivered or mailed prior to the time payment was due on an Application for Payment from the Contractor. Thereafter the Owner shall proceed as provided in this section of Articles.

12.10.9 Any assignment by a Subcontractor of the rights under this section to a surety company furnishing a bond under the provisions of Massachusetts General Laws, Chapter 149, Section 29 shall be invalid. The assignment and subrogation rights of the surety to amounts included in a demand for direct payment which are in the possession of the Owner or which are on deposit pursuant to Article 12.10.6 shall be subordinate to the rights of all Subcontractors who are entitled to be paid under this section and who have not been paid in full.

12.10.10 A Contractor or a Subcontractor shall enforce a claim to any portion of the amount of a demand for direct payment deposited as provided in this Article by a petition in equity in the superior court against the other and the bank shall not be a necessary party. A Subcontractor shall enforce a claim for direct payment or a right to require a deposit as provided in this article by a petition in equity in the superior court against the Owner and the Contractor shall not be a necessary party. Upon motion of any party the court shall advance for speedy trial any petition filed as provided in this paragraph. Massachusetts General Laws, Chapter 231, Sections 59 and 59B shall apply to such petitions. The court shall enter an interlocutory decree upon which execution shall issue for any part of a claim found due pursuant to Sections 59 and 59B and, upon motion of any party, shall advance for speedy trial the petition to collect the remainder of the claim. Any party aggrieved by such interlocutory decree shall have the right to appeal therefrom as from a final decree. The court shall not consolidate for trial the petition of any Subcontractor with the petition of one or more Subcontractors or the same general contract unless the court finds that a substantial portion of the evidence of the same events during the course of construction (other than the fact that the claims sought to be consolidated arise under the same general contract) is applicable to the petitions sought to be consolidated and that such consolidation will prevent unnecessary duplication of evidence. A decree in any such proceeding shall not include interest on the disputed amount deposited in excess of the interest earned for the period of any such deposit. No person except a Subcontractor filing a demand for direct payment for which no funds due the Contractor are available for direct payment shall have a right to file a petition in court of equity against the Owner claiming a demand for direct payment is premature and such Subcontractor must file the petition before the Owner has made a direct payment to the Subcontractor and has made a deposit of the disputed portion as provided in this Article.

12.10.11 In any petition to collect any claim for which a Subcontractor has filed a demand for direct payment the court shall, upon motion of the Contractor, reduce by the amount of any deposit of a disputed amount by the Owner as provided in this Article by any amount held under a trustee writ or pursuant to a restraining order or injunction.

ARTICLE 13. PROTECTION OF PERSONS AND PROPERTY

13.1 SAFETY PRECAUTIONS AND PROGRAMS:

The Contractor shall be responsible for initiating, maintaining and supervising all safety precautions and programs at the Work Site and in connection with the performance of the Contract.

13.1.1 (Statutory reference: M.G.L. Ch.149, S44F(1) The Contractor shall install suitable weather protection in the Work area for all employees, material or equipment that may be affected thereby, and that he shall furnish adequate heat in the Work area so protected during the months of November through March.

13.1.2 (Statutory reference: M.G.L. Ch.149, §129A) Any part of the Work pursuant to this Contract Document in which a trench is to be dug to a depth of five feet or more, except a trench for laying of water pipes dug to a depth of six and one-half feet which will be open less than forty-eight hours, such trench shall be shored and braced in conformity with the rules and regulations for the prevention of accidents in construction operations, as adopted and enforced by the attorney general.

13.2 SAFETY OF PERSONS AND PROPERTY:

The Contractor shall take reasonable precautions for safety of, and shall provide reasonable protection to prevent damage, injury or loss to:

- a. Employees on the Work and other persons who may be affected thereby;
- b. The Work and materials and equipment to be incorporated therein, whether in storage on or off the Work site, under care, custody or control of the Contractor or the Contractor's Subcontractors or Sub-subcontractors; and
- c. Other property at the site or adjacent thereto, such as trees, shrubs, lawns, walks, pavements, roadways, structures and utilities not designated for removal, relocation or replacement in the course of construction.

13.2.1 The Contractor shall comply with and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities bearing on safety of persons or property or their protection from damage, injury or loss.

13.2.2 The Contractor shall erect and maintain, as required by existing conditions and performance of the Contract, reasonable safeguards for safety and protection, including posting danger signs and other warnings against hazards, promulgating safety regulations and notifying owners and users of adjacent sites and utilities.

13.2.3 When use or storage of explosives or other hazardous materials or equipment or unusual methods are necessary for execution of the Work, the Contractor shall exercise utmost care and carry on such activities under supervision of properly qualified or licensed personnel.

13.2.4 The Contractor shall promptly remedy damage any loss (other than damage or loss insured under property insurance required by the Contract Documents) to property referred to in this Article caused in whole or in part by the Contractor, a Subcontractor, a Sub-subcontractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be liable and for which the Contractor is responsible under this Article, except damage or loss attributable to acts or omissions of the Owner or Consulting Engineer or anyone directly or indirectly employed by either of them, or by anyone for whose acts either of them may be liable, and not attributable to the fault or negligence of the Contractor. The foregoing obligations of the Contractor are in addition to the Contractor's obligations under Article 3.20.

13.2.5 The Contractor shall designate a responsible member of the Contractor's organization at the Work Site whose duty shall be the prevention of accidents. This person shall be the Contractor's Project Manager/Superintendent unless otherwise designated by the Contractor in writing to the Owner and Consulting Engineer.

13.2.6 The Contractor shall not permit any part of the Work or Work site to be loaded so as to cause damage or create an unsafe condition.

13.3 INJURY OR DAMAGE TO PERSON OR PROPERTY:

If Contractor shall notify the Town in writing of any injury or damage to person or property occurring at the Work Site, whether or not such injury or damage is insured, within twenty-

one (21) days after discovery. The notice shall provide sufficient detail to enable the Town to investigate the matter.

13.3.1 The Contractor shall be responsible for the adequate support and safety of all scaffolding, staging and hoisting equipment and for temporary shoring and bracing.

13.3.2 The Contractor shall furnish approved hard hats, safety colored vests, eye protection and other personal protective equipment as required, approved first aid supplies, name of first aid attendant and a posted list of emergency facilities and telephone numbers.

13.3.3 The Contractor shall take immediate action to correct any dangerous conditions revealed or made known to it.

13.3.4 No unauthorized visitors shall be allowed on the Work site without the permission from the Owner, Contractor, Consulting Engineer or Owner's Project Manager.

13.3.5 On projects in which the Work is performed adjacent to a building in which any portion of said building is occupied or used by employees or the public, the Contractor shall take any and all measures to protect them from harm as they pass through or otherwise utilize approved areas of the Work site.

13.3.6 The Contractor shall comply with the requirements of the Occupational Safety and Health Act (OSHA), the OSHA 10 hour safety certification program for workers and the Construction Act of 1969, which are incorporated herein by reference and all standards and regulations promulgated by the governmental and regulatory bodies responsible for administration thereof. The Contractor shall be responsible for compliance with such Acts, standards and regulations by its officers, agents, employees, Subcontractors, Sub-subcontractors, and material and equipment suppliers. The Contractor shall indemnify and hold harmless the Owner and its employees and the Consulting Engineer from any and all fines, costs and expenses, including, but not limited to reasonable attorneys' fees incurred by the Owner, the Owner's Project Manager and Consulting Engineer due to violation of such Acts, standards and regulations for all Contractor and Subcontractor employees must be held on the Work site.

13.4 HAZARDOUS MATERIALS:

The Contractor is responsible for compliance with any requirements included in the Contract Documents regarding hazardous materials. If the Contractor encounters a hazardous material or substance not addressed in the Contract Documents, and if reasonable precautions will be inadequate to prevent foreseeable human exposure, bodily injury or death to persons resulting from a material or substance, including but not limited to asbestos or polychlorinated biphenyl (PCB), encountered at the Work site by the Contractor, the Contractor shall, upon recognizing the condition, immediately stop Work in the affected area and immediately report the condition to the Owner and Consulting Engineer in writing.

13.4.1 Upon receipt of the Contractor's written notice, the Owner shall obtain the services of a licensed laboratory to verify the presence or absence of the material or substance reported by the Contractor and, in the event such hazardous material or substance is found to be present, to cause it to be rendered harmless. Unless

otherwise required by the Contract Documents, the Owner shall furnish in writing to the Contractor and Consulting Engineer the name and qualifications of the person or entity that shall be responsible for the removal or safe containment of such hazardous material or substance. The Contractor and Consulting Engineer will promptly reply to the Owner in writing stating whether or not either has reasonable objection to the person or entity proposed by the Owner. If either has an objection, the Owner shall propose another to whom the Contractor and Consulting Engineer have no reasonable objection. When the hazardous material or substance has been removed or contained, Work in the affected area shall resume upon written notice by the person or entity that the affected area is safe to resume the Work. By Change Order, the Contract Time shall be extended appropriately and the Contract Sum may be increased in the amount of the Contractor's reasonable additional costs of shutdown, delay and start-up.

13.4.2 The Owner shall not be responsible under this Article 13.4 for materials or substances the Contractor brings to the Work that is necessary to carry out the Work.

13.4.3 The Contractor shall indemnify the Owner for the cost and expense the Owner incurs resulting from:

- a. Remediation of a material or substance the Contractor brings to the site and negligently handles; or
- b. Contractor's failure to perform its obligations under Article 13.4, except to the extent that the cost and expense are due to the Owner's fault or negligence.

13.5 **EMERGENCIES:**

In an emergency affecting safety of persons or property, the Contractor shall act at their discretion to prevent threatened bodily injury, damage or loss. The Owner shall be notified immediately of any on-site emergency. Changes to the Contract Sum or Contract Time claimed by the Contractor on account of an emergency shall be determined pursuant to Articles 10 and 18.

ARTICLE 14. INSURANCE AND BONDS

14.1 INSURANCE GENERALLY

14.1.1 The Contractor shall purchase and maintain insurance of the type and limits listed in this Article with respect to the operations as well as the completed operations of this Contract. This insurance shall be provided at the Contractor's expense and shall be in full force and effect for the full term of the Contract or for such longer period as this Article requires. Failure of the Contractor to obtain and maintain in force any insurance policy required herein may be deemed by the Owner as a material breach of this Contract, and may constitute sufficient grounds for immediate termination of the Contract.

14.1.2 All policies shall be written on an occurrence basis and be issued by companies lawfully authorized to write that type of insurance under the laws of the Commonwealth with a financial strength rating of A- or better as assigned by AM Best Company, or an equivalent rating assigned by a similar rating agency acceptable to the Awarding Authority, or otherwise acceptable to the Awarding Authority.

14.1.3Contractor shall submit three originals of each certificate of insurance,
acceptable to the Awarding Authority, simultaneously with the execution of this
Contract. Certificates shall show each type of insurance, insurance company, policyFY25 Bismore Park Marina Bulkhead RehabilitationPage 60 of 106

number, amount of insurance, deductibles and/or self-insured retentions, and policy effective and expiration dates. Certificates shall show the Awarding Authority, the Owner and anyone else the Awarding Authority requests as an additional insured as to all policies of liability insurance. Certificates shall specifically note the following:

- that the automobile liability, umbrella liability and pollution liability policies include the Awarding Authority as an additional insured;
- that all policies include the coverage and endorsements in accordance with the terms and conditions as required by this construction contract;
- that the Builders' Risk or Installation Floater is on an all risk basis including earthquake and flood, and includes the Awarding Authority as a named insured or loss payee as their interests may appear; and
- that none of the coverages shall be cancelled, terminated, or materially modified unless and until 30 days prior notice is given in writing to the Awarding Authority.

14.1.4 Contractor shall submit updated certificates prior to the expiration of any of the policies referenced in the certificates so that the Awarding Authority shall at all times possess certificates indicating current coverage.

14.1.5 If requested by the Town, the Contractor shall file one certified complete copy of all policies and endorsements with the Awarding Authority. If the Awarding Authority is damaged by the Contractor's failure to maintain such insurance and to comply with the terms of this Article, then the Contractor shall be responsible for all costs and damages to the Awarding Authority attributable thereto.

14.1.6 The Owner's authority to review certificates and policies of insurance, and its decision to raise or not to raise any objections about those certificates and policies shall not in any way give rise to any duty or responsibility on the part of the Owner to exercise this authority for the benefit of the Contractor, any Subcontractor, Subsubcontractor, material or equipment supplier, or any other party.

14.1.7 Termination, cancellation, or material modification of any insurance required by this Contract, whether by the insurer or the insured, shall not be valid unless written notice thereof is given to the Awarding Authority at least thirty days prior to the effective date thereof, which shall be expressed in said notice.

14.1.8 The Contractor is responsible for the payment of any and all deductibles under all of the insurance required below. The Awarding Authority shall not in any instance be responsible for the payment of deductibles, self-insured retentions, or any portion thereof.

14.1.9 The Contractor shall require each Subcontractor and Sub-subcontractor to purchase and maintain insurance in the amount and type at least equal to the limits stated in Article 14, and the provisions of Article 14 shall be equally applicable to them. The Contractor shall provide certificates of such insurance to the Owner prior to such Subcontractor or Sub-subcontractor performing work at the Work Site.

14.2 CONTRACTOR'S COMMERCIAL GENERAL LIABILITY

14.2.1 The Contractor shall purchase and maintain broad form general liability coverage on the ISO form CG 00 01 or equivalent, including products and completed operations, on an occurrence basis. The form must be amended to state that the

aggregate limit applies on a per location/project basis. The policy shall provide the following minimum coverage to protect the Contractor from claims with respect to the operations performed by Contractor and any employee, subcontractor, or supplier, or by anyone for whose acts they may be liable unless a higher coverage is specified in Exhibit A to the Owner - Contractor Agreement, in which case the Contractor shall provide the additional coverage:

Bodily Injury & Property Damage Products & Completed Operations Personal & Advertising Injury Medical Expenses \$1,000,000 each occurrence\$3,000,000 general aggregate per project\$3,000,000 annual aggregate\$1,000,000 each occurrence\$5,000

14.2.2 This policy shall include coverage relating to explosion, collapse, and underground property damage.

14.2.3 This policy shall include contractual liability coverage.

14.2.4 The completed operations coverage shall be maintained for a period of three (3) years after Substantial Completion and acceptance by the Awarding Authority. The Contractor shall provide renewal certificates of insurance to the Awarding Authority as evidence that this coverage is being maintained.

14.2.5 If the Work includes work to be performed within 50 feet of a railroad, any exclusion for liability assumed under contract for work within 50 feet of a railroad shall be deleted.

14.2.6 This policy shall include the Awarding Authority and anyone else requested by the Awarding Authority as an additional insured via endorsements CG 20 10 for ongoing operations and CG 20 37 for completed operations This policy shall be primary and non-contributory with respect to any other insurance available to additional insureds.

14.2.7 The policy shall include endorsement CG 24 04, a Waiver of Subrogation in favor of the Awarding Authority.

14.3 AUTOMOBILE LIABILITY

14.3.1 The Contractor shall purchase and maintain the following minimum coverage with respect to the operations of any owned, non-owned, and hired vehicles including trailers used in the performance of the work, unless a higher coverage is specified in Exhibit A to the Owner - Contractor Agreement, in which case the Contractor shall provide the additional coverage:

Bodily Injury & Property Damage \$1,000,000 combined single limit per accident

14.3.2 The policy shall include a CA 99 48 Broadened Pollution Endorsement. If specified in the Contract Documents, the Contractor, if hauling contaminants and/or pollutants, must adhere to Sections 29 and 30 of the Motor Carrier Act of 1980, which shall include coverage Form MCS-90.

14.3.3 The policy shall include the Awarding Authority as an additional insured.

14.3.4 The policy shall contain a Waiver of Subrogation in favor of the Awarding Authority.

14.4 CONTRACTOR'S POLLUTION LIABILITY

14.4.1 The Contractor shall purchase and maintain coverage for bodily injury and property damage resulting from liability arising out of pollution related exposures such as asbestos abatement, lead paint abatement, tank removal, removal of contaminated soil, etc. The insurance policy shall cover the liability of the Contractor during the process of removal, storage, transport and disposal of hazardous waste and contaminated soil and/or asbestos abatement. The policy shall include coverage for on-Site and off-Site bodily injury and loss of, damage to, or loss of use of property, directly or indirectly arising out of the discharge, dispersal, release or escape of smoke, vapors, soot, fumes, acids, alkalis, toxic chemicals, liquids or gas, waste materials or other irritants, contaminants or pollutants into or upon the land, the atmosphere or any water course or body of water, whether it be gradual or sudden and accidental. The policy shall also include defense and clean-up costs. The Awarding Authority shall be named as an additional insureds and coverage must be on an occurrence basis. The amount of coverage shall be as follows unless a higher amount is specified in the Contract Documents, in which case the Contractor shall provide the additional coverage:

Limit of liability \$1,000,000 per occurrence \$3,000,000 aggregate

14.5 WORKER'S COMPENSATION

14.5.1 The Contractor shall provide the following coverage in accordance with M.G.L. c.149 §34A and c.152 as amended, unless a higher coverage is specified in Exhibit A to the Owner - Contractor Agreement, in which case the Contractor shall provide the higher coverage:

Worker's Compensation	Statutory limits
Employer's Liability	\$ 1,000,000 each accident
	\$ 1,000,000 disease per employee
	\$ 1,000,000 disease policy aggregate

14.5.2 If specified in the Contract Documents the policy must be endorsed to cover United States Longshoremen & Harborworkers Act (USLHW), or Maritime Liability.

14.5.3 The policy shall contain a Waiver of Subrogation in favor of the Awarding Authority.

14.6 UMBRELLA COVERAGE

14.6.1 The Contractor shall provide Umbrella Coverage in a form at least as broad as primary coverages required by Sections 2, 3 and 5 of this Article in the following amount unless a higher amount is specified in the Contract Documents, in which case the Contractor shall provide the higher amount:

Contract Price:	Limit of Liability:
Under \$1,000,000	\$2,000,000 per occurrence
\$1,000,001 \$5,000,000	\$5,000,000 per occurrence
\$5,000,001 \$10,000,000	\$10,000,000 per occurrence
\$10,000,001 and over	\$25,000,000 per occurrence

14.7 ADDITIONAL TYPES OF INSURANCE

The Contractor shall provide such other types of insurance as may be required by the Contract Documents.

14.8 OWNER'S LIABILITY INSURANCE:

The Owner shall be responsible for purchasing and maintaining the Owner's usual liability and property insurance during the term of the Work, or at such times as the Owner has made a determination of Substantial Completion in a portion of or the total Work as identified in Article 12.7.

14.8.1 The Owner shall purchase and maintain boiler and machinery insurance required by the Contract Documents, or by law, which shall specifically cover such insured objects during installation and until final acceptance by the Owner. This insurance shall include interests of the Owner, Contractor, Subcontractor and Subsubcontractors in the Work, and the Owner and Contractor shall be named insureds.

14.9 **PROPERTY INSURANCE:**

Property insurance shall be on an "all-risk" or equivalent policy form and shall include, without limitation, insurance against the perils of fire (with extended coverage) and physical loss or damage including, without duplication of coverage, theft, vandalism, malicious mischief, collapse, earthquake, flood, windstorm, false work, testing and start-up, temporary buildings and debris removal, including demolition occasioned by enforcement of any applicable legal requirements, and shall cover reasonable compensation for Consulting Engineer's and Contractor's services and expenses required as a result of such insured loss.

14.9.1 This property insurance shall cover portions of the Work stored off the site, and also portions of the Work in transit.

14.9.2 If the Contractor requests in writing that insurance for risks other than those described herein or other special causes of loss be included in the property insurance policy, the Owner shall, if possible, include such insurance, and the cost thereof shall be charged to the Contractor by an appropriate Change Order.

14.9.3 If, during the Work construction period, the Owner insures properties, real or personal, or both, at or adjacent to the Work site by property insurance under policies separate from those insuring the Project, or if after Final Payment, property insurance is to be provided on the completed Project through a policy or policies other than those insuring the Project during the construction period, the Owner shall waive all rights in accordance with the terms of Article 14.4 for damages caused by fire or other causes of loss covered by this separate property insurance.

14.9.4Before an exposure to loss may occur, upon request by the Contractor, the
Owner shall file with the Contractor a copy of each policy that includes insuranceFY25 Bismore Park Marina Bulkhead RehabilitationPage 64 of 106

coverages required by Article 14.4. Each policy shall contain all generally applicable conditions, definitions, exclusions and endorsements related to this Project. Each policy shall contain a provision that the policy will not be canceled or allowed to expire, and that its limits will not be reduced until at least thirty (30) days prior written notice has been given to the Contractor.

14.9.5 The Owner shall have the power to adjust and settle with its insurer any loss for which it has obtained insurance pursuant to this Article.

14.9.6 Upon the occurrence of an insured loss, the Owner and the Contractor shall cooperate with each other and with each other's insurer in the submission of claims and related information, and the distribution of any insurance proceeds. If, after such a loss, no special agreement is made, replacement of damaged Work shall be covered by an appropriate Change Order or Change Directive.

ARTICLE 15. UNCOVERING AND CORRECTION OF WORK

15.1 UNCOVERING WORK:

If a portion of the Work is covered contrary to the Consulting Engineer's request or to requirements specifically expressed in the Contract Documents, it must, if requested in writing by the Engineer Consulting Engineer, be uncovered for the Consulting Engineer's examination and be replaced at the Contractor's time and expense without change to either the Contract Sum or Contract Time.

15.1.1 If a portion of the Work has been covered that the Consulting Engineer has not specifically requested to examine prior to its being covered, the Consulting Engineer, with the Owner's Contract, may request to see such Work and it shall be uncovered by the Contractor. If such Work is in accordance with the Contract Documents, costs of uncovering and replacement shall, by appropriate Change Order, be at the Owner's expense. If such Work is not in accordance with the Contract Documents, such costs and the cost of correction shall be at the Contractor's expense, unless the condition was caused by the Owner or a separate contractor, in which event the Owner shall be responsible for payment of such costs. Depending on the cause of the faulty Work, change to either the Contract Sum or Contract Time may be considered.

15.2 CORRECTION OF WORK:

The Contractor shall promptly correct Work rejected by the Consulting Engineer or failing to conform to the requirements of the Contract Documents, whether discovered before or after Substantial Completion, and whether or not fabricated, installed or completed. Costs of correcting such rejected Work, including additional testing and inspections, the costs of uncovering and replacement, and compensation for the Consulting Engineer's services and expenses made necessary thereby, shall be at the Contractor's expense without change to either the Contract Sum or Contract Time.

15.2.1 In addition to the Contractor's obligations under Article 3.5, if, within one (1) year after the date of Substantial Completion of the Work or designated portion thereof, or after the date for commencement of warranties established under Article 12.9, or by terms of an applicable special warranty required by the Contract Documents, any of the Work is found to be not in accordance with the requirements of the Contract Documents, the Contractor shall correct it promptly after receipt of written

notice from the Owner to do so unless the Owner has previously given the Contractor a written acceptance of such condition. The Owner shall give such notice promptly after discovery of the condition. During the one (1) year period for correction of Work, if the Owner fails to notify the Contractor and give the Contractor an opportunity to make the correction, the Owner waives the rights to require correction by the Contractor and to make a claim for breach of warranty. If the Contractor fails to correct nonconforming Work within a reasonable time during that period after receipt of notice from the Owner or Consulting Engineer, the Owner may correct it in accordance with Article 2.6.

15.2.2 The one (1) year period for correction of Work shall be extended with respect to portions of Work first performed after Substantial Completion by the period of time between Substantial Completion and the actual completion of that portion of the Work.

15.2.3 The one (1) year period for correction of Work shall not be extended by corrective Work performed by the Contractor after Substantial Completion for correction work necessary as a result of damage caused by the Owner or a separate contractor.

15.2.4 The Contractor shall remove from the Work site portions of the Work that are not in accordance with the requirements of the Contract Documents and are neither corrected by the Contractor or accepted by the Owner.

15.2.5 The Contractor shall bear the cost of correcting destroyed or damaged construction, whether completed or partially completed, of the Owner or separate contractors caused by the Contractor's correction or removal of Work that is not in accordance with the requirements of the Contract Documents.

15.2.6 Nothing contained in the entirety of Article 15 shall be construed to establish a period of limitation with respect to other obligations the Contractor has under the Contract Documents. Establishment of the one (1) year period for correction of Work as described in this Article relates only to the specific obligation of the Contractor to correct the Work, and has no relationship to the time within which the obligation to comply with the Contract Documents may be sought to be enforced, nor to the time within which proceedings may be commenced to establish the Contractor's liability with respect to the Contractor's obligations other than specifically to correct the Work.

15.3 ACCEPTANCE OF NON-CONFORMING WORK:

If the Owner prefers to accept Work that is not in accordance with the requirements of the Contract Documents, the Owner may do so instead of requiring its removal and correction, in which case the Contract Sum will be reduced as appropriate and equitable. Such adjustment shall be effected whether or not Final Payment has been made. Owner's acceptance of Work under this provision must be in writing, signed by the Owner's Project Manager.

ARTICLE 16. TESTS, INSPECTIONS AND APPROVALS

16.1 TESTS, INSPECTIONS AND APPROVALS:

Tests, inspections and approvals of portions of the Work shall be made as required by the Contract Documents and by applicable laws, statutes, ordinances, codes, rules and regulations or lawful orders of public authorities. Unless otherwise provided, the Contractor shall make arrangements for such tests, inspections and approvals with an independent

testing laboratory or entity acceptable to the Owner, or with the appropriate public authority, and shall bear all related costs of tests, inspections and approvals. The Contractor shall give the Consulting Engineer timely notice of when and where tests and inspections are to be made so that the Consulting Engineer may be present for such procedures. The Owner shall bear costs of:

- a. Tests, inspections or approvals that do not become requirements until after bids are received or negotiations concluded; and
- b. Tests, inspections or approvals where applicable laws or regulations prohibit the Owner from delegating their cost to the Contractor.

16.1.1 If the Consulting Engineer, Owner or public authorities having jurisdiction determine that portions of the Work require additional testing, inspection or approval not included under Article 16, the Consulting Engineer will, upon written authorization from the Owner, instruct the Contractor to make arrangements for such additional testing, inspection or approval by an individual or entity acceptable to the Owner, and the Contractor shall give timely notice to the Consulting Engineer of when and where tests and inspections are to be made so that the Consulting Engineer may be present for such procedures. Such costs, except as provided in Article 16.1.3, shall be at the Owner's expense.

16.1.2 If such procedures for testing, inspection or approval under Articles 16.1 and 16.1.1 reveal failure of the portions of Work to comply with requirements established by the Contract Documents, all costs, including the cost of re-testing made necessary by such failure, including those of repeated procedures and compensation for the Consulting Engineer's services and expenses, shall be at the Contractor's expense.

16.1.3 The Contractor shall obtain and deliver promptly to the Consulting Engineer any permits, certificates of final inspection of any part of the Contractor's Work, and operating permits for any mechanical apparatus (if applicable). Receipt of such permits or certificates by the Consulting Engineer shall be a condition precedent to determining that the Work is Substantially Complete.

16.1.4 If the Consulting Engineer is to observe tests, inspections or approvals required by the Contract Documents, the Consulting Engineer will do so promptly, and where practicable, at the normal place of testing.

16.1.5 Tests or inspections conducted pursuant to the Contract Documents shall be made promptly to avoid unreasonable delay in the Work.

ARTICLE 17. TERMINATION OR SUSPENSION OF THE CONTRACT

17.1 TERMINATION BY THE OWNER FOR CAUSE:

The Owner may terminate the Contract if the Contractor:

- a. Repeatedly refuses or fails to supply enough properly skilled workers or proper materials;
- b. Fails to make payment to Subcontractors for materials or labor in accordance with the respective Contracts between the Contractor and Subcontractors;
- c. Repeatedly disregards applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of a public authority;
- d. Otherwise is guilty of substantial breach of a provision of the Contract Documents;

- e. Assigns any portion of the Contract between the Contractor and the Owner, or otherwise assigns any portion of the Contractor's responsibilities to a third party without the express approval of the Owner; or
- f. The Contractor files for bankruptcy.

17.1.1 When any of the above reasons exist, the Owner may without prejudice to any other rights or remedies of the Owner, and after giving the Contractor and the Contractor's Surety, if any, seven (7) days written notice, terminate employment of the Contractor and may, subject to any prior rights of the Surety:

- a. Exclude the Contractor from the site and take possession of all materials, equipment, tools and construction equipment and machinery thereon owned by the Contractor;
- b. Accept assignment of subcontract pursuant to Article 4.4; and
- c. Finish the Work by whatever reasonable method the owner may deem expedient. Upon written request of the Contractor, the Owner shall furnish to the Contractor a detailed accounting of the costs incurred by the Owner in finishing the Work.

17.1.2 When the Owner terminates the Contract for one of the reasons stated in Article 17, the Contractor shall not be entitled to receive further payment until the Work is finished.

17.1.3 If the unpaid balance of the Contract Sum exceeds costs of finishing the Work, including compensation for the Consulting Engineer and Owner's Project Manager services and expenses made necessary thereby, and other damages incurred by the Owner and not expressly waived, such excess shall be paid to the Contractor. If such costs and damages exceed the unpaid balance, the Contractor shall pay the difference to the Owner. The amount to be paid to the Contractor or Owner, as the case may be, application, and this obligation for payment shall survive termination of the Contract.

17.2 TERMINATION BY OWNER FOR EXTRAORDINARY DELAY:

Should the Contractor's performance of the Work be delayed for ninety (90) or more days pursuant to any provision in Article 11.3, the Owner shall have the authority to terminate the Contract.

17.2.1 In the case of such termination due to delay, the Contractor shall only be entitled to receive payment for Work executed to the date of such termination.

17.3 TERMINATION BY THE OWNER FOR CONVENIENCE:

The Owner may, at any time, terminate the Contract for the Owner's convenience and without cause.

17.3.1 Upon receipt of written notice from the Owner of such termination for the Owner's convenience, the Contractor shall:

a. Cease operations as directed by the Owner in the notice;

b. Take actions necessary, or that the Owner may direct, for the protection and preservation of the Work; and

c.Except for Work directed to be performed prior to the effective date of termination stated in the notice, terminate all existing subcontracts and purchase orders and enter into no further subcontracts and purchase orders.

17.3.2 In the case of such termination for the Owner's convenience, the Contractor shall be entitled to receive payment for Work executed, and costs incurred by reason of such termination, along with reasonable overhead and profit on the Work not executed.

17.3.3 For any termination under Article 17.1 or 17.2 which is later determined to be unwarranted, without cause or not in compliance with the terms of the Contract Documents, such termination shall be deemed a termination for convenience under this section.

17.4 SUSPENSION BY THE OWNER FOR CONVENIENCE:

The Owner may, without cause, order the Contractor in writing to suspend, delay or interrupt the Work in whole or in part for such period as the Owner may determine.

17.4.1 The Contract Sum and Contract Time shall be adjusted for increases thereto caused by suspension, delay or interruption as described in Article 17.2. Adjustment of the Contract Sum shall include reasonable overhead and profit as may be defined in the Contract Documents. No adjustment shall be made to the extent that:

- a. Performance is, was or would have been so suspended, delayed or interrupted by another cause for which the Contractor is responsible; or
- b. An equitable adjustment is made or denied under another provision of the Contract.

ARTICLE 18. CLAIMS AND DISPUTES

18.1 DEFINITION OF CLAIM:

A claim is a demand or assertion by one of the parties seeking, as a matter of right, payment of money or other relief with respect to the terms of the Contract. The term "Claim" also includes other disputes and matters in question between the Owner and the Contractor arising out of or relating to the Contract. The responsibility to substantiate Claims shall rest with the party making the Claim.

18.2 NOTICE OF CLAIMS:

Claims by either the Owner or Contractor must be initiated by written notice, as provided herein, to the other party. Claims by either party must be initiated within twenty-one (21) days after occurrence of the event giving rise to such Claim or within twenty-one (21) days after the claimant first recognizes the condition giving rise to the Claim, whichever is later.

18.2.1 The failure of either party to provide written notice in strict accordance with this Article shall result in either party having waived its claim. Written notice submitted by the Contractor must including pricing of the Claim in accordance with Article 10.

18.3 CONTINUING CONTRACT PERFORMANCE:

Pending final resolution of a Claim, except as otherwise agreed in writing or as provided in Article 12.7 and Article 17 in its entirety, the Contractor shall proceed diligently with performance of the Contract and the Owner shall continue to make payments in accordance with the Contract Documents.

FY25 Bismore Park Marina Bulkhead Rehabilitation

18.4 CLAIMS FOR ADDITIONAL COST:

If the Contractor wishes to make a Claim for an increase in the Contract Sum, written notice as provided herein shall be given before proceeding to execute the Work. Prior notice is not required for Claims relating to an emergency endangering life or property arising under Article 13.4 or 13.5.

18.4.1 The Contractor hereby acknowledges that the Owner has the contractual right to delay the Work. Such right may not be exercised unreasonably. In addition, the Contractor shall not be entitled to additional compensation as a result of delay, even if caused by the Owner or those for whom the Owner is responsible. The Contractor's sole remedy for any delay is an extension of time, notwithstanding the above.

18.5 CLAIMS FOR ADDITIONAL TIME:

If the Contractor wishes to make a Claim for an increase in the Contract Time, written notice as provided herein shall be given. The Contractor's Claim shall include an estimate of effect of delay on progress of the Work. In the case of a continuing delay, only one Claim is necessary.

18.5.1 If adverse weather conditions are the basis for a Claim for additional time, such Claim shall be documented by data substantiating that weather conditions were abnormal for the period of time, could not have been reasonably anticipated and had an adverse effect on the scheduled Work.

18.5.2 No increase in Contract Time will be allowed for Work which is delayed as a result of the Contractor's failure to timely submit, revise or re-submit shop drawings, product data and/or samples.

18.5.3 Any claim or dispute outstanding, or otherwise not settled between the Owner and Contractor may be brought by either party to the appropriate State of Massachusetts court having jurisdiction in the matter. The Appellant shall notify the other party promptly at the time of the filing of the matter in court.

18.6 NOTIFICATION OF SURETY:

In the event of a Claim against the Contractor, the Owner may, but is not obligated to notify the Surety, if any, of the nature and amount of the Claim. If the Claim relates to a possibility of a Contractor's default, the Owner may, but is not obligated to, notify the Surety and request the Surety's assistance in resolving the Claim.

18.7 MECHANIC'S LIEN:

If a Claim relates to or is the subject of a mechanic's lien or payment bond claim, the party asserting such Claim may proceed in accordance with applicable law to comply with the lien/bond notice or filing deadlines.

ARTICLE 19. MISCELLANEOUS PROVISIONS

19.1 FOREIGN CORPORATIONS:

Each and every contractor awarded a contract pursuant to this bid shall comply with the provisions of Massachusetts General Laws, Chapter 30, Section 39L and Chapter 156D as they relate to Foreign Corporations.

ARTICLE 20. TRAFFIC CONTROL AND BARRICADES

Contractor shall coordinate with the police and fire departments and shall initiate all measures to include erection of barricades, to insure the safety of vehicular and pedestrian traffic in the area adjacent to construction, if required for the project.

20.1 EXCAVATION

No excavation shall be left open overnight.

20.2 NOTICE OF PAVEMENT CUTS

Contractor shall notify the Town 24 hours in advance of any pavement cut and shall at that time supply any estimate of the duration of work involving disruption of traffic.

20.3 OVERNIGHT PAVEMENT CUTS

Any paving cuts left overnight shall be marked with an approved illuminated warning device.

20.4 WARNING DEVICES

Refer to Standard Specification Section 850 for general policy and description of warning devices.

ARTICLE 21. PARTIAL AWARD

21.1 The Owner reserves the right to award all or part of the Contract item stated in the specification or to reduce the amount of work under any item by agreement with the lowest eligible bidder.

21.2 A number of alternate prices may be requested in the Bid and the Town reserves the right to award the Contract on the basis of any one of the proposed alternatives.

ARTICLE 22. PRE-BID CONFERENCE

A non-mandatory pre-bid conference will be held at 12:00pm noon on at the Department of Public Works 382 Falmouth Road, Hyannis, MA 02601. All bidders' questions regarding the project shall be submitted in writing through the Town's Bonfire online Procurement Portal via the Bonfire Opportunity Q&A by the date shown on the cover page of this bid document.

ARTICLE 23. ROAD OPENING/TRENCH PERMIT

The Contractor, if required, shall apply to the Town for a Road Opening/Trenching Permit at least 24 hours in advance of commencement of work in the Town right-of-way or on Town property. A form for this purpose can be obtained from the Department of Public Works Administration and Technical Support Division. The cost of the Permit is \$160.00.

ARTICLE 24. NOTIFICATION OF UTILITIES

In accordance with Chapter 502 of the Acts of 1980, the Contractor shall notify the applicable Water District, Eversource, Comcast, Verizon Telephone Company and National Grid, 72 hours, Saturdays, Sundays and holidays excluded prior to commencing work on the site. Evidence of this notification must be furnished to the Town in order to obtain the road opening permit of Article 23 of these Contract General Conditions. NOTIFICATION OF UTILITIES

The following utility companies which may maintain underground lines or equipment in the project area may be contacted for the required notification of excavation by a single call to DIG-SAFE CENTER 1-888-344-7233.

EVERSOURCE (electric) 484 Willow Street Yarmouth, MA 02673

VERIZON (telephone) 44 Old Town House Road South Yarmouth, MA 02664

NATIONAL GRID (gas) 40 Sylvan Road South Yarmouth, MA 02664

COMCAST (cable) 10 Old Townhouse Road South Yarmouth, MA 02664

Open Cape (fiber optic) 3195 Main Street Barnstable, MA 02603 Brian Mello, Sr. Account Executive 508-44-5832 (Emergencies 800-592-2000) brian.mello@eversource.com

Kevin Overko 774-218-6039 Kevin.overko@verizonwireless.com

Richard Logue 781-907-3155 (O), 617-839-5268(C) richard.logue@nationalgrid.com

Jeffrey Mello 774-907-3155 (O), 617-839-5268 (C) Jeffrey_mello@comcast.com

Steven Johnson 508-524-5905 sjohnson@opencape.org

In addition, direct contact must be made with the applicable water district and municipal utilities.

HYANNIS

Water Supply Division 47 Old Yarmouth Road Hyannis, MA 02601 Hans Keijser 508-775-0063 Hans.Keijser@town.barnstable.ma.us

CENTERVILLE, MARSTONS MILLS AND OSTERVILLE

C.O.M.M. Water Department 1138 Main Street Osterville, MA 02655

<u>COTUIT</u>

Cotuit Fire District Water Department 4300 Falmouth Road Cotuit, MA 02635 Craig Crocker 508-438-6691 <u>CCrocker@commfiredistrict.com</u>

Chris Wiseman 508-438-2687 chris@cotuitwater.org

BARNSTABLE VILLAGE

FY25 Bismore Park Marina Bulkhead Rehabilitation

Barnstable Fire District Water Department 1841 Phinney's Lane Barnstable, MA 02630

TOWN WIDE

I.T. Department (fiber optic) 367 Main Street Hyannis, MA 02601

Water Pollution Control Division 617 Bearse's Way Hyannis MA 02601 Tom Rooney 508-362-6498 <u>bfdwatersupt@comcast.net</u>

James Benoit 508-862-4624 james.benoit@town.barnstable.ma.us

Andrew Boulé 508-790-6335 Andrew.Boule@town.barnstable.ma.us

Dig-Safe cannot be relied upon to locate electric utilities that are "privately" owned. This can include electric cables located in public ways that run from utility poles to buildings.

END OF SECTION

SECTION 4 – SPECIAL CONDITIONS

SPECIAL CONDITIONS

SECTION 5 – BID SUBMITTAL REQUIREMENTS

Bidders must complete Required Bid Submittal Forms and other requested information in their entirety and upload to the procurement portal prior to the closing date in the format requested. Additional instructions available on the procurement portal and technical assistance available by emailing <u>Support@GoBonfire.com</u>.

Bids must be accompanied by a bid security in the amount of <u>five percent (5%) of the Bid Price</u> in the form of a bid bond or certified, treasurer's or cashier's check issued by a responsible bank or trust company (herein "check"). Both the bid bond or check deposits are to be scanned and uploaded to the Town of Barnstable Bonfire Procurement Portal as a pdf file.

IMPORTANT: If bidder elects to make a bid deposit in the form of a "check" the bidder must scan a copy of the check and upload to the procurement portal where requested and have the check physically delivered to the Awarding Authority to the Town of Barnstable Procurement Office Lockbox, 230 South Street, Hyannis, MA 02601, prior to the date and time of the bid opening.

Bid Submittal Checklist:

- ✓ Form for General Bid
- ✓ Bid Table
- ✓ Combined Certificate of Compliance
- ✓ Required 5% Bid Bond
- ✓ Contractor References
- ✓ List of Sub-Contractors
- ✓ List of All Equipment to be Used
- ✓ Proposed Construction Schedule
- ✓ Minority & Women Business Enterprise (MWBE) Letters of Intent

Note: 100% Payment and Performance Bonds and Certificate of Insurance will be required with submission of the signed contract.

REQUIRED BID SUBMITTAL FORM – FORM FOR GENERAL BID

FY25 BISMORE PARK MARINA BULKHEAD REHABILITATION

Town of Barnstable

1. FORM FOR GENERAL BID

To: Town of Barnstable, Awarding Authority

From:

Α.

Β.

Name of Contractor <u>**Bid</u></u>: The undersigned proposes to furnish all labor and materials required for:</u>**

(FY25 Bismore Park Marina Bulkhead Rehabiltation) Town of Barnstable, Massachusetts, in accordance with the accompanying plans and specifications prepared by Foth Infrastructure and Environmental LLC. as specified subject to the additions and deductions according to the terms of the specifications, for the lump sum price of:

Total Contract Price in Words (which governs)

Dollars \$

(in Numbers)

<u>Alternate Prices</u>: The undersigned Bidder submits the following alternate prices, as

described in the Bidding Documents, which are to be added to or deducted from the above stated Bid submittal, as may be selected by the Awarding Authority for inclusion into this Contract. (In the event that as alternate does not affect the Contract Price, the Bidder shall remark, "No Change"). The following prices for listed alternates will be accepted at the Owner's discretion in the following order to the base contract lump sum Bid. The contract will be awarded based upon the lowest total price of base bid and accepted alternates.

Alternate No. 1:

\$

Construction Debris Boom

Total Alternate Price 1 in Words (which governs)

(Total Alternate Price 1 in Numbers)

Rule For Award:

The contract will be awarded to the lowest responsible, eligible bidder based on base bid (plus selected alternates, if acceptable).

This bid includes addenda

- C. <u>Addendum</u>: numbered
- **D.** The undersigned, as bidder, declares under penalties of perjury that the only persons or parties interested in this Bid as principals are those named herein; that this Bid is made and submitted in good faith and without collusion or fraud with any other person, firm for corporation; that he has filed all state tax returns and paid all state taxes under law; that he has carefully examined the locations of the proposed work, the proposed form of contract, the standard specifications and plans therein referred to and the Special Conditions herein annexed; and he proposes and agrees, if this Bid is accepted, that he will contract with the Awarding Authority, in the form of the contract referred to herein and to be annexed hereto, to provide all necessary machinery, tools, apparatus and other means of construction and to do all the work and furnish all the materials specified in the contract, in the manner and time herein prescribed, and according to the requirements of the Engineer as therein set forth.
- E. The undersigned agrees that if presented with the Notice of Acceptance for this contract, he will within ten (10) days, Saturdays, Sundays and legal holidays excluded, execute a contract in accordance with the terms of this bid and furnish a performance bond and a payment bond, each of a surety company qualified to do business under the laws of the Commonwealth and satisfactory to the awarding authority and each in the sum of one hundred (100%) percent of the contract price, the premiums for which are to be paid by the Contractor and are included in the contract price; and within five (5) days of the bid opening, the bidder will submit all M/WBE required documentation, per Section 6, as applicable.

Bidder accepts the provisions of the Contract as to liquidated damages.

Contractor Firm Name:	
Contractor Address:	
Authorized Signature: Print Name of Authorized Signer:	
Title:	
Date:	
Telephone:	
FY25 Bismore Park Marina Bulkhead Rehabilitation	Page 76 of 106

Email: For Town of Barnstable tracking purposes, is the submitting bidder a state certified Not a MBE or WBE (please check which is _____MBE or _WBE ____applicable)

REQUIRED BID SUBMITTAL FORM – BID TABLE

BID TABLE (FORMERLY SCHEDULE OF PRICES)

Download the Bid Table from the Town's online Bonfire Procurement Portal: https://townofbarnstable.bonfirehub.com in the FY25 Bismore Park Marina Bulkhead Rehabilitation

Complete the table as instructed and upload to the procurement portal.

Technical support available at <u>Support@GoBonfire.com</u>.

BID TABLE

PART 1: BASE BID

Item No.	Item Description	Units	Estimated Quantity	Unit Price (In Figures)	Total Price (In Figures)
02100-1	Mobilization and Demobilization	Lump Sum	1		
09900-1	Bulhead Section Loss Inspection (UT Readings)	Lump Sum	1		
09900-2	Bulkhead Planting	SF	450		
09900-3	Epoxy Repairs	SF	200		
09900-4	Splash Zone Epoxy Repairs	SF	400		
13110-1	Cathodic Protection	Lump Sum	1		
33-0500-1	Utility Installation and Pedestrian Walkway Restoration	Lump Sum	1		
				TOTAL:	

PART 2: BID ALTERNATE

Item No.	Item Description	Units	Estimated Quantity	Unit Price (In Figures)	Total Price (In Figures)
09900-5	Construction Debris Boom	Lump Sum	1		

REQUIRED BID SUBMITTAL FORM – COMBINED CERTIFICATIONS

CERTIFICATIONS REQUIRED BY LAW FOR PUBLIC CONSTRUCTION CONTRACTS

You must COMPLETE and SIGN the following certifications. You must also print, at the bottom of this page, the name of the contractor for whom these certifications are submitted.

TAX COMPLIANCE

Pursuant to Chapter 62C of the Massachusetts General Laws, Section 49A(b), I, the undersigned, authorized signatory for the below named contractor, do hereby certify under the pains and penalties of perjury that said contractor has complied with all laws of the Commonwealth of Massachusetts relating to taxes, reporting of employees and contractors, and withholding and remitting child support.

NON-COLLUSION

The undersigned certifies under the penalties of perjury that this bid is in all respects bona fide, fair and made without collusion or fraud with any other person. As used in this subsection the word "person" shall mean any natural person, joint venture, partnership, corporation or other business or legal entity.

PUBLIC CONTRACTOR DEBARMENT

The undersigned certifies under penalty of perjury that the below named contractor is not presently debarred from doing public construction work in the commonwealth under the provisions of section twenty-nine F of chapter twenty-nine, or any other applicable debarment provisions of any other chapter of the General Laws or any rule or regulation promulgated thereunder.

OSHA TRAINING

Pursuant to G.L. c. 30, §39S, the Contractor hereby certifies under penalties of perjury as follows:

- (1) Contractor is able to furnish labor that can work in harmony with all other elements of labor employed or to be employed in the work;
- (2) All employees to be employed at the worksite will have successfully completed a course in construction safety and health approved by the United States Occupational Safety and Health Administration that is at least 10 hours in duration at the time the employee begins work and they shall furnish documentation of successful completion of said course with the first certified payroll report for each employee; and
- (3) All employees to be employed in the work subject to this contract have successfully completed a course in construction safety and health approved by the United States Occupational Safety and Health Administration that is at least 10 hours in duration.

IF A FOREIGN CORPORATION

I hereby certify that I comply with the provisions of Massachusetts General Laws, Chapter 30, Section 39L and Chapter 156D as they relate to Foreign Corporations.

(Check one) 🗌 Not Applicable

□ I comply and am registered and in good stating with the Massachusetts Secretary of State

COMPLETE AND SIGN BELOW:

Authorized Person's Signature

Date

Print Name & Title of Signatory

Name of Contractor

FY25 Bismore Park Marina Bulkhead Rehabilitation

REQUIRED BID SUBMITTAL FORM – CONTRACTOR REFERENCES

FY25 Bismore Park Marina		
Bulkhead Rehabilitation:		

BIDDER NAME:

Bidders must provide a list of at least five (5) references to which similar size and scope projects have been completed within the past five (5) years, along with a name of a contact person and phone numbers. (Municipalities desired, if applicable). Additionally, please attach to this form a complete list of ongoing projects, projects completed within the past two years including project contact names, values and contact names and numbers. The Town reserves the right to obtain additional references at their option. The Town shall interpret whether project references are similar in scope and size.

1.	Owner:	Project Date:	
	Contact:	Phone:	
	Project Value and Description:		
2.	Owner:	Project Date:	
	Contact:		
	Project Value and Description:		
3.	Owner:	Project Date:	
	Contact:		
	Project Value and Departmetion		

REQUIRED BID SUBMITTAL FORM – REFERENCES - continued

4.	Owner:	Project Date:	
	Contact:		
	Project Value and Description:		
5.	Owner:	Project Date:	
	Contact:	Phone:	
	Project Value and Description:		

REQUIRED BID SUBMITTAL FORM – LIST OF SUB-CONTRACTORS

LIST OF SUB-CONTRACTORS

FY25 BISMORE PARK MARINA BULKHEAD REHABILITATION

Any person making a bid or offer to perform the work, shall in his or her bid or offer, set forth: (a) The name and location of the place of business of each subcontractor who will perform work or labor or render service to the prime Contractor in or about the construction of the work or improvement, or a subcontractor licensed by the Commonwealth of Massachusetts who, under subcontract to the primary Contractor specially fabricates and installs a portion of the work or improvement according to detailed drawings contained in the plans and specifications, in an amount in excess of one-half of 1 percent of the prime Contractor's total bid; (b) The portion of the work which will be done by each such subcontractor under this act. The prime Contractor shall list only one subcontractor for each such portion as defined by the prime Contractor in his or her bid.

Any item of work, which does not set forth a designated Subcontractor will be done by the Prime Contractor.

Name & Address	Portion of Work

(Prime Contractor)

Signed by:

Title:

REQUIRED BID SUBMITTAL FORM – LIST OF ALL EQUIPMENT TO BE USED

LIST OF ALL EQUIPMENT TO BE USED

FY25 BISMORE PARK MARINA BULKHEAD REHABILITATION

Contractor is required to submit for review copies of all applicable current inspections and certificates, as applicable. Use additional sheets as necessary.

Proposed Equipment:

Type/Name	Manufacturer	Age & Condition

Signed by:

Title:

SECTION 6 – MINORITY & WOMEN OWNED BUSINESS PARTICIPATION REQUIREMENTS

TOWN OF BARNSTABLE

Minority and Woman Owned Business Participation Requirements

Minority Business Enterprise (MBE) and Women's Business Enterprise (WBE) policies of the Town of Barnstable are applicable. The "Fair Share" construction <u>requirement</u> for this project is a minimum of **10.4%** combined MBE/WBE participation by state certified MBEs and WBEs. Within five days after the Bid Opening, the Bidder shall submit a "Schedule for Participation by Minority Business Enterprises" with accompanying Letters of Intent by each minority subcontractor proposed to be used by the Bidder. The Bidder shall submit a "Schedule for Participation by Women Business Enterprises" with accompanying Letters of Intent by each WBE subcontractor to be used by the Bidder including their SBA approval letter. The Letters of Intent shall include, among other things, the contract items the M/WBE is proposing to perform and the prices that the M/WBE proposed to charge for the work. Original signed copies of the letters will be required prior to the signing of a contract. The Schedule of Participation shall list these M/WBE subcontractors with whom the Contractor intends to contract and state the total price to be paid each M/WBE contractor as taken from each Letter of Intent submitted with the bid. Failure to comply with the requirements of this paragraph may be deemed to render a Bid non-responsive.

See Town of Barnstable MBE/WBE Policy – Available on the Town of Barnstable website: <u>https://www.townofbarnstable.us/Departments/purchasing/Bulletin_Board/Town-of-Barnstable-MinorityWoman-Business-Enterprise-Plan-(MBEWBE)-.pdf?tm=2/28/2023%208:44:24%20AM</u>

Requirement – 10.4% combined MBE/WBE for this project

SECTION 7 – PREVAILING WAGE RATES

PREVAILING WAGE RATES

Prevailing wages, attached hereinafter, apply to this project and are a part of this Contract.

SECTION 8 - OWNER-CONTRACTOR CONTRACT FORM

CONTRACT BETWEEN CONTRACTOR AND TOWN OF BARNSTABLE

THIS CONTRACT is made as of the _____ day of _____ 20___ by and between the TOWN OF BARNSTABLE, Massachusetts, a municipal corporation with its main address at 367 Main Street, Hyannis, MA 02601 (the "Town") and **CONTRACTOR NAME** with legal address and principal place of business at ______ (the "Contractor"):

WITNESSETH: That for and in consideration of the payments to be made by the Town, the Contractor hereby agrees with the Town to commence and complete the **FY25 BISMORE PARK MARINA BULKHEAD REHABILITATION**. (the "Project), for the consideration set forth in the Bid, as defined herein, and all extra work in connection therewith, under the terms as stated in the Invitation for Bid dated ______, 20____ including Addendums __ and ") including all drawings, specification and attachments thereto (the "IFB") and the additional terms and conditions set forth herein; and at the Contractor's own proper cost and expense to furnish all the materials supplies, machinery, equipment, tools, superintending, labor, insurance, and other accessories and services necessary to complete said Project in accordance with the prices stated in the Contractor's bid submittal dated ______, 20____ (the "Bid"), all of which are made a part hereof by reference and collectively evidence and constitute the Contract.

<u>Term</u> – The Term of this Contract shall commence as of the date set forth above and shall terminate on _____, 20___ unless sooner terminated as provided herein. Construction is anticipated to commence upon the issuance of a written notice to proceed by the Town after execution of this Contract.

<u>Contract Value</u> – Total payments to be made hereunder shall not exceed \$______(PRICE IN WORDS)

The Minority and Woman Owned Participation commitment for this project is \$______ (PRICE IN WORDS). Compliance with this commitment shall be monitored by the Town project manager and reported to the Contract Compliance Officer.

The Town agrees to pay the Contractor for the performance of this Contract, subject to additions and deductions, as provided in the General Conditions of the IFB, and make payments on account thereof as provided in Article 12 Payments and Completion in the General Conditions of the IFB.

LIQUIDATED DAMAGES: Contractor and Owner recognize that time is of the essence and that Owner will suffer financial loss if the Work is not completed within the times specified in the Contract, plus any extensions thereof allowed in accordance the General Conditions and this contract. The parties also recognize the delays, expense, and difficulties involved in proving in a legal or arbitration proceeding the actual loss suffered by Owner if the Work is not completed on time. Accordingly, instead of requiring any such proof, Owner and Contractor agree that as liquidated damages for delay (but not as a penalty), Contractor shall pay Owner \$_____ for each day that expires after the time specified in the Contract for Substantial Completion until the Work is substantially complete unless otherwise extended per an agreed and approved change order for work that was not included in the original bid. After Substantial Completion, if Contractor shall neglect, refuse, or fail to complete the remaining Work within the Contract Time or any proper extension thereof granted by Owner, Contractor shall pay Owner \$______ for each day Owner \$______ for each day that expires after the time specified in the original bid. After Substantial Completion, if Contractor shall neglect, refuse, or fail to complete the remaining Work within the Contract Time or any proper extension thereof granted by Owner, Contractor shall pay Owner \$_______ for each day that expires after the time or and proved thereof and approved change or any proper extension thereof granted by Owner, Contractor shall pay Owner \$________ for each day that expires after the or any proper extension thereof granted by Owner, Contractor shall pay Owner \$________ for each day that expires after the

time specified in the Contract for completion and readiness for final payment until the Work is completed and ready for final payment.

<u>Force Majeure</u> – This Contract shall be subject to Force Majeure considerations. Either party hereto shall be excused from performance of any act under this Contract if prevented from the performance of any act required by reasons of strikes, lockouts, labor trouble, failure of power, fire, winds, Acts of God, riots, insurrections, war or other reason of a like nature not reasonably within the control of the party. The period for the performance of such obligation shall be extended for an equivalent period for no additional cost. Continued prevention from performance by such causes for periods aggregating sixty (60) or more days shall be deemed to render performance impossible, and either party shall thereafter have the right to terminate this Contract upon written notice to the other party.

<u>Termination of Contract</u> - Subject to the provisions of the section above explaining <u>Force</u> <u>Majeure</u>, if the Contractor shall fail to fulfill in a timely and satisfactory manner its obligations under this Contract, or if the Contractor shall violate any of the covenants, conditions, or stipulations of this Contract, the Town may terminate this Contract in accordance with Article 17, Termination or Suspension of the Contract of the IFB.

<u>Insurance</u> - The Contractor shall maintain insurance with minimum limits as defined in the Invitation for Bid, Special Conditions for the entire duration of the project work to be performed, and provide a certificate of insurance with the Town of Barnstable named as an additional insured. Renewal certificates of insurance must be submitted to the Town of Barnstable, Risk Management, 367 Main Street, Hyannis, MA 02601 on a yearly basis.

Governing Law, Jurisdiction and Venue. This Contract shall be governed by and construed pursuant to the laws of the Commonwealth of Massachusetts without regard to its conflict of laws principles. Venue for any legal actions initiated concerning this Contract or arising in any way from and out of this Contract shall be brought in the appropriate state court sitting in Barnstable County, having jurisdiction over said claim. The parties waive any right they may have to venue in any other jurisdiction.

Massachusetts General Laws Chapter 30, sec. 39M hereby applies to this Contract. Prevailing Wage Rates dated 10/31/2024 apply to this Contract. The Contractor shall submit weekly certified payrolls with invoices to Town of Barnstable, Attn: NAME/ADDRESS. OSHA 10 certification required for all employees and subcontractors performing work on the job site. A one hundred percent (100%) payment bond and a one hundred percent (100%) performance bond are required with this signed Contract. Contractor hereby agrees to comply with all Town, State or Federal Directives and guidelines in regard to construction and health safety related to COVID-19 Pandemic.

It is expressly understood and agreed that Contractor shall at all times during the term of this Contract act as an independent contractor, and shall not have any authority to bind the Town. Contractor and Contractor's employees, agents and/or subconsultants shall not be deemed to be employees or agents of the Town. Further, nothing contained herein shall be construed to create a joint venture, partnership, association or other affiliation between the Contractor and the Town.

To the fullest extent permitted by law, the Contractor shall indemnify, defend, and save harmless the Town and all of the Town officers, agents and employees from and against all suits and claims of liability of every name and nature, including attorney's fees and costs of defending any action or claim, for or on account of any claim, loss, liability or injuries to persons or damage to property of the Town or any person, firm, corporation or association arising out of or resulting from any act, omission, or negligence of the Contractor, subcontractors and their agents or employees in the performance of the work covered by this Contract and/or their failure to comply with terms and conditions of this Contract, regardless of whether said claim is caused in part by the Town or any third party. The foregoing provisions shall not be deemed to be released, waived or modified in any respect by reason of any surety or insurance provided by the Contractor under this Contract with the Town. The provisions of this paragraph shall survive the termination or expiration of the Contract.

IN WITNESS WHEREOF, the parties to these present have executed this on the dates set forth below.

Approved as to form: By:	By:	
Karen L. Nober, Town Attorney	CONTRACTOR	DATE
	Print Name of Authorized Signer &	Title
	By: TOWN OF BARNSTABLE	

Mark S. Ells, Town Manager I hereby certify that the Town of Barnstable has an appropriation to cover the cost of this Contract in accordance with Ch. 44 §31C of the Massachusetts General Laws. By:

Mark Milne, Finance Director

SIGNATORY AUT	HORIZATION FORM	DATE:
FY25 BISMORE		
PARK MARINA		
BULKHEAD REHABILITATION :		CWSRF#: (as applicable)
At a duly constitu	Ited meeting of	held on
at which all Direc	tors were present or wa	(Name of Corporation) (Date of Meeting) ived notice, it was voted that:
behalf of said co	be and he/she is hereb	(Title of Officer) by authorized to execute contracts and bonds in the name and orporate Seal thereto, and such execution of any contract or behalf of such
		under seal of the company, shall be valid and
binding upon this	(Officer)	
binding upon this	s company.	A TRUE COPY, ATTEST:
Place of Business	:	(Clerk of the Corporation) (Print Name & Signature)
I hereby certify th	nat I am the clerk of the	and that
		is duly elected of
Company, and th	lame of Officer) e above vote has not be nature of this contract.	(Print Signature Name & Title) en amended or rescinded and remains in full force and effect as
(CORPORATE SEA	L)	(Clerk of the Corporation)
	5	ertify that I comply with the provisions of Massachusetts d Chapter 156D as they relate to Foreign Corporations.
<mark>(Check one)</mark>	Not Applicable	I comply and am registered and in good stating with the Massachusetts Secretary of State.
NOTARIZATION:		
On this	_day of	, 20, before me, the undersigned notary public,
	ence of identification, wh	, proved to me through nich were n the preceding or attached document in my presence.
	_	
Notary Public		
Print Name of No	·	
My commission e	expires:	

SECTION 9 – LABOR AND MATERIALS PAYMENT BOND

LABOR AND MATERIALS PAYMENT BOND

KNOWN ALL MEN AND WOMEN BY THESE PRESENT, THAT:

as principal, and

______as surety, are held and firmly bound unto the Town of Barnstable, Hyannis, Massachusetts in the sum of:

lawful money of the United State of America, to be paid to the Town of Barnstable, Hyannis, Massachusetts, for which payments, well and truly to be made, we bind ourselves, our respective heirs, executors, administrators, successors and assigns, jointly and severally, firmly by these presents.

WHEREAS, the said principal has made a contract with the Town of Barnstable, Hyannis,

Massachusetts, bearing the date _____ of _____, 20____, for the construction

project:

FY25 BISMORE PARK MARINA BULKHEAD REHABILITATION

Now the condition of this obligation is such that if the principal shall pay for all labor performed or furnished and for all materials used or employed in said contract and in any and all duly authorized modifications, alterations, extensions of time, changes or additions to said contract that may hereafter be made, notice to the surety of such modifications, alterations, extensions of time, changes or additions being hereby waived, the foregoing to include any other purpose or items set out in, and subject to, the provisions of Massachusetts General Laws, Chapter 30, Section 39A, and Chapter 149, Section 29, as amended, then this obligation shall become null and void; otherwise it shall remain in full force and virtue.

IN WITNESS THEREOF, we hereunto set our hands and seals this _____ day of

______, 20____.

(Seal)

By: _____

Ву: _____

SECTION 10 – PERFORMANCE BOND

PERFORMANCE BOND

KNOW ALL MEN AND WOMEN BY THESE PRESENT, THAT:

as principal,

and

as surety,

are held and firmly bound unto the Town of Barnstable, Hyannis, Massachusetts, in the sum of \$______ lawful money of the United States of America, to be paid to the Town of Barnstable, Hyannis, Massachusetts, for which payments, well and truly to be made, we bind ourselves, our respective heirs, executors, administrators, successors and assigns, jointly and severely, firmly by these presents.

WHEREAS, the said principal has made a contract with the Town of Barnstable, Hyannis, Massachusetts, bearing the date of ______, 20____, for the construction of Project:

FY25 BISMORE PARK MARINA BULKHEAD REHABILITATION

Now the condition of this obligation is such that if the principal shall well and truly keep and perform all the undertakings, covenants, agreements, terms and conditions of said contract on its part to be kept and performed during the original term of said contract any extensions thereof that may be granted by the Town of Barnstable, Hyannis, Massachusetts, with or without notice to the surety, and during the life of any guarantee required under the contract, and shall also well and truly keep and perform all the undertakings, covenants, agreements, terms and conditions of any and all duly authorized modifications, alterations, changes or additions to said contract that may be hereafter made, notice to the surety of such modifications, alterations, changes or additions being hereby waived, then this obligation shall become null and void; otherwise it shall remain in full force and virtue.

IN WITNESS WHEREOF we hereunto set of	our hands and seal this day of
, 20	·
I	Зу:
	Зу:

SECTION 11 – ACCEPTANCE OF BID NOTICE

ATE:			
	(Contra	actor Name/Address)	
hereby noti	fied that their bid for the:		
	(FY25 Bismore Parl	K Marina Bulkhead Re	ehabilitation)
accordance	e with the Invitation for Bid date	d	
nd Addendu	m(s) There	to, in the amount of	
		(in words)	
	dated		Has been accepted.
nall provide a	a 100% payment and performa	nce bond per the Invi	Town of Barnstable. Contractor tation for Bid and supply a certificate red with the contract document.
		Mark S. Ells, Tow	/n Manager
	Procurement Department Amber Patterson, Chief Pro S: 367 Main Street Hyannis, MA 02601		
EMAIL:	Amber.Patterson@town.bar		
By:	Receipt is hereby acknowle	dged for the above A	CCEPTANCE OF BID
	Signature		Date
	Print Name		Print Title

SECTION 12 – APPLICATION & CERTIFICATION FOR PAYMENT

CONTRACT #:		
TITLE:		
CONTRACTOR:		
TO:	Town of Barnstable, Department of P	ublic Works Owner's Project Manager
Application Date:		
Period:	From	to

NOTE: In order to receive payment for the monthly period covered by this application, this form shall be delivered to the Town of Barnstable on the third Monday of each month or the working day immediately preceding. Amounts not so applied for shall carry over to the next scheduled billing period.

CHANGE ORDER SUMMARY

Number	Date	
TOTALS		

ORIGINAL CONTRACT SUM	\$
Net Change by Change Order	\$
Contract Sum to Date	\$
TOTAL COMPLETED TO DATE	\$
Retainage	\$
Total Earned Less Retainage	\$
Less, Previous Certificates for Payment	\$
Current Payment Due	\$

The undersigned certifies that the work covered by this application has been completed in accordance with the Contract Documents, that all amounts have been paid by them for Work and Materials for which previous Certificates for Payments have been issued and payments received from the Town of Barnstable, that all Contractor and Sub-contractor payroll data for the time period covered by this application has been submitted to the Town and that the current payment shown herein is now due.

CONTRACTOR:	

DATE:

SECTION 13 – TAX EXEMPTION NUMBER

TOWN OF BARNSTABLE

HYANNIS, MA 02601

(508) 862-4090

TAX EXEMPTION NUMBER

Date: _____

TO WHOM IT MAY CONCERN:

This is to certify that whenever

purchases material and supplies for projects awarded by bid for the Town of Barnstable, all materials used on these projects are tax exempt.

Our Tax Exempt number is E-046-001-079.

Owner's Project Manager

SECTION 14 – TOWN OF BARNSTABLE CHANGE ORDER

CHANGE OF	RDER NO.	DATE:
CONTRACT	NO	PROJECT NO.
CONTRACT	OR'S NAME:	
CONTRACT	OR'S ADDRESS:	
LIST ORE	IGINAL CONTRACT AMOUNT ALL PREVIOUS CHANGE DERS HANGE ORDER#(s)	\$ \$
AMC	OUNT OF THIS ORDER	
_	_decrease increase /ISED CONTRACT AMOUNT	\$ \$
authorized.	ase \Box decrease \Box no change of o	
This order o	overs the contract modification hereunder	described:
	chments: overed by this order shall be performed un the original construction contract.	der the same terms and conditions as
Change App	proved:	
Ву:	Contractor	Date:
Title: TOWN	OF BARNSTABLE	
Ву:	Project Manager	Date:
Ву:	Griffin Beaudoin, P.E., Town Engineer	Date:
Ву:	Daniel Santos, P.E., DPW Director	Date:
Ву:	Mark Milne, Town Accountant Verify funds are available for this Char Order	Date:
Ву:	Mark S. Ells, Town Manager	Date:

SECTION 15 – NOTICE TO PROCEED

TOWN OF BARNSTABLE NOTICE TO PROCEED

DATE: _____, 20_____

SUBJECT CONTRACT: FY25 BISMORE PARK MARINA BULKHEAD REHABILITATION

To:

1. You are hereby given formal <u>NOTICE TO PROCEED</u> in accordance with the provisions of the subject contract.

2. It is requested that acknowledgment of this NOTICE be indicated by endorsement hereon, and that the original be returned to this office. The duplicate should be retained in your office files.

Paul Graves, P.E., Project Manager

FIRST ENDORSEMENT

TO: Town of Barnstable Owner's Project Manager

Receipt is hereby acknowledged of the above **NOTICE TO PROCEED** under this contract.

FY25 BISMORE PARK MARINA BULKHEAD REHABILITATION:

Ву: _			
Title:			
Date:			

APPENDIX A Town of Barnstable Fuel Adjustment Clauses Barnstable Rev 8/2018

Document 00811; Speciation Provisions, Monthly Price Adjustment for Hot Mix Asphalt (HMA) Mixtures

Document 00812; Speciation Provisions, Monthly Price Adjustment for Diesel Fuel and Gasoline

Document 00813; Speciation Provisions, Price Adjustment for Structural Steel and Reinforcing Steel

Document 00814; Speciation Provisions, Price Adjustment for Portland Cement Concrete Mixes

DOCUMENT 00811- Speciation Provisions, Monthly Price Adjustment for Hot Mix Asphalt (HMA) Mixtures

TOWN OF BARNSTABLE SPECIAL PROVISIONS MONTHLY PRICE ADJUSTMENT FOR HOT MIX ASPHALT (HMA) MIXTURES ENGLISH AND METRIC UNITS Revised: 07/08/2016

This provision applies to all projects using hot mix asphalt (HMA) mixtures containing liquid asphalt cement as stipulated in the Notice to Contractors section of the bid documents.

Price Adjustments will be based on the variance in price, for the liquid asphalt component only, between the Base Price and the Period Price. They shall not include transportation or other charges. Price Adjustments will occur on a monthly basis.

Base Price

The Base Price of liquid asphalt on a project as listed in the Notice to Contractors section of the bid documents is a fixed price determined by the Department at the time of the bid using the same method as the determination of the Period Price detailed below. The Base Price shall be used in all bids.

Period Price

The Period Price is the price of liquid asphalt for each monthly period as determined by the Department using the average selling price per standard ton of PG64-28 paving grade (primary binder classification) asphalt, FOB manufacturer's terminal, as listed under the "East Coast Market - New England, Boston, Massachusetts area" section of the Poten & Partners, Inc. "Asphalt Weekly Monitor". This average selling price is listed in the issue having a publication date of the second Friday of the month and will be posted as the Period Price for that month. The Department will post this Period Price on its website at http://www.mhd.state.ma.us/ within two (2) business days following its receipt of the relevant issue of the "Asphalt Weekly Monitor". Poten and Partners has granted the Department the right to publish this specific asphalt price information sourced from the Asphalt Weekly Monitor. This method of period price determination was formerly called the New Asphalt Period Price Method. Separate website postings using both the New Asphalt Period Price Method and the Old Asphalt Period Price Method were discontinued after June 2013.

Price Adjustment Determination, Calculation and Payment

The Contract Price of the HMA mixture will be paid under the respective item in the Contract. Price Adjustments, as herein provided, either upwards or downwards, will be made after the work has been performed using the monthly period price for the month during which the work was performed.

Price Adjustments will be paid only if the variance from the Base Price is 5% or more for a monthly period. The complete adjustment will be paid in all cases with no deduction of the 5% from either upward or downward adjustments.

The Price Adjustment applies only to the actual virgin liquid asphalt content in the mixture placed on the job in accordance with the Standard Specifications for Highways and Bridges, Division III, Section M3.11.03.

Price Adjustments will be separate payment items. The pay item numbers are 999.401 for a positive price adjustment (a payment) and 999.402 for a negative price adjustment (a deduction). Price Adjustments will be calculated using the following equation: Price Adjustment = Tons of HMA Placed X Liquid Asphalt Content % X RAP Factor X (Period Price - Base Price) No Price Adjustment will be allowed beyond the Completion Date of this Contract, unless there is a Department-approved extension of time.

FY25 Bismore Park Marina Bulkhead Rehabilitation

TOWN OF BARNSTABLE SPECIAL PROVISIONS MONTHLY PRICE ADJUSTMENT FOR DIESEL FUEL AND GASOLINE – ENGLISH UNITS

Revised: 01/26/2009

This monthly fuel price adjustment is inserted in this contract because the national and worldwide energy situation has made the future cost of fuel unpredictable. This adjustment will provide for either additional compensation to the Contractor or repayment to the Commonwealth, depending on an increase or decrease in the average price of diesel fuel or gasoline.

This adjustment will be based on fuel usage factors for various items of work developed by the Highway Research Board in Circular 158, dated July 1974. These factors will be multiplied by the quantities of work done in each item during each monthly period and further multiplied by the variance in price from the Base Price to the Period Price.

The Base Price of Diesel Fuel and Gasoline will be the price as indicated in the Department's web site

(<u>www.mhd.state.ma.us</u>) for the <u>month in which the contract was bid</u>, which includes State Tax.

The Period Price will be the average of prices charged to the State, including State Tax for the bulk purchases made during each month.

This adjustment will be effected only if the variance from the Base Price is 5% or more for a monthly period. The complete adjustment will be paid in all cases with no deduction of the 5% from either upward or downward adjustments.

No adjustment will be paid for work done beyond the extended completion date of any contract.

Any adjustment (increase or decrease) to estimated quantities made to each item at the time of final payment will have the fuel price adjustment figured at the average period price for the entire term of the project for the difference of quantity.

The fuel price adjustment will apply <u>only</u> to the following items of work at the fuel factors shown:

ITEMS COVERED	FUEL FACTORS		
	Diesel	Gasoline	
Excavation: and Borrow Work: Items 120, 120.1, 121, 123, 124, 125, 127, 129.3, 140, 140.1, 141, 142, 143, 144., 150, 150.1, 151 and 151.1 (Both Factors used)	0.29 Gallons / CY.	0.15 Gallons / CY	
Surfacing Work: All Items containing Hot Mix Asphalt	2.90 Gallons /	Does Not Apply	

******** END OF DOCUMENT 00812 ********

TOWN OF BARNSTABLE

SPECIAL PROVISIONS

PRICE ADJUSTMENTS FOR STRUCTURAL STEEL AND REINFORCING STEEL FOR CONTRACTS BID ON OR AFTER APRIL 5, 2011

ENGLISH UNITS December 12, 2016

This provision applies to projects containing a price adjustment for structural steel and reinforcing steel as stipulated in the Notice to Contractors section of the Bid Documents. It applies to all structural steel as defined below and all reinforcing steel on the project. Compliance with this provision is mandatory, i.e., there are no "opt-in" or "opt-out" clauses. Price adjustments will be handled as described below and shall only apply to unfabricated structural steel material, consisting of rolled shapes, plate steel, sheet piling, pipe piles, steel castings and steel forgings, and unfabricated reinforcing steel bars.

Price adjustments will be variances between Base Prices and Period Prices. Base Prices and Period Prices are defined below.

Price adjustments will only be made if the variances between Base Prices and Period Prices are 5% or more. A variance can result in the Period Price being either higher or lower than the Base Price. Once the 5% threshold has been achieved, the adjustment will apply to the full variance between the Base Price and the Period Price.

Price adjustments will be calculated by multiplying the number of pounds of unfabricated structural steel material or unfabricated reinforcing steel bars subject to a price adjustment by the index factor calculated as shown below under <u>Example of a Period</u> <u>Price Calculation</u>.

Price adjustments will <u>not</u> include the costs of shop drawing preparation, handling, fabrication, coatings, transportation, storage, installation, profit, overhead, fuel costs, fuel surcharges, or other such charges not related to the cost of the unfabricated structural steel and unfabricated reinforcing steel.

The weight of steel subject to a price adjustment shall not exceed the final shipping weight of the fabricated part by more than 10%. <u>Base Prices and Period Prices are defined</u> as follows:

<u>Base Prices</u> of unfabricated structural steel and unfabricated reinforcing steel on a project are fixed prices determined by the Department and found in the Notice to Contractors section of the Bid Documents.

The Base Price Date is the month and year in which MassDOT opened bids for the project. This date is used to select the Base Price Index.

<u>Period Prices</u> of unfabricated structural steel and unfabricated reinforcing steel on a project are variable prices calculated based on the purchase date of the steel (Period Price Date) using an index of steel prices to adjust the Base Price.

The Period Price Date is the date the steel was delivered to the fabricator as evidenced by an official bill of lading submitted to the Department containing a description of the shipped materials, weights of the shipped materials and the date of shipment. This date is used to select the Period Price Index.

The index used for the calculation of Period Prices is the U.S. Bureau of Labor Statistics (BLS) Producer Price Index (PPI) Series ID WPU101702 (Not Seasonally Adjusted, Group: Metals and Metal Products, Item: Semi-finished Steel Mill Products.) As this index is subject to revision for a period of up to four (4) months after its original publication, no price adjustments will be made until the index for the period is finalized, i.e., the index is no longer suffixed with a "(P)".

Period Prices are determined as follows:

Period Price = Base Price X Index Factor Index Factor = Period Price Index / Base Price Index

Example of a Period Price Calculation:

Calculate the Period Price for December 2009 using a Base Price from March 2009 of \$0.82/Pound for 1,000 Pounds of ASTM A709 (AASHTO M270) Grade A36 Structural Steel Plate.

The Period Price Date is December 2009. From the PPI website*, the Period Price Index = 218.0.

The Base Price Date is March 2009. From the PPI website*, the Base Price Index = 229.4.

Index Factor = Period Price Index / Base Price Index = 218.0 / 229.4 = 0.950 Period Price = Base Price X Index Factor = \$0.82/Pound X 0.950 = \$0.78/Pound

Since 0.82 - 0.78 = 0.04 is less than 5% of 0.82, no price adjustment is required. If the 0.04 difference shown above was greater than 5% of the Base Price, then the price adjustment would be 1,000 Pounds X 0.04/Pound = 40.00. Since the Period Price of 0.78/Pound is less than the Base Price of 0.82/Pound, indicating a drop in the price of steel between the bid and the delivery of material, a credit of 40.00 would be owed to MassDOT. When the Period Price is higher than the Base Price, the price adjustment is owed to the Contractor.

* To access the PPI website and obtain a Base Price Index or a Period Price Index, go to http://data.bls.gov/cgi-bin/srgate

END OF EXAMPLE.

The Contractor will be paid for unfabricated structural steel and unfabricated reinforcing steel under the respective contract pay items for all components constructed of either structural steel or reinforced Portland cement concrete under their respective Contract Pay Items.

Price adjustments, as herein provided for, will be paid separately as follows: <u>Structural</u> Steel

Pay Item Number 999.449 for positive (+) pay adjustments (payments to the Contractor)

Pay Item Number 999.457 for negative (-) pay adjustments (credits to MassDOT Highway Division)

Reinforcing Steel Pay Item Number 999.466 for positive (+) pay adjustments (payments to the Contractor)

Pay Item Number 999.467 for negative (-) pay adjustments (credits to MassDOT Highway Division)

FY25 Bismore Park Marina Bulkhead Rehabilitation

No price adjustment will be made for price changes after the Contract Completion Date, unless the MassDOT Highway Division has approved an extension of Contract Time for the Contract.

********* END OF DOCUMENT 00813*******

TOWN OF BARNSTABLE SPECIAL PROVISIONS PRICE ADJUSTMENT FOR PORTLAND CEMENT CONCRETE MIXES

This provision applies to all projects using greater than 100 Cubic Yards (76 Cubic Meters) of Portland cement concrete containing Portland cement as stipulated in the Notice to Contractors section of the Bid Documents. This Price Adjustment will occur on a monthly basis.

The Price Adjustment will be based on the variance in price for the Portland cement component only from the Base Price to the Period Price. It shall not include transportation or other charges.

The Base Price of Portland cement on a project is a fixed price determined at the time of bid by the Department by using the same method as for the determination of the Period Price (see below) and found in the Notice to Contractors.

The Period Price of Portland cement will be determined by using the latest published price, in dollars per ton (U.S.), for Portland cement (Type I) quoted for Boston, U.S.A. in the <u>Construction Economics</u> section of *ENR Engineering News-Record* magazine or at the ENR website http://www.enr.com under <u>Construction Economics</u>. The Period Price will be posted on the MassHighway website the Wednesday immediately following the publishing of the monthly price in ENR, which is normally the first week of the month.

The Contract Price of the Portland cement concrete mix will be paid under the respective item in the Contract. The price adjustment, as herein provided, upwards or downwards, will be made after the work has been performed, using the monthly period price for the month during which the work was performed.

The price adjustment applies only to the actual Portland cement content in the mix placed on the job in accordance with the Standard Specifications for Highways and Bridges, Division III, Section M4.02.01. No adjustments will be made for any cement replacement materials such as fly ash or ground granulated blast furnace slag.

The Price Adjustment will be a separate payment item. It will be determined by multiplying the number of cubic yards of Portland cement concrete placed during each monthly period times the Portland cement content percentage times the variance in price between the Base Price and Period Price of Portland cement.

This Price Adjustment will be paid only if the variance from the Base Price is 5% or more for a monthly period. The complete adjustment will be paid in all cases with no deduction of the 5% from either upward or downward adjustments.

No Price Adjustment will be allowed beyond the Completion Date of this Contract, unless there is a Department-approved extension of time.

****END OF DOCUMENT 00814****